THE ZOOGEOGRAPHY OF THE CETACEANS IN ALGOA BAY

A thesis submitted in fulfilment of the requirements for the degree of

MASTER OF SCIENCE of RHODES UNIVERSITY

by

BRIGITTE LEIGH MELLY

January 2011

ABSTRACT

The most recent study on cetaceans in Algoa Bay, South Africa, was conducted over 14 years ago. Consequently, knowledge of the cetacean species visiting this bay is currently based on incidental observations and stranding data. A number of developments in recent years: a deepwater port, proposed oil refinery, increased boating and fishing (commercial and recreational), a proposed Marine Protected Area, and the release of a whale-watching permit, all of which may impact these animals in some way, highlight the need for a baseline study on cetaceans. Therefore, the aim of this study was to determine the spatial and temporal distribution, and habitat preference of cetaceans in Algoa Bay. Boat-based surveys were conducted monthly between March 2009 and July 2010. At each sighting the GPS location, species, group size and composition, and behaviour were recorded. Using GIS, the sighting data was related to data layers of geographical variables such as sea surface temperature, depth and sea-floor substrate.

Approximately 365 hours of search effort were completed over 57 surveys, with a total of 346 sightings. Species observed were: southern right whales (*Eubalaena australis*), humpback whales (*Megaptera novaeangliae*), Bryde's whales (*Balaenoptera brydei*), Indian Ocean bottlenose dolphins (*Tursiops aduncus*), Indo-Pacific humpback dolphins (*Sousa chinensis*), and long-beaked common dolphins (*Delphinus capensis*). Southern right whales were observed during austral winter, utilising the shallow, protected areas of the bay as a mating and nursery ground. Humpback whales were also recorded extensively during winter, in more offshore waters, with a significant number of mother-calf pairs sighted. Bryde's whales were recorded in offshore waters during summer and autumn, where they were primarily observed travelling and foraging. Bottlenose dolphins were the most prolific species sighted. They were recorded year-round throughout the inshore waters of the bay, with large group sizes (up to 500 animals), and displayed a wide variety of behaviours. Humpback dolphins were observed in extremely shallow

and inshore waters (mean bottom depth of 6.6 m) along the south-west corner of the bay, in small groups of approximately three individuals. Common dolphins were the least observed species, and were mainly observed foraging in large groups of up to 800 individuals.

The results of this study indicate how cetaceans utilise the bay in significantly different ways. Geographical and anthropogenic factors have influenced the spatial and temporal distribution of these animals and have resulted in habitat preferences, as well as potential key habitats, in the bay Thus, this study has provided baseline information for future research and for better informed conservation and management strategies in Algoa Bay.

DECLARATION

The data obtained in this research was carried out under the auspices of a three year project on the spatial and temporal distribution of cetaceans in Algoa Bay run by Dr Stephanie Plön. The work presented in this thesis was carried out between February 2009 and January 2011, under the supervision of Ms Gillian McGregor (Geography Department, Rhodes University), in fulfilment of the academic requirements for the degree of Master of Science in Geography. This study represents original work by the author and is in accordance with the Rhodes University plagiarism policy.

TABLE OF CONTENTS

ABSTRACT	II
DECLARATION	IV
TABLE OF CONTENTS	V
LIST OF TABLES	X
LIST OF FIGURES	XI
LIST OF PLATES	XVII
ACKNOWLEDGEMENTS	XVIII
ACRONYMS	XIX
CHAPTER 1: GENERAL INTRODUCTION	1
1.1. Introduction	1
1.2. Summary of the study site: Algoa Bay	1
1.3. Motivation	1
1.4. Aim	
1.5. Research objectives and questions	
1.6. Research design and methods	4
1.7. Thesis outline	6
CHAPTER 2: LITERATURE REVIEW	7
2.1. Introduction	7
2.2. Zoogeography	7
2.3. Role of cetaceans in aquatic ecosystems	
2.4. The concept and importance of critical habitats	9
2.5. Review of cetacean habitat studies	11
2.6. Marine mammal surveys	12
2.7. Applying GIS in mapping distribution patterns	14
2.8. Outline of Cetacea	16
2.8.1. Introduction	
2.8.2. Mysticetes	17
2.8.3. Odontocetes	25
2.9. Summary and Conclusion	

CHAPTER 3: STUDY AREA	
3.1. Introduction	
3.2. Geographical features	
3.3. Climate and weather patterns	
3.4. Hydrological features	
3.5. The Biota associated with habitats in Algoa Bay	
3.5.1. Sandy beaches	
3.5.2. Rocky shores	
3.5.3. Reefs	
3.5.4. Marine Protected Areas	
3.5.5. The Sardine Run	
3.6. Anthropogenic activities	
3.6.1. Industry	
3.6.2. Commercial exploitation of marine resources	
3.6.3. Recreational and tourism activities	
3.6.4. Conservation strategies	
3.7. Conclusion	49
CHAPTER 4: METHODS	
4.1 Introduction	

4.1. Introduction	
4.2. Objective one (A): Dedicated boat-based surveys	
4.2.1. Introduction	
4.2.2. Survey design	
4.2.3. Cetacean sightings data	54
4.3. Objective one (B): Opportunistic data	57
4.4. Objective two	58
4.5. Objectives three and ffour	59
4.5.1. Introduction	59
4.5.2. Overview of the environmental parameters	59
4.5.3. Search effort	60
4.5.4. Generating a grid for Algoa Bay	60
4.5.5. Overview of sightings	61
4.5.6. Spatial sightings per unit effort	
4.5.7. Temporal sightings per unit effort	

4.5.8. Group dynamics and relative density	63
4.5.9. Observed behaviour of the cetaceans	64
4.5.10. Cetacean relative densities	64
4.5.11. Sightings and environmental variables	64
4.5.12. Sightings and anthropogenic variables	66
4.6. Sighting rate and relative abundance modelling	66
4.6.1. Exploratory data analysis	67
4.6.2. Generalised models	67
4.6.3. Hurdle models	68
4.7. Defining key habitats for cetaceans in Algoa Bay	69
4.8. Conclusion	70

CHAPTER 5: RESULTS	71
5.1. Introduction	
5.2. Overview of environmental parameters recorded	71
5.3. Search effort	71
5.4. Cetacean sightings	73
5.4.1. Overview of sightings	73
5.4.2. Spatial distribution of sightings	
5.4.3. Spatial sightings per unit effort (SPUE)	
5.4.4. Temporal sightings per unit effort (SPUE)	
5.5. Group dynamics and relative density	
5.6. Observed behaviour of the cetaceans	
5.7. Cetacean relative densities	
5.8. Sightings and environmental variables	
5.9. Sightings and anthropogenic activities	
5.10 Sighting rate and relative abundance modelling	
5.10.1. Introduction	
5.10.2. Southern right whales	
5.10.3. Humpback whales	
5.10.4. Bryde's whales	
5.10.5. Bottlenose dolphins	
5.10.6. Humpback dolphins	
5.10.7. Summary of the statistical models	

5.11. Defining potential key habitats for cetaceans in Algoa Bay	112
5.12. Summary of findings	
CHAPTER 6: DISCUSSION	117
6.1. Introduction	
6.2. Southern right whale	
6.2.1. General spatial and temporal distribution patterns	
6.2.2. Group dynamics and behaviour	
6.2.3. Description of the habitat preference within Algoa Bay	
6.3. Humpback whale	
6.3.1. General spatial and temporal distribution patterns	
6.3.2. Group dynamics and behaviour	
6.3.3. Description of the habitat preference within Algoa Bay	
6.4. Bryde's whales	
6.4.1. General spatial and temporal distribution patterns	
6.4.2. Group dynamics and behaviour	
6.4.3. Description of the habitat preference within Algoa Bay	
6.5. Bottlenose dolphin	
6.5.1. General spatial and temporal distribution patterns	
6.5.2. Group dynamics and behaviour	
6.5.3. Description of the habitat preference within Algoa Bay	132
6.6. Humpback dolphin	
6.6.1 General spatial and temporal distribution patterns	
6.6.2. Group dynamics and behaviour	
6.6.3. Description of the habitat preference within Algoa Bay	
6.7. Common dolphin	
6.7.1. General spatial and temporal distribution patterns	
6.7.2. Group dynamics and behaviour	
6.7.3. Description of the habitat preference within Algoa Bay	
6.8. Anthropogenic influences and impacts on cetaceans in Algoa Bay	
6.9. Defining potential key habitats for the cetaceans in Algoa Bay	
6.10. Implications for conservation and management	
6.11. Conclusion	145

CHAPTER 7: CONCLUSIONS AND RECOMMENDATIONS	147
7.1. Introduction	
7.2. Potential caveats	
7.3. Conclusions	148
7.4. Recommendations for mangement, conservation and research	
APPENDICES	156
APPENDICES Appendix 1: Effort form	156 156
APPENDICES Appendix 1: Effort form Appendix 2: Sighting record	156 156 157
APPENDICES. Appendix 1: Effort form. Appendix 2: Sighting record. Appendix 3: Maps of group sizes.	156
APPENDICES. Appendix 1: Effort form. Appendix 2: Sighting record. Appendix 3: Maps of group sizes. Appendix 4: Maps of individuals per unit effort (IPUE).	156

REFERENCES	165
Personal communications	

LIST OF TABLES

Table 2.1. Summary table of the cetaceans found within the inshore areas of Algoa Bay33
Table 4.1. Research Matrix 51
Table 4.2. Definitions for cetacean sightings
Table 4.3. Behavioural terms for mysticetes and odontocetes
Table 4.4. Description of the spatial data files used in this research. 'Manipulation' indicates whether further processing was done on the file in preparation for spatial analysis
Table 5.1. Summary of sightings for the six cetacean species observed in Algoa Bay. The results are based on the dedicated boat-based surveys unless otherwise stated
Table 5.2. Group dynamics and relative density of the cetaceans in Algoa Bay
Table 5.3. The significant differences in mean depths (white blocks) and mean distances from land (grey blocks), between the different cetacean species
Table 5.4. Comparison of the variables included in the final sighting rate models (GAMs and GLMs) for each species. Linear fits are represented by 'L 1', smoothing splines by 'S #', and polynomial fits by 'P #', where '#' indicates the associated degrees of freedom. The pseudo R ² value represents a goodness-of-fit measure. P-values were displayed if some level of significance was shown: 0 '***' 0.001 '**' 0.01'*' and 0.05 '.'
Table 5.5. Outcomes of the hurdle models for the different cetacean species. Only variables that were significant in the two stages of the model were noted with their appropriate levels of significance (0 '***' 0.001 '**' 0.01'*' and 0.05 '')
Table 6.1. Outline of the anthropogenic threats facing cetaceans in Algoa Bay
Table 6.2. A summary of the key findings associated with each of the cetacean species

LIST OF FIGURES

Figure 1.1. Research design
Figure 2.1. Worldwide distribution of the southern right whale (Reilly <i>et al.</i> , 2008b)19
Figure 2.2. Worldwide distribution of the humpback whale (Reilly <i>et al.</i> , 2008c)22
Figure 2.3. Worldwide range of Bryde's whales (Reilly <i>et al.</i> , 2008a)24
Figure 2.4. Worldwide range of Indo-Pacific bottlenose dolphins (Hammond <i>et al.</i> , 2008b)27
Figure 2.5. Worldwide distribution of Indo-pacific humpback dolphins (Reeves <i>et al.</i> , 2008)30
Figure 2.6. Worldwide distribution of common dolphins (Hammond <i>et al.</i> , 2008a)32
Figure 3.1. Features of Algoa Bay on the south Eastern Cape coastline of South Africa
Figure 3.2. Interpolated sediment map of Algoa Bay derived from SANHO point sample data using Thiessen polygons (note: sparse data in the south-west region has led to the construction of large Thiessen polygons)
Figure 3.3. Mean monthly temperatures for all three stations in Algoa Bay, and mean monthly precipitation for Port Elizabeth. Temperatures recorded at PE airport, Coega Port and Bird Island
Figure 4.1. Outline of survey tracks performed in Algoa Bay. Tracks 1, 2 and 3 are described in the text. Dashed lines represent opportunistic tracks covered during surveys
Figure 5.1. Beaufort sea state and cloud cover measured at sea for the duration of the study. Dashed lines represent the mean sea state (Beaufort 2) and cloud cover (5/8)72
Figure 5.2. Sea surface temperature (SST) recorded at sea for the duration of the study. Dashed line represents the overall mean SST of 19.2 °C
Figure 5.3. Survey tracks completed during boat-based surveys. A one square kilometre grid was used, displaying the total distance of tracks covered within each block, to illustrate intensity of effort. Track distance also includes the paths covered in sightings. Moran's I test detected no spatial autocorrelation ($p < 0.05$)
Figure 5.4. Sightings in Algoa Bay between March 2009 and July 2010. Sightings observed opportunistically are also displayed

Figure 5.5. Distribution of sightings for each species observed in Algoa Bay March 2009 to July 2010. Data collected during dedicated boat-based surveys is represented by circles and Figure 5.6. Spatial distribution of cetacean SPUE. Four levels of cells were used: empty cells, where search effort was carried out without any sightings recorded, and three coloured cells to represent low, medium and a high SPUE. Note: low, medium and high SPUE is a relative value defined for each species and is therefore not directly comparable across the different species....79 Figure 5.8. Spatial distribution of humpback whale SPUE......80 Figure 5.11. Spatial distribution of humpback dolphin SPUE......81 Figure 5.12. Sightings per 100 km search effort (SPUE) for all species, and search effort (SE) per month, for the duration of the study. Dashed line represents the mean SPUE (8.6) and the mean Figure 5.13. Sightings per 100 km effort (SPUE) for each species, and search effort per month. Dashed line represents the mean SPUE......84 Figure 5.14. The locations of mother-calf pairs for southern right whales and humpback Figure 5.15. Behavioural budget for each cetacean species. Note: 'mating' behaviour in bottlenose dolphins was classed as socialising, as these behaviours were closely related in the Figure 5.19. Kernel density map of the six cetacean species observed in Algoa Bay. Density estimates take into account the number of individuals in a sighting (i.e. values represent the number of individuals per square kilometre)......90 Figure 5.20. Kernel density map of the southern right whales in Algoa Bay. Density estimates

Figure 5.21. Kernel density map of the humpback whales in Algoa Bay. Density estimates take into account the number of individuals in a sighting
Figure 5.22. Kernel density map of the Bryde's whales in Algoa Bay. Density estimates take into account the number of individuals in a sighting
Figure 5.23. Kernel density map of the bottlenose dolphins in Algoa Bay. Density estimates take into account the number of individuals in a sighting
Figure 5.24. Kernel density map of the humpback dolphins in Algoa Bay. Density estimates take into account the number of individuals in a sighting
Figure 5.25. The median sea surface temperature (SST) and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles. Environmental (ENV) data was data collected both 'on' and 'off' effort during boat-based surveys
Figure 5.26. The median bottom depth and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles. Environmental (ENV) data was data collected both 'on' and 'off' effort during boat-based surveys
Figure 5.27. The proportion of depth classes surveyed in Algoa Bay
Figure 5.28. The log distance from land, and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles
Figure 5.29. Mean bottom depth (m) for two mysticete species comparing sightings of mother- calf pairs and non mother-calf pairs
Figure 5.30. Mean distance from land (km) for two mysticete species comparing sightings of mother-calf pairs and non mother-calf pairs
Figure 5.31. The log distance from rivers and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles
Figure 5.32. The proportion of sea-floor substrate associated with each species. Substrate data was made available from SANHO
Figure 5.33. Proportion of cetacean sightings within the shipping zones100
Figure 5.34. Proportion of mother-calf pairs inside and outside the proposed GAENP MPA (numbers in histogram represent actual number of sightings)100
Figure 5.35 Scatterplots showing the relationship between southern right whole group size

Figure 5.35. Scatterplots showing the relationship between southern right whale group size (BestEst) and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance

Figure 5.37. Scatterplots showing the relationship between humpback whales and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river...104

Figure 5.39. Scatterplots showing the relationship between Bryde's whales and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river...106

Figure 5.41. Scatterplots showing the relationship between bottlenose dolphins and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river...108

Figure 5.43. Scatterplots showing the relationship between humpback dolphins and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river...110

Figure 5.45. Potential key habitats (A – G) for the mysticetes in Algoa Bay.....114

Figure 5.46. Potential key habitats (H – L) for the odontocetes in Algoa Bay.....114

Figure 6.1. Diagram illustrating the distribution of southern right whales in southern Africa. The grey area represents the previously known limits (Baia dos Tigres to Maputo Bay). The blue and

Figure 6.3. Postulated movement of humpback whales in the Algoa Bay region......126

Figure 6.4. Proposed movement of Bryde's whales along South Africa.....128

Figure A4.2. Spatial distribution of southern right whale IPUE......161

Figure A4.3. Spatial distribution of humpback whale IPUE	
Figure A4.4. Spatial distribution of Bryde's whale IPUE	162
Figure A4.5. Spatial distribution of bottlenose dolphin IPUE	163
Figure A4.6. Spatial distribution of humpback dolphin IPUE	163
Figure A4.7. Spatial distribution of common dolphin IPUE	164

LIST OF PLATES

Plate 2.1. The head of a southern right whale. <i>Photo: B. Melly</i>	19
Plate 2.2. Humpback whale breaching (Cape Padrone in background). Photo: B. Melly	22
Plate 2.3. Bryde's whale. Photo: B. Melly	24
Plate 2.4. Bottlenose dolphins leaping out the back of a wave (Alexandria Dunefield	1 in
background). Photo: B. Melly	27
Plate 2.5. Humpback dolphin. Photo: B. Melly	30
Plate 2.6. Common dolphins porpoising. Photo: B. Melly	32

ACKNOWLEDGEMENTS

I would like to thank Marine and Coastal Management and the South African Environmental Observation Network (SAEON) for financial and logistical support. I am also grateful for the personal funding from the National Research Foundation (NRF) and the Andrew Mellon Scholarship. Thanks also to Rhodes University and the European Cetacean Society (ECS) for providing funding to present at a conference in Stralsund, Germany. I am grateful to PE Museum/ Bayworld for providing a base from which to conduct fieldwork. I would also like to thank the South African National Hydrographic Office (SANHO), the South African Weather Service (SAWS) and South African National Parks (SAN Parks) for access to data.

To Gillian McGregor, my supervisor, I am extremely grateful for your continued support, encouragement and constructive criticism throughout my masters. I really appreciate everything you have done for me, and for always believing in me. I would like to extend my gratitude to Dr Stephanie Plön for designing this project and taking me on as a student during the fieldwork phase of this research. I would like to thank Dr Angus Paterson for his input, logistical support, and guidance throughout the project. To Morven Maclean, who skippered and towed the boat when an extra hand was needed, thank you for your assistance. I am grateful to Dot Pitman for always going an extra mile to help me find valuable resources during my research. To the many, many volunteers who assisted on the cetacean surveys, thank you for all your help and your smiling faces early in the morning, and for enduring the long (and sometimes cold and wet) boat days.

I am grateful to everyone who has contributed to the final stages of this dissertation, especially Prof. Tony Booth for his help in the statistical analyses. I would also like to thank Leanne Kelly for her input and advice. To all the staff and students in the Rhodes University Geography Department, thank you for your input and assistance over these years, I really appreciate it.

To my parents and family, thank you for always supporting my many different endeavours and for encouraging me to pursue my dreams, I am deeply grateful. To my Grahamstown and PE friends - thank you for always supporting me and for constantly challenging me, I am incredibly privileged to know so many wonderful people. I would especially like to mention Shan Ambrose, Pearl Mzobe, Tarryn Govoni and Riana van der Sandt, I sincerely value your friendship and support. Finally, my greatest thanks go to my Father in Heaven for creating such an incredibly beautiful world, and for allowing me the privilege of learning more about it.

ACRONYMS

ANOVA	Analysis of Variance
CES	Coastal and Environmental Services
Coega IDZ	Coega Industrial Development Zone
Cos	Mathematical function 'cosine'
ESRI	Environmental Systems Research Institute
GAENP	Greater Addo Elephant National Park
GAMs	Generalised Additive Models
GIS	Geographical Information Systems
IUCN	International Union for the Conservation of Nature
IWC	International Whaling Commission
kn	Knots
KZN	Kwa-Zulu Natal
MPA	Marine Protected Area
nd	No data
II. u .	No uale
nm	Nautical mile
nm NMBM	No date Nautical mile Nelson Mandela Bay Municipality
n.u. nm NMBM PE Port	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port
nm NMBM PE Port SAEON	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network
nm NMBM PE Port SAEON SAGs	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups
nm NMBM PE Port SAEON SAGs SANHO	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups South African National Hydrographic Office
nm NMBM PE Port SAEON SAGs SANHO SAN Parks	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups South African National Hydrographic Office South African National Parks
nm NMBM PE Port SAEON SAGs SANHO SAN Parks SAWS	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups South African National Hydrographic Office South African National Parks South African Weather Service
nm NMBM PE Port SAEON SAGs SANHO SAN Parks SAWS Sin	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups South African National Hydrographic Office South African National Parks South African Weather Service Mathematical function 'sine'
nm NMBM PE Port SAEON SAGs SANHO SAN Parks SAWS Sin SPUE	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups South African National Hydrographic Office South African National Parks South African Weather Service Mathematical function 'sine' Sightings per unit (100 km) effort
nm NMBM PE Port SAEON SAGs SANHO SAN Parks SAWS Sin SPUE STEP	No date Nautical mile Nelson Mandela Bay Municipality Port Elizabeth Port South African Environmental Observation Network Surface active groups South African National Hydrographic Office South African National Parks South African Weather Service Mathematical function 'sine' Sightings per unit (100 km) effort Science, Technology and Environment Programme

"The scientist does not study nature because it is useful; He studies it because he delights in it, and he delights in it because it is beautiful" ~ Henri Poncaré ~



Humpback whale breaching in Algoa Bay. Photo: B. Melly.

CHAPTER 1: GENERAL INTRODUCTION

1.1. INTRODUCTION

This chapter provides an introduction to the research conducted on the cetaceans in Algoa Bay between March 2009 and July 2010. A brief overview of the study site and motivation for this research will precede the establishment of the aims and objectives. A synopsis of the methodological approach to the study will be addressed before concluding with an outline of the structure of this dissertation.

1.2. SUMMARY OF THE STUDY SITE: ALGOA BAY

Algoa Bay is a large (3 100 km²), moderately exposed bay off the south-east coast of South Africa, with a coastline length of approximately 135 km between two headlands, Cape Recife and Cape Padrone (see Chapter Three for a detailed map). The Sundays, Swartkops and Coega Rivers are the three major rivers, all entering the western half of the bay.

Port Elizabeth city and the port are situated on the south-west corner of Algoa Bay, one of two major developed areas bordering the bay. The second developed area is the Coega Industrial Development Zone (IDZ) and newly constructed deep-water Ngqura (Coega) Port situated 20 km north of Port Elizabeth Port (PE Port). These two ports are used extensively for both recreational and commercial purposes.

Three marine conservation zones are also situated in the bay. They are comprised of the Bird Island and St Croix Island groups which are Marine Protected Areas (MPAs), and the larger proposed greater Addo Elephant National Park (GAENP) MPA situated between Coega Port and the Bird Island group. These conservation zones and the developed areas highlight some of the contrasting needs and uses of Algoa Bay, which are discussed in detail in Chapter Three.

1.3. MOTIVATION

On a global scale, some cetacean stocks are vastly below their original numbers due to historical whaling (Friedmann and Daly, 2004). The IUCN and the Red Data List status of South Africa notes that a number of cetacean species found off the South African coastline are currently

considered as 'threatened', 'vulnerable', 'endangered' or 'data deficient' (Friedmann and Daly, 2004; Hammond *et al.*, 2008a; Hammond *et al.*, 2008b; Reeves *et al.*, 2008; Reilly *et al.*, 2008a). This includes the cetaceans found in the Algoa Bay region, illustrating the need for further research on these species.

Globally, there are a variety of threats likely to face cetacean populations in the near future, which have been described in detail by several authors (see: Richardson *et al.*, 1995; Nowacek *et al.*, 2007; Simmonds and Isaac, 2007; MacLeod, 2009). In Algoa Bay, these threats include: increased shipping traffic from the Coega development, which will be fully operational in the next five years, ship strikes (collisions with vessels), entanglement in fishing gear, the associated risks with offshore oil and gas development, coastal pollution (industrial and urban sources), a rise in recreational boating (particularly powerboats and jet skis), the destruction or alteration of coastal habitats, overfishing and climate change (Klinowska, 1991; Best *et al.*, 2001a; Friedmann and Daly, 2004).

Recently there has been increased interest in boat-based whale watching in Algoa Bay, with a permit set to be released in 2010/ 2011 (Government Gazette, 2009a; Government Gazette, 2009b). Worldwide, whale-watching is on the increase, however, little is presently known about the short- or long-term effects of this activity on cetaceans (Wilson *et al.*, 1997; Constantine *et al.*, 2004; Hoyt, 2005; Bejder *et al.*, 2006). A long-term study by Bejder *et al.* (2006), illustrated that the primary contributor to the decline in dolphin abundance in Shark Bay, Australia, was associated with harassment by vessels, and that future management decisions need to take into account these well-documented long-term studies in order to maintain healthy cetacean populations.

The proposed MPA, which would form part of the GAENP owned by SAN Parks (Coastal and Environmental Services, n.d.), requires baseline information on which cetaceans visit MPAs, and their distribution patterns and habitat preference, as it affects how the MPAs will be controlled and managed in the future (Hooker *et al.*, 1999; Cañadas *et al.*, 2005). Determining which species utilise the MPA may assist in elucidating potential threats to the biodiversity. These need to be determined as cetaceans are considered to be an important indicator of ecosystem health, which is required to be at a high level in and around the MPAs (Bowen, 1997; Hooker and Gerber, 2004; Cañadas *et al.*, 2005).

The most recent study on cetaceans in Algoa Bay was conducted over 14 years ago by Karczmarski (1996) on humpback and bottlenose dolphins. Thus, most current knowledge of the cetacean species visiting the bay is based on incidental observations and stranding data (discussed further in Chapter Two). This deficiency in research has resulted in a lack of data pertaining to the distributional patterns of cetaceans in the region. Many anthropogenic and biological factors impact the abiotic and biotic components of this marine ecosystem, and these impacts are expected to increase in the near future, stressing the importance of conducting a baseline survey of these keystone species.

1.4. AIM

The aim of this research is to determine the spatial and temporal distribution and habitat preference, of the cetaceans in Algoa Bay. This includes areas within and outside the proposed MPA. In addition, this study aims to relate this distribution to geographical parameters.

1.5. RESEARCH OBJECTIVES AND QUESTIONS

In order to achieve the aim of this thesis, five objectives were defined with their associated research questions (R.Q.), which are described below. The methods associated with these objectives are discussed in detail in Chapter Four.

Objective 1: To collect primary data on cetacean location, behaviour and other related geographical variables via different survey and opportunistic techniques.

R.Q. Which cetaceans are found in Algoa Bay?

Objective 2: To acquire secondary spatial and non-spatial data on geographical and anthropogenic variables.

R.Q. Can various geographical and anthropogenic variables be used to explain cetacean distribution (in objective four)?

Objective 3: To determine the spatial and temporal distribution and behaviour of the different cetacean species in Algoa Bay.

R.Q. Where (spatial) and when (temporal) are cetaceans located in Algoa Bay? *R.Q.* Where are key behaviours observed, and do the behaviours of the animals help explain cetacean distribution?

Objective 4: To relate the spatial and temporal distribution and behaviour of cetaceans to the geographical and anthropogenic variables determined in objective two.

R.Q. Are there associations between cetacean distribution and the surrounding geographical and anthropogenic variables, identified in objective two? R.Q. Are there potential key habitats for each of the cetacean species in Algoa Bay?

Objective 5: To produce a set of recommendations and guidelines, using the maps and information produced, for improved management, conservation and research.

R.Q. How can the outcomes of this research be used to form a baseline for future research, conservation and development strategies?

1.6. RESEARCH DESIGN AND METHODS

The processes that were employed in this dissertation are summarised in Figure 1.1. A variety of methods were used to meet the different objectives of this study. A literature review formed the foundation of this research in order to determine appropriate aims and objectives, as well as the choice of methods used to collect the data produced (Figure 1.1). Data was obtained quantitatively through dedicated boat-based surveys and opportunistic sightings, with additional environmental data collected from secondary sources. Various spatial and statistical analyses were performed on the data, after which the results were discussed and compared to other relevant studies, using the literature acquired in the review.

The marked growth in information technology over the last two decades has resulted in the development of new graphical and statistical techniques. More recently, the integration of GIS tools and environmental models has improved the analysis of species distributions in relation to the environment, and thus made a significant contribution to the field of applied ecology and biogeography. An understanding of the effectiveness of these tools and the need for baseline data

on cetaceans in the Algoa Bay region, has led to the formation of the aim and objectives for this study. A detailed explanation of these methods used in this research is described in Chapter Four.



Figure 1.1. Research design.

1.7. THESIS OUTLINE

Chapter One provides an introduction to the study by establishing the context and rationale of this research. It ends with the aims, objectives, and the research questions, which form the base of the research matrix from which this study is designed.

Chapter Two provides a review of literature pertaining to the theoretical framework of this thesis. Included in this chapter is an examination of studies that have contributed to the methodological approach of this dissertation. An overview of some of the cetacean species previously recorded in the Algoa Bay region is also provided.

An overview of the geographical, hydrological and biological features of the study area, are outlined in Chapter Three. In order to establish the context of this study, the overall weather patterns and several development and conservation strategies in Algoa Bay are also discussed.

Chapter Four presents a detailed examination of the methods used to achieve the aforementioned objectives. It expounds on the methodology behind the chosen procedures for data acquisition, and describes how this data was processed and analysed using the relevant tools for spatial and statistical data analysis.

Chapter Five delineates the results of the analyses performed on the data obtained from this research. These results are presented in various formats, including tables, graphs and maps, thereby fully describing the spatial and temporal distribution and habitat preference of the cetaceans in Algoa Bay.

Chapter Six discusses the results of the spatial and temporal distribution of the cetaceans in Algoa Bay. These results were integrated with the theory detailed in Chapter Two, and the knowledge of the study area described in Chapter Three. This chapter concludes with an overview of the critical habitats for the cetaceans, and the implications for conservation and management.

Chapter Seven concludes this dissertation by providing a synthesis of the outcomes of this research, based on the aim and objectives established in Chapter One. Furthermore, it outlines the limitations of this study and provides recommendations for future research, conservation and management.

2.1. INTRODUCTION

This chapter reviews the literature pertaining to the theoretical framework of this thesis. It begins by defining the concept of zoogeography and how it forms the basis of this study. The review then examines the role of cetaceans in their aquatic environment and the importance of defining this role through determining areas of high use (critical habitats) for each cetacean species. This research is also based on case studies of previous habitat studies and marine mammal surveys, outlined in this chapter. This review will outline how GIS has played a crucial role in increasing the understanding of the complex spatial and temporal relationships that cetaceans have with their marine environment. Current knowledge of the cetacean species studied in this research is also discussed towards the end of this chapter.

2.2. ZOOGEOGRAPHY

Explaining, quantifying and monitoring patterns in population distribution is a fundamental aspect of understanding a species' ecology in a particular environment (Evans and Hammond, 2004). Identifying and monitoring these changes is the essence of conservation research (Evans and Hammond, 2004). Patterns in the spatial and temporal distribution of animals enables scientists to determine whether there are areas of animal concentration in relation to various anthropogenic activities, that can form key focus areas for conservation efforts (Evans and Hammond, 2004). This forms the foundation for the field of zoogeography.

Zoogeography is a branch of the science of biogeography that is concerned with the geographic distribution of animal species (Brown and Lomolino, 1998). It studies the past, present and future patterns of animal distribution (and their attributes) in the natural environment and the processes that affect these distributions (Brown and Lomolino, 1998). Most geographical locations have information on the presence or absence of cetacean species, but a large gap still remains with regards to their basic biology, life history and distribution along much of the world's coastlines (Mignucci-Giannoni, 1998; Parra *et al.*, 2006). Algoa Bay in the Eastern Cape Province in South Africa is no exception, with four fundamental questions that need to be answered for this area:

Which cetacean species are found in the bay?What is their spatial distribution?What is their temporal distribution?What environmental parameters determine their distribution?

This study will attempt to answer the first three questions, forming a basis from which the fourth question can be answered through defining critical habitats/ areas. These questions will be further examined in Chapters Four to Seven.

2.3. ROLE OF CETACEANS IN AQUATIC ECOSYSTEMS

Marine mammals are large and abundant in the oceans, living and feeding in almost every part of the aquatic environment from rivers, estuaries and continental shelves, to tropical and polar waters (Katona and Whitehead, 1988; Bowen, 1997). They interact with a wide variety of organisms in these different systems, playing an important role in the structure and function of various communities in the aquatic environment (Katona and Whitehead, 1988; Bowen, 1997).

Cetaceans are major consumers on most trophic levels (Bowen, 1997; Hoyt, 2005). Being longlived, they can be used as indicators of ecosystem health and productivity, reflecting the effects of human and natural factors on lower trophic levels in the marine environment (Hooker and Gerber, 2004; Wells *et al.*, 2004). Cetaceans are also used as indicators due to the many feeding associations that exist between cetaceans and various bird and fish species (Katona and Whitehead, 1988). Their diverse role in the marine environment provides a context within which the impacts of environmental change (both anthropogenic and natural) can be understood at both large and small scales (Bowen, 1997; Hoyt, 2005).

These long-lived consumers can also influence prey dynamics through their responses to changes in prey density by fasting, migrating or changing their diet (Katona and Whitehead, 1988; Bowen, 1997). Where a substantial overlap occurs, this has resulted in competition between commercial fisheries and small cetaceans, with most studies indicating that the impact is felt more by cetaceans than fisheries (Sekiguchi *et al.*, 1992; Young, 2000; Kaschner *et al.*, 2006a). An example of this is the dramatic decrease in numbers of the formerly abundant common dolphin (*Delphinus delphis*) in the Mediterranean Sea as a result of the depletion of small pelagic fish stocks (Bearzi *et al.*, 2006).

The ecological role of cetaceans in aquatic ecosystems is still poorly understood despite the value of cetaceans to humans and the environment (Bowen, 1997). While some cetacean populations are recovering at rapid rates, others are becoming endangered or extinct (Van Blaricom, 2000). Thus, long-term, interdisciplinary research is required to cover a wide range of spatial and temporal scales, in order to fully understand their role in marine ecosystems (Wilson *et al.*, 1997; Bearzi *et al.*, 2009).

2.4. THE CONCEPT AND IMPORTANCE OF CRITICAL HABITATS

A home-range is the area used by an individual or population to perform various activities such as foraging, mating, breeding and calving (Shane *et al.*, 1986; Ingram and Rogan, 2002). There are vast differences in cetacean distribution patterns, with some populations resident within confined areas, and others, nomadic or migratory (Wilson *et al.*, 1997; Rowntree *et al.*, 2001; Campbell *et al.*, 2002). These differences are influenced by factors such as the biological requirements of a species and habitat heterogeneity, both of which determine patterns of distribution and use of critical habitats within the home range of a species (Ballance, 1992).

Interactions between cetaceans and their habitat are complex, and occur at various spatial and temporal scales, resulting in high spatial heterogeneity within their home-range (Kenney and Winn, 1986; Reilly, 1990; Allen *et al.*, 2001). This is influenced by geographic conditions such as SST, bottom depth, distance to land, tidal flow, currents, fronts, upwellings, salinity, chlorophyll, bottom substrate, and bottom topography; as well as anthropogenic disturbances including noise, pollution and shipping traffic (Smith *et al.*, 1986; Selzer and Payne, 1988; Richardson *et al.*, 1995; Mignucci-Giannoni, 1998; Davis *et al.*, 2002; Bräger *et al.*, 2003; Picanço *et al.*, 2009; Weir *et al.*, 2009). These geographical (physiographic and hydrographic) and anthropological features result in distinct core areas within their home-range (Davis *et al.*, 1998).

These core areas are known as critical habitats, which are defined as areas of high use or parts of an animal's range that are vital for survival and maintenance of a healthy population (Harwood, 2001; Ingram and Rogan, 2002; Hoyt, 2005). For cetaceans, examples of critical habitats are areas of high concentrations of prey for foraging, or where they avoid predation (Hanson and Defran, 1993; Keiper *et al.*, 2005; Sironi *et al.*, 2008). Areas where high concentrations of prey are located include regions where there are significant structural features such as steep gradients or complex topographies (shelf edges and continental slopes), which may increase the efficiency

of prey detection through prey aggregation (Selzer and Payne, 1988; Baumgartner, 1997; Robinson *et al.*, 2009).

Areas where cetaceans perform important activities such as mating, calving, raising offspring, and resting, are also considered to be critical habitats (Elwen and Best, 2004; Lusseau and Higham, 2004; Stensland *et al.*, 2006). Migratory whale species use suitable habitats such as shallow protected bays and soft substrate, found in areas on the continental shelf along the South African coastline, for rearing their young (Best *et al.*, 2001a; Elwen and Best, 2004). Bays provide shelter from strong currents and greater protection from predators, enabling calves to utilise the energy obtained from suckling for growing, rather than moving, so that migration to the feeding grounds in the Antarctic region can happen as soon as possible (Thomas and Taber, 1984; Best and Rüther, 1992; Elwen and Best, 2004).

The identification and definition of a critical habitat is necessary for the establishment of spatially appropriate conservation plans which can be used to delineate marine protected areas (MPAs) (Hastie *et al.*, 2003; Cañadas *et al.*, 2005). As marine mammals have complex interactions within their environment, information needs to be collected continuously on cetacean distribution, at various scales, in order to design and manage effective MPAs (Hastie *et al.*, 2003). However, MPAs should protect an ecosystem rather than an individual species, and in theory, should take into account that boundaries in the marine environment are fluid and dynamic (Harwood, 2001; Hooker and Gerber, 2004).

Management strategies are dependent on both intensity of habitat use and location (Kenney and Winn, 1986; Ingram and Rogan, 2002). As each individual cetacean population uses various habitats within their range at different intensities, site-specific studies are necessary to determine habitat use before recommendations are made regarding coastal conservation and management (Kenney and Winn, 1986; Guinet *et al.*, 2001; Ingram and Rogan, 2002). Most marine management decisions are based on physiographic features as they are more stable over time. However, predator-prey interactions and other influences that key indicator species (such as cetaceans) have on their environment must also be taken into account, despite their temporal and spatial variability (Cañadas *et al.*, 2005; Robinson *et al.*, 2009).

Resources in the marine environment are patchy and largely hidden from view. Consequently, a population cannot necessarily move to a new area if an important habitat is being negatively

impacted on by humans activities (Arthur *et al.*, 1996; Wilson *et al.*, 1997). For this reason, to understand the social structure and zoogeography of these fauna, and define critical habitats for conservation strategies and effective marine management (for example, the establishment of MPAs), these areas need to be identified, well defined and understood (Wilson *et al.*, 1997; Hastie *et al.*, 2003; Hoyt, 2005; Cañadas and Hammond, 2008).

2.5. REVIEW OF CETACEAN HABITAT STUDIES

Cetacean studies have used a variety of geographical and biological parameters to help define the habitat preferences of a species (Davis *et al.*, 1998; Baumgartner *et al.*, 2001). This includes the use of variables such as bottom depth, SST and prey distribution, to delineate distribution patterns and therefore the habitat preferences within a study area (Allen *et al.*, 2001; Cañadas *et al.*, 2002). Three examples of studies that have used geographical features to define cetacean distribution and habitats are outlined below.

A large-scale study conducted by Cañadas *et al.* (2002), illustrated the use of two variables, depth and slope, to describe cetacean distribution in the Mediterranean waters off southern Spain. The distribution of all the species studied varied significantly, and could be classified into two groups: deep water and shallow water species, with the dividing depth line at 600 m. This classification was used to determine which habitats were important for the common prey species consumed by each species in a group, and therefore, where the key feeding areas were located. This study formed a baseline from which other studies in the region could analyse cetacean distribution incorporating other variables, in order to create MPAs and protect vulnerable species in the area.

Elwen and Best (2004) hypothesised that several variables would be associated with the discontinuous distribution of southern right whale (*Eubalaena australis*) mother-calf pairs compared to unaccompanied whales off the south coast of South Africa. The variables tested included: water calmness, sea-floor sediment, water depth, distance to land and slope of the sea-floor. It was found that unaccompanied whales were found in significantly deeper waters and further offshore than mother-calf pairs, and that mother-calf pairs were generally found off sandy beaches in more protected areas (calm waters). Through clarifying what the key habitats of the mother-calf pairs consisted of, Elwen and Best (2004) were able to hypothesise the potential long-term benefits of these animals using such habitats, leading to more defined and directed future studies.

Weir *et al.* (2009) focused on three habitat parameters to determine the fine-scale habitat selection of white-beaked dolphins (*Lagenorhynchus albirostris*) and short-beaked common dolphins (*Delphinus delphis*) in the Minch, Scotland. Water depth, SST and distance to land, were used to determine whether habitat partitioning occurs between these two species in the study area. This study revealed that both species illustrated a degree of spatial overlap, however, common dolphins appeared to be more widely distributed, in more shallow waters and closer to land than the white-beaked dolphins.

Basing conservation measures, such as the establishment of MPAs, on geographical variables is more manageable and more easily defined compared to using biological variables (for example, prey distribution), as these geographical variables tend to fluctuate less over space and time (Cañadas *et al.*, 2002; Viddi *et al.*, 2010). These studies illustrate the applicability of using habitat parameters to define both fine- and broad-scale cetacean distribution patterns.

2.6. MARINE MAMMAL SURVEYS

Conducting surveys in the marine environment poses unique problems in both data capture and data analysis. In order to use the most appropriate survey technique, several factors need to be considered such as the project objectives and outcomes, the size of the study area, the location and conditions of the study area, and the amount of logistical and financial support available (Aragones *et al.*, 1997; Evans and Hammond, 2004; Dawson *et al.*, 2008).

There are three primary methods for conducting surveys on marine mammals: land-based surveys, aerial surveys and ship/ boat-based surveys (Aragones *et al.*, 1997). Land-based surveys use observation points on land to provide basic information on the biology and ecology of marine mammals visiting the coastal areas of a study site (Bristow *et al.*, 2001; Evans and Hammond, 2004). This method has proven successful in areas such as New Quay Bay, Wales, where shore-based monitoring methods were used to observe group size, group structure and site fidelity of bottlenose dolphins (*Tursiops truncatus*) (Bristow *et al.*, 2001). Conversely, this method is not applicable in study areas where large stretches of coastline are isolated, as is the case in Algoa Bay.

Aerial surveys use low-flying aircrafts to carry out rapid line-transects over large areas (Aragones *et al.*, 1997; Best, 2000). In South Africa, coastal aerial surveys were conducted 300 – 600 m

offshore between Muizenberg and Woody Cape (on the south coast) to determine the coastal distribution, movements and site fidelity of southern right whales (Best, 2000). These aerial surveys yielded accurate information on population trends in the region (Best, 2000). However, this method is costly and requires strong logistical support (Aragones *et al.*, 1997).

Boat-based surveys allow for the investigation of spatial and temporal trends of a species, habitat use, the estimation of the relative abundance of a species (for long-term monitoring of population trends), and can also provide a platform for photo-identification studies (Aragones *et al.*, 1997). Many cetacean studies have successfully adopted this method for a wide-range of study areas which survey cetaceans (for example: Holt *et al.*, 1987; Barlow, 1995; Weir *et al.*, 2009; de Boer, 2010). However, there are several assumptions which are made when conducting these surveys, and they are also assumed in this study. These assumptions, described in detail by Aragones *et al.* (1997), Evans and Hammond (2004), Dawson *et al.* (2008) and Hammond (2009), are as follows:

1. All individuals/ groups are observed and accurately identified along a trackline

2. All individuals/ groups are observed only once within a survey and the animal does not move in response to the vessel before detection

3. Group size (and other variables) is measured or estimated accurately

There are large financial and logistical costs of performing equal-coverage sighting surveys (Dawson *et al.*, 2008). This, combined with the reality of conducting surveys where a good survey design was intended but not achieved, has lead to the development of spatial data analysis methods (see section 2.7) to overcome these problems (Dawson *et al.*, 2008; Hammond, 2009). Study areas which survey cetaceans using platforms of opportunity (Mignucci-Giannoni, 1998; Evans and Hammond, 2004), as well as areas where line transects were not possible but detailed environmental data were collected, have also applied similar methods of spatial analysis (for example: Cañadas *et al.*, 2005; Cañadas and Hammond, 2008; Dawson *et al.*, 2008). These methods include the development of spatial and statistical modelling, which have illustrated how the distribution and relative abundance of an individual is a function of the surrounding environmental conditions such as SST, bottom depth, sea-floor substrate, latitude and longitude (Evans and Hammond, 2004).

This research aims to use methods adapted from Cañadas *et al.* (2005), and Cañadas and Hammond (2008), where both used a non-systematic line-transect sampling method to conduct boat-based surveys. These studies collected sighting data and additional data on the physical and

environmental features, such as depth and slope of the sea-floor. The study area was divided into grid cells, with a cell resolution of two minutes, and was used to correct sighting data with the amount of search effort carried out within each cell. These two papers demonstrate how GIS and spatial analysis techniques can be used to compensate for these aforementioned problems of using unequal coverage survey designs. Data generated in this manner may be used to implement appropriate conservation and management strategies.

2.7. APPLYING GIS IN MAPPING DISTRIBUTION PATTERNS

Spatial data analysis refers to the analytical methods applied to raw geographical data (such as sighting locations) in order to produce useful spatial information (Booth, 2004; Longley et al., 2005). A key tool used to perform spatial data analysis is a geographic information system (GIS). GIS is a system of hardware, software and liveware that assembles, stores, manipulates and displays geographically referenced data (Longley et al., 2005; Maguire et al., 2005). GIS in the context of this research, was used to observe and examine the relationship between marine mammals and their environment by comparing animal locations and densities to both environmental variables (such as bottom depth and SST) and anthropogenic activities (Stensland et al., 2006; Norman, 2008; Vigness-Raposa et al., 2009). This helps to define distribution patterns and critical habitats, which ultimately leads to a better understanding of a species biogeography, in order to produce applicable and spatially appropriate marine management strategies and conservation measures (Stensland et al., 2006; Norman, 2008). GIS is consequently useful in analysing and interpreting data from marine mammal surveys through elucidating information on the abundance, behaviour and distribution of a particular species (Nelson et al., 2008; Nelson et al., 2009). Examples of studies that have used GIS to examine the relationship between the spatial distribution of marine mammal species and the surrounding environment are outlined below.

de Stephanis *et al.* (2008) examined the spatial association between various cetacean species based on their distribution over several seasons and their link to a certain habitat type (bottom depth and slope). This research took place in the Straight of Gibraltar and used the distribution of sightings and the associated cetacean behaviour to define their habitats. Although this study defines the habitats well, it does not extend to a discussion of the significance of having well-defined habitats for management and conservation purposes.

Kaschner *et al.* (2006b) took a similar approach to de Stephanis *et al.* (2008) in determining the spatial association of cetacean species, to certain habitat types. However, this idea was developed further by qualitatively and quantitatively defining the habitats used by marine mammals based on several environmental parameters, to determine their distribution at a regional scale. This distribution was then further modelled using statistics, to review the species distribution at a global scale. This was done through creating habitat suitability models (using spatial modelling and GIS) that quantitatively determined the maximum range of a species, and predicted the distributions of various cetacean species. The application of this was to assist in re-evaluating existing assumptions and knowledge on a species distribution, which is needed for direction in future research and management decisions.

Parra *et al.* (2006) used GIS and several statistical tests to construct a spatial database for three bays off the northern section of the Great Barrier Reef in Australia. Three geographical layers were used and combined with the geo-referenced sighting data: the coastline, river mouths and bottom depth. The correlation between these three environmental features and the two cetacean species, the snubfin (*Orcaella heinsohni*) and Indo-Pacific humpback dolphins (*Sousa chinensis*) were analysed, thereby outlining suitable and important areas for these two populations. Distinct and workable critical areas for coastal and marine managers were defined to establish whether the existing protected areas in the region were sufficiently protecting the critical habitats of these two dolphin species.

A GIS study by Robinson *et al.* (2009), in the outer southern Moray Firth, north-east Scotland, revealed a strong spatial preference of minke whales (*Balaenoptera acutorostrata*) for various geographical parameters. Bottom depths between 20 and 50 m, steep slopes with a north-facing aspect, and sandy-gravel sediment were preferred by this species, illustrating a highly specific and non-uniform use of their habitat that varied significantly both spatially and temporally. These parameters therefore provided a reference for deriving suitable conservation measures.

GIS can also be used to manage the impacts of dolphin-based tourism through defining critical habitats as demonstrated in a study by Lusseau and Higham (2004). These critical habitats were areas mostly used for resting or socialising by bottlenose dolphins. They were used to design a multi-level marine mammal sanctuary in Doubtful Sound, New Zealand, that took into account the economic sustainability of tourism operations while still maximising the protection of the dolphins throughout the different seasons.

Despite the application of GIS and statistics in spatial data analysis, studies on marine mammals have only used these tools to a limited extent (Nelson et al., 2009; Booth, A.J., 2010: Pers. Comm.). Nonetheless, GIS and statistics can be useful in determining the relationship between cetaceans and their environment, where these variables tend to occur at different spatial and temporal scales, and can be both static and dynamic (Booth, 2004; Robinson *et al.*, 2009). The studies mentioned in this review have used GIS as a tool to indicate the strength of the relationship between the ecology of cetaceans (i.e. their distribution, abundance and behaviour), and the surrounding habitat. Thus, GIS provides a cost-effective approach to improved monitoring and conservation of the marine environment, through collecting and storing a wide range of environmental and species data (Hooker *et al.*, 1999; Parra *et al.*, 2006; Robinson *et al.*, 2009).

GIS is therefore ideally suited to integrating data on biotic and abiotic variables, and presenting the data in a format suitable for further analysis of these relationships. Thus, the present project will draw on the methods and viewpoints from the studies reviewed to examine the spatial and temporal distribution of the cetaceans in Algoa Bay. These research findings will establish whether there is a link between the species distribution patterns and various geographical parameters. This will be used to assist in defining key habitats.

2.8. OUTLINE OF CETACEA

2.8.1. Introduction

There are approximately 83 species of cetaceans worldwide, with 51 species found in the coastal waters off southern Africa (Rice, 1998; Best, 2007). This high diversity off southern Africa can be attributed to the unique oceanographic conditions, which gives rise to a wide variety of oceanic conditions in a relatively small region (Ross and Best, 1989; Reilly *et al.*, 1997; Best, 2007).

Several key authors have contributed to the understanding of the cetacean species which visit the Eastern Cape coastline of South Africa. This knowledge, combined with stranding data and anecdotal evidence collected from biologists, whaling statistics and incidental sightings, have contributed to the acknowledgment of six species which potentially utilise the inshore areas of Algoa Bay.
Aerial surveys in the 1980s and 1990s observed the status, trends and site fidelity of southern right whales off the southern Cape coastline (Best, 1990; Best, 2000). Inshore and offshore forms of Bryde's whales were also observed during this period (Best, 1977; Best *et al.*, 1984; Best, 2001). Dwarf minke whales have been observed in the inshore areas off the west and east coast of South Africa, with no direct observations off the south coast (Best, 1985; Best, 2007). However, this could be due to this species being easily confused with Bryde's whales which are similar in appearance (Best, 2007). Humpback whales are known to occur world-wide, with the winter breeding grounds occurring in lower latitudes such as Angola and Mozambique (Findlay *et al.*, 1994; Findlay and Best, 1995). This species has also been recorded in other areas of the sub-region, including South Africa, utilising the coastal waters as a migratory corridor from the Antarctic feeding grounds to their tropical breeding grounds (Findlay *et al.*, 1994; Best and Sekiguchi, 1996; Best, 2007).

More extensive research/ reviews have been done on the smaller cetaceans off the south-east coast of South Africa (Saayman *et al.*, 1972; Ross, 1984; Findlay *et al.*, 1992). Observations of bottlenose dolphins, humpback dolphins and common dolphins were recorded in Algoa Bay in a study by Saayman *et al.* (1972). The presence and knowledge of these species have since been updated by numerous reviewers since (see: Ross, 1984; Cockcroft and Ross, 1989; Ross *et al.*, 1989; Cockcroft and Peddemors, 1990; Findlay *et al.*, 1992; Best, 2007; Reisinger and Karczmarski, 2009).

There are many debates on the taxonomy of several cetacean species/ sub-species, including those that have been addressed in this study. This research relies on species descriptions by Ross (1984), Karczmarski (1996), and Best (2007). The following species descriptions summarise the present knowledge of the cetaceans located within the inshore areas of Algoa Bay. The identifiable features, habitat, ecology, range and population status are outlined for each of these species.

2.8.2. Mysticetes

Baleen whales are distinctive with their large mouths and body size (Würsig, 1989). They have long keratinous plates that extend from the upper jaw to filter out invertebrates from the water in order to feed (Würsig, 1989). Their diet mainly consists of euphausiid crustaceans (krill) and copepods, and they usually feed independently, with little or no group cooperation involved in

prey capture (Würsig, 1989; Best, 2007). A thick blubber layer acts as a fat reserve, allowing these animals to undergo vast migrations from summer feeding grounds in the high latitudes to winter breeding grounds in the lower latitudes (Würsig, 1989).

With the exception of the southern right whale, the whale species observed in the coastal areas of the Eastern Cape are rorquals (Würsig, 1989; Best, 2007). Unlike the right whales, rorquals lunge for their prey (either below or at the water surface), ingesting krill and various small fish species (Würsig, 1989; Best, 2007). Presently, little is known about the behaviour and movements of this order, as most activity takes place at a range of depths (Würsig, 1989).

2.8.2.1. Southern right whale (Eubalaena australis)

Characteristic features

Southern right whales have an average length of 13.9 m (International Whaling Commission, 1986; Ross and Best, 1989; Best, 2007). Sexual maturity is reached between seven and nine years of age with a calving interval of three to four years (Payne *et al.*, 1990). These animals are distinguishable from other cetaceans by the lack of dorsal fin and the large wart-like callosities on their heads, forming a pattern unique to every individual (Plate 1) (Payne, 1986; Best, 2007).

Range

Southern right whales are found in circumpolar and temperate waters, extending towards the tropics with the warm ocean currents during austral summer (Figure 2.1) (Cummings, 1985a; Best, 2007; Belgrano *et al.*, 2008). These wintering grounds are used from June to December, with a peak in activity occurring during September/ October (Burnell and Bryden, 1997; Elwen and Best, 2004). In southern Africa, these whales have been observed from Baia dos Tigres in Angola to Maputo Bay in Mozambique, with the highest concentration on the southern coastline of South Africa, between Cape Town and Port Elizabeth (Best, 1990; Best *et al.*, 2003; Mate *et al.*, 2010). However, after the drastic population decline during the extensive whaling period, their range was more limited along the coastline, with southern right whales predominantly observed along the south coast of South Africa (Best, 2007; Belgrano *et al.*, 2008). With the significant population growth over the last three decades, this species has once again been expanding their range to their previously known limits (Best, 2007; Belgrano *et al.*, 2008).



Plate 2.1. The head of a southern right whale. *Photo: B.Melly.*



Figure 2.1. Worldwide distribution of the southern right whale (Reilly et al., 2008b).

Habitat and ecology

The coastal regions of the southern continents, including South Africa, are used as breeding and calving grounds during austral winter (Cummings, 1985a; Burnell and Bryden, 1997). Off the coast of South Africa, southern right whales are found in groups of one to ten animals, with the nursery groups mostly comprised of mother-calf pairs (Best *et al.*, 2003; Patenaude, 2003; Costa *et al.*, 2005). These animals utilise shallow waters which are less than two kilometres from land, but occasionally have been observed further offshore (Payne, 1986; Best *et al.*, 2001b).

These coastal habitats are comprised of shallow, gently sloping bays sheltered from wind and swell, where females give birth and nurse their young (Thomas and Taber, 1984; Elwen and Best, 2004; Sironi *et al.*, 2008). Mating activities are generally observed in separate areas to these calving grounds (Payne, 1986; Patenaude, 2003; Elwen and Best, 2004). These mating activities are often classed as 'surface active groups' (SAGs), which consist of two to ten animals (with one focus animal that is generally a female) that perform various behaviours such as mating, exposing their bellies and flippers at the surface of the water, rolling, spyhopping and splashing (Best *et al.*, 2003; Patenaude, 2003; Best, 2007). Other behaviours associated with southern right whales in the coastal areas include breaching, and sailing (Cummings, 1985a).

Population status

The southern right whale population is estimated to be approximately 3100, with a population growth rate of 7% per year since 1971, they therefore have a national Red List status of 'least concern' (Friedmann and Daly, 2004; Reilly *et al.*, 2008b). Nevertheless, this population is only 10 to 20% of the estimated original population size off southern Africa (Best *et al.*, 2001a).

2.8.2.2. Humpback whale (Megaptera novaeangliae)

Characteristic features

Humpback whales grow to a maximum length of 15.2 m in females and 14.3 m in males, with their head comprising up to 30% of the length of the body (Best, 2007). These whales mature sexually between four and five years of age with a calving interval of approximately two years (Findlay *et al.*, 1994; Best *et al.*, 1999). Humpback whales have a characteristic dorsal fin situated two-thirds of the way down the back, towards the fluke, with the back arching in this area before a dive (Best, 2007). They have large, 'wing-like' flippers with white markings that are occasionally seen during surface activity. The trailing edge of the fluke is serrated with white

and black markings on the ventral side that is often displayed before a dive (Winn and Reichley, 1985; Best, 2007). The head is rounded, with the surface of the rostrum encompassing a pattern of knob-like protuberances, apparent at a close range (Würsig, 1989; Findlay and Best, 1995).

Range

Humpback whales are a cosmopolitan species, migrating from the summer feeding grounds in the polar regions (up to 64° S), to lower latitude breeding grounds in winter (Figure 2.2) (Jefferson and Schiro, 1997; Rice, 1998). Off southern Africa, the main breeding grounds are from Angola to Gabon on the west coast, to Mozambique and Tanzania on the east coast, as well as along the various islands that are located in this region (Findlay *et al.*, 1994; Rice, 1998; Best *et al.*, 1999; Barendse *et al.*, 2010).

Habitat and ecology

Humpback whales utilise both coastal and offshore areas (along the continental shelf) in their wintering grounds (Best *et al.*, 1998; Keiper *et al.*, 2005; Clapham, 2009). Contrastingly, mother-calf pairs are generally found in more sheltered and shallow areas along the coastline (Picanço *et al.*, 2009). Although mother-calf pairs are recorded along the South African coastline, this region is predominantly used as a migratory corridor, with most calving and mating occurring in the more tropical regions (Angola and Mozambique) (Dawbin, 1966; Best, 1994; Best, 2007). Births occur between July and October, and peak in August (Klinowska, 1991). Group sizes consist of an average of two individuals, but range from one to five animals (Best and Sekiguchi, 1996; Best, 2007). Social interactions and mother-calf pairs are found in approximately 25% of these groups (Best and Sekiguchi, 1996; Best, 2007).

This species uses either lunge-feeding or 'bubble-netting' to feed on euphausiids, amphipods, stomatopods, and krill in the sub-Antarctic (Winn and Reichley, 1985; Findlay and Best, 1995). Opportunistic feeding has also occasionally been observed along the west coast of South Africa (Findlay and Best, 1995; Best, 2007; Barendse *et al.*, 2010). Other behaviours in their wintering grounds include: spy hopping, breaching, fluking, tailslapping, flippering and lobtailing (Würsig, 1989; Clark and Clapham, 2004; Smith *et al.*, 2008).



Plate 2.2. Humpback whale breaching (Cape Padrone in background). Photo: B. Melly.



Figure 2.2. Worldwide distribution of the humpback whale (Reilly et al., 2008c).

Population status

A survey off Mozambique in 2003 estimated a humpback whale population of over 5800 individuals along the east coast of South Africa, with a population increase of eight percent since 1990 (Best and Sekiguchi, 1996; Friedmann and Daly, 2004). Data along the west coast is insufficient for determining the population size or the species Red List status, but humpback whales are considered to be 'near threatened' in this region (Klinowska, 1991; Friedmann and Daly, 2004). However, they are listed as 'least concern' by the IUCN due to steadily increasing population numbers over the last few decades (Klinowska, 1991; Friedmann and Daly, 2004).

2.8.2.3. Bryde's whale (Balaenoptera brydei)

Characteristic features

Bryde's whales are similar in size and shape to sei whales and minke whales (Best, 2007; Kato and Perrin, 2009). The size of the inshore form of *B. brydei*, ranges from 13 to 15 m in length (Best *et al.*, 1984; Cummings, 1985b). The dorsal fin is falcate and situated two thirds along the back, but unlike the minke whale, this species has dark flippers and three parallel ridges running along the rostrum (Best, 1977; Best, 2007). Bryde's whales are counter-shaded with a grey-black dorsal and white ventral surface, with irregular white patterns on the dorsal surface of their body (Best, 1977; Best, 2007). Sexual maturity appears to occur at body lengths between 12 and 13 m (Best, 2007). However, knowledge of the breeding and calving cycles and locations for this species is limited (Best, 2007; Wiseman, 2008; Penry, 2010). These animals do not appear to show a marked seasonality in breeding, with conception occurring throughout the year (Omura, 1959; Best, 2007; Penry, 2010).

Range

Bryde's whales are a non-migratory, tropical and subtropical species, found worldwide (Figure 2.3) (Tershy *et al.*, 1990; Mullin *et al.*, 1994; Barlow, 2006). In southern Africa, three subpopulations exist, one of which is an inshore, resident population south of 30 °S occurring within 37 km of the coast (Best *et al.*, 1984; Best, 2001; Penry, 2010). Two offshore, migratory populations are found along the west coast of southern Africa and the south of Madagascar, possibly extending towards the Seychelles (Best, 2001; Friedmann and Daly, 2004).



Plate 2.3. Bryde's whale. Photo: B. Melly.



Figure 2.3. Worldwide range of Bryde's whales (Reilly et al., 2008a).

Habitat and ecology

Little is known about the habitat preferences for this species (Best, 2007; Wiseman, 2008; Penry, 2010). Group sizes of Bryde's whales range from one to eight animals (O'Callaghan and Baker, 2002; Wiseman, 2008), but they have also been observed in groups as large as 100 individuals, usually in response to high prey densities (Best, 2001). Bryde's whales feed on pelagic fish, using vertical or lunge feeding, and often undergo small seasonal migrations to follow the movement of their prey (Cummings, 1985b; O'Callaghan and Baker, 2002). These animals also have a number of feeding associations with other marine animals such as gannets, dolphins, seals, penguins and gulls (Zerbini *et al.*, 1997; O'Callaghan and Baker, 2002; O'Donoghue *et al.*, 2010b).

Population status

Friedmann and Daly (2004) estimate an inshore Bryde's whale population of 519 off South Africa, which is similar to the estimate given by Best *et al.* (1984) of approximately 580 animals. Both approximations were thought to be underestimated due to many secondary sightings being omitted (Best, 2007). However, a recent study conducted on these animals suggests that the population size of these animals may be as small as 130 to 250 individuals (Penry, 2010).

The three Bryde's whale subpopulations appear to have little gene flow between them, and are therefore generally considered separately for management strategies (Best, 2007). The inshore stock are listed as 'vulnerable' based on the population being less then 1000 individuals, with the two offshore stocks being listed as 'data deficient' (Friedmann and Daly, 2004; Reilly *et al.*, 2008a). The population trend for all of the subpopulations is currently unknown (Friedmann and Daly, 2004).

2.8.3. Odontocetes

Odontocetes differ from mysticetes as they have a smaller body size and bear teeth (Best, 2007). Their diet consists of a much wider variety of prey, predominantly squid and different fish species (Saayman *et al.*, 1972; Klinowska, 1991; Best, 2007). Compared to mysticetes, odontocetes are limited more to coastal waters, with restricted habitats, and have much larger group sizes (Rice, 1998).

2.8.3.1. Indo-Pacific bottlenose dolphin (*Tursiops aduncus*)

Characteristic features

T. aduncus has a large dorsal fin and a dark dorsal region with dark spots on the belly on many individuals (absent in *T. truncatus*) (Ross, 1984; Best, 2007). Average lengths are 2.38 m in females and 2.43 m in males, with this form is being less robust and smaller than *T. truncatus* (Cockcroft and Ross, 1990; Mann *et al.*, 2000). Births occur primarily in summer and autumn and calving intervals range from 3 to 6.2 years (Saayman *et al.*, 1973; Cockcroft *et al.*, 1992).

Range

Bottlenose dolphins are cosmopolitan species, found in a wide range of habitats in both tropical and temperate waters world-wide (Weigle, 1990; Defran and Weller, 1999; Wang and Yang, 2009). Despite this species being well studied, their taxonomy is still unclear (Ross, 1977; Ross, 1984; Rice, 1998). The Indo-Pacific bottlenose dolphin (hereafter referred to as bottlenose dolphin) is found in the temperate and tropical waters of the Indian Ocean basin, extending from east Africa, along south and south-east Asia to Japan and Australia (Figure 2.4) (Findlay *et al.*, 1992; Fury and Harrison, 2008). In South Africa, these dolphins are found all along the coastline from Cape Agulhas to Mozambique (Fury and Harrison, 2008). A seasonal movement along the South African coastline is thought to occur with the annual Sardine Run drawing these dolphins towards Kwa-Zulu Natal (KZN) (Ross, 1984). Movements between Plettenberg Bay and Algoa Bay have also been observed (Saayman *et al.*, 1972; Cockcroft *et al.*, 1992).

Habitat and ecology

Bottlenose dolphins commonly occupy coastal waters less than 30 m deep (within one kilometre from land) (Fury and Harrison, 2008). However, they are also known to utilise a wider range of habitats including deep ocean basins, lagoons and estuaries (Ross, 1984; Best, 2007). This species lives in fission-fusion societies that are comprised of short-term acquaintances that last several hours to a few days (Lusseau *et al.*, 2006). Thus, group sizes vary significantly. These animals are usually seen in groups of around 40 dolphins, with larger group sizes of approximately 140 animals, having been reported in the Eastern Cape (Cockcroft and Ross, 1990; Klinowska, 1991).



Plate 2.4. Bottlenose dolphins leaping out the back of a wave (Alexandria Dunefield in background). *Photo: B. Melly*.



Figure 2.4. Worldwide range of Indo-Pacific bottlenose dolphins (Hammond et al., 2008b).

Cooperative feeding has been observed in this species, where the group herds the prey, allowing individuals to feed (Lockyer, 1990; Acevedo, 1991; Constantine *et al.*, 2004). This is more common in deeper waters, with independent feeding predominating in more shallow waters (Klinowska, 1991). Bottlenose dolphins have also been observed interacting with other cetacean species, especially when foraging in baitballs (for example, with Bryde's whales and common dolphins) (Deakos *et al.*, 2010; O'Donoghue *et al.*, 2010b). They are also known to voluntarily interact with humans, but research has also illustrated distinct boat and human avoidance (Rice, 1998; Fury and Harrison, 2008).

Population status

Due to the many discrepancies between the various species, *Tursiops* spp. are listed as 'data deficient' by the IUCN, and *T. aduncus* is classified as 'vulnerable' in the Red Data book, with the migratory subpopulation being classified as 'endangered' in South Africa (Friedmann and Daly, 2004; Hammond *et al.*, 2008b). Population estimates are vague with estimates of between 520 (Ross *et al.*, 1989) and 900 individuals along the Natal coastline (Reisinger and Karczmarski, 2009). A photo-identification study carried out in the early 1990s estimated a population of 28 482 individuals off the Eastern Cape, suggesting that these dolphins inhabit a larger range along the South African coastline than what was originally estimated (Wells and Scott, 1999; Reisinger and Karczmarski, 2009).

2.8.3.2. Indo-Pacific humpback dolphin (Sousa chinensis)

Currently there is confusion over which *Sousa* species inhabits the south-east coast of South Africa (Best, 2007). On the west coast of South Africa, the Atlantic form is known as *Sousa teuszii* (Ross *et al.*, 1994). In the Indian Ocean, two populations exist, namely, the Indian humpback dolphin (*S. plumbea*) and Indo-Pacific humpback dolphin (*S. chinensis*) (Ross *et al.*, 1994; Best, 2007). Morphometrically there is little or no difference between these two populations (Ross and Best, 1989; Kaschner *et al.*, 2006b). For the purposes of this study, the species found off the south coast of South Africa is referred to as the Indo-Pacific humpback dolphin (hereafter referred to as humpback dolphin) based on literature and studies in the area (see: Durham, 1994; Ross *et al.*, 1994; Karczmarski, 1996; Rice, 1998; Jefferson and Karczmarski, 2001; Keith *et al.*, 2002; Guissamulo and Cockcroft, 2004; Best, 2007).

Characteristic features

Humpback dolphins show sexual dimorphism, with males being larger (average length of 2.79 m) than females (average length of 2.49) (Ross *et al.*, 1994; Best, 2007). These animals have a brownish-grey body, long beak, and a distinctive hump with a small, hooked dorsal fin (Ross *et al.*, 1994; Karczmarski *et al.*, 1999).

Range

Humpback dolphins are found in coastal areas of the Indian Ocean ranging from the south coast of South Africa, along the eastern coastline of Africa, south Asia, the Philippines and the associated islands in the ocean basin (Figure 2.5) (Ross *et al.*, 1994; Jefferson and Karczmarski, 2001; Friedmann and Daly, 2004). These animals are primarily found in bays, river mouths, and estuaries off South Africa, extending from False Bay to the KZN coastline and into East Africa. Females with calves tend to be resident within their habitat, with other individuals moving along the coastline to neighbouring bays (Karczmarski, 2000; Friedmann and Daly, 2004).

Habitat and ecology

Humpback dolphins are observed in waters less than 15 m in depth, and within one kilometre from land (Findlay *et al.*, 1992; Karczmarski, 1996; Keith, 1999). Group sizes are typically less than 25 individuals, with the average group size less than 10 animals (Karczmarski, 1999; Jefferson, 2000; Parra and Ross, 2009). Larger groups are generally associated with calves, with births peaking in summer along the southern African coastline (Karczmarski and Cockcroft, 1999).

Humpback dolphins feed individually on a variety of fish species in sandy-sheltered bays and along rocky shores (Keith *et al.*, 2002; Atkins *et al.*, 2004). A diurnal pattern of feeding has been previously observed in Algoa Bay, with peaks occurring in the early morning and late afternoon during the winter months (Karczmarski *et al.*, 2000a). These animals tend to use exposed and disturbed areas of water (for example, areas of high human activity) for travelling (Saayman and Tayler, 1979; Keith, 1999; Best, 2007).

Bottlenose dolphins and humpback dolphins are rarely seen integrating, despite very similar niches. However, single humpback dolphins have been seen within a group of bottlenose dolphins (Karczmarski *et al.*, 1997). Humpback dolphins also tend to avoid boats and human activities, and are known to increase dive times and disperse when approached by boats or

humans (Ross *et al.*, 1994; Best, 2007). Like bottlenose dolphins, humpback dolphins display social behaviours such as leaping and chasing (Friedmann and Daly, 2004).



Plate 2.5. Humpback dolphin. Photo: B. Melly.



Figure 2.5. Worldwide distribution of Indo-pacific humpback dolphins (Reeves et al., 2008).

Population status

Approximately 1000 individuals are present along the South African coastline and it is thought that the numbers are decreasing (Rice, 1998; Fury and Harrison, 2008; Reeves *et al.*, 2008). Best (2007), suggests a fluctuation in growth rates between -3% and 2%, which is suggestive of some stability. At a global level, this species is considered 'data deficient', and at a local level it is classified as 'vulnerable' (Friedmann and Daly, 2004) or 'near threatened' (Reeves *et al.*, 2008).

2.8.3.3. Long-beaked common dolphin (Delphinus capensis)

The taxonomy of the common dolphin is still unresolved (Rice, 1998). Currently, it is recognised that the short-beaked common dolphin (*D. delphis*) occupy mainly offshore areas off the South African coastline, compared to their long-beaked counterparts (*D. capensis*) which occupy more inshore habitats (Ross, 1984; Evans, 1994). Recently, it has been proposed that common dolphins could even be considered as one widely distributed 'super-species', with local differentiation between the various populations (International Whaling Commission, 2009).

Characteristic features

Long-beaked common dolphins (hereafter referred to as common dolphins) are slender and streamlined with yellow and grey 'hourglass' or 'v' shaped saddle markings on the side of their bodies (Ross, 1984; Ross and Best, 1989; Heyning and Perrin, 1994). They have a maximum length of 2.54 m in males and 2.22 m in females, being both larger and more streamlined than their short-beaked counterparts (Ross, 1984; Best, 2007). Neonates measure just under one metre, with the majority of births occurring in summer (Klinowska, 1991; Best, 2007). Sexual maturity in displayed in animals over 1.5 m in length, with a calving interval of approximately three years (Cockcroft and Peddemors, 1990; Evans, 1994).

Range

Common dolphins are a widely distributed species situated in tropical and temperate oceans, especially in South America, California, western and southern Africa, and south-east Asia (Figure 2.6) (Findlay *et al.*, 1992; Rice, 1998). Annual winter migrations along the east coast of South Africa are common, following the Sardine Run, which begins in the Eastern Cape and travels up to the north coast of KZN (Friedmann and Daly, 2004; Best, 2007; van der Lingen *et al.*, 2010).



Plate 2.6. Common dolphins porpoising. Photo: B. Melly.



Figure 2.6. Worldwide distribution of common dolphins (Hammond et al., 2008a).

Habitat and ecology

Group sizes vary from 100s to 1000s of individuals, with an average group size of 302 (Evans, 1994). However, they appear to have social units of around 30 animals (Evans, 1994; Barlow, 1995). Off South Africa, common dolphins occupy more coastal habitats of less then 500 m in depth, while short-beaked common dolphins are found further offshore (Findlay *et al.*, 1992; Peddemors, 1999).

These animals are known to travel along escarpments in search of prey, and therefore tend to have more offshore distributions compared to other delphinid species (Peschak, 2005; Samaai *et al.*, 2005; Filby *et al.*, 2010). They forage individually or use group herding strategies to feed on pelagic shoaling fish (Best *et al.*, 1984; Best, 2007). Several species have been reported in feeding associations with common dolphins, including Bryde's whales, bottlenose dolphins, fur seals, seabirds and sharks (Heyning and Perrin, 1994; Frantzis and Herzing, 2002; Best, 2007).

Population status

In 1988/ 89 15 000 to 20 000 common dolphins were estimated along the south-east coast of South Africa, which does not include the full range of the species (Würsig, 1989). Although this is outdated, no other estimates currently exist (Best, 2007). At a national level they have a status of 'least concern', with the IUCN classifying this species as 'data deficient' (Friedmann and Daly, 2004; Best, 2007; Hammond *et al.*, 2008a).

2.9. SUMMARY AND CONCLUSION

Common Name	Scientific name	IUCN Population status	Global range	
Southern right whale	Eubalaena australis	Least concern	Circumpolar - temperate	
Humpback whale	Megaptera novaeangliae	Least concern	Cosmopolitan	
Bryde's whale	Balaenoptera brydei	Data deficient	Tropical - subtropical	
Indo-Pacific humpback dolphin	Sousa chinensis	Near threatened	Tropical - warm temperate	
Indo-Pacific bottlenose dolphin	Tursiops aduncus	Data deficient	Tropical - warm temperate	
Long-beaked common dolphin	Delphinus capensis	Data deficient	Tropical - warm temperate	

Table 2.1. Summary table of the cetaceans found within the inshore areas of Algoa Bay.

Table 2.1 provides a synopsis of the name, IUCN population status and range of the six cetacean species found within the inshore areas of the Algoa Bay, the knowledge of which is based on

previous studies. Two species, the dwarf minke whale (*Balaenoptera acutorostrata*) and killer whale (*Orcinus Orca*), which have been occasionally sighted in the bay, were not discussed as they were not observed during the time span of this study.

Despite continued research on cetaceans in South Africa, there are still a large number of questions related to the distribution and habitat use of these animals along this coastline, as well as the geographical variables affecting their distribution. In order for appropriate conservation and management strategies to be outlined and implemented for any cetacean species, these parameters need to be more accurately defined. Non-systematic line-transect sampling methods have proved to be useful in collecting cetacean sighting data. Spatial data analysis methods such as GIS and statistics have proven to be valuable in defining the spatial and temporal distribution and habitat of cetaceans. Consequently, this study uses a combination of these methods (discussed in Chapter Four) for researching cetacean species that inhabit the inshore areas of Algoa Bay.

CHAPTER 3: STUDY AREA

3.1. INTRODUCTION

This chapter provides a detailed description of Algoa Bay. Prominent geographical, hydrological and biological features have been delineated. Current activities that potentially affect cetacean distribution, as well as proposed future developments and conservation strategies, are also mentioned.

Algoa Bay is situated on the south Eastern Cape coastline of South Africa. It is bordered by two headlands, Cape Padrone (33°46' S 26°28' E) on the east and Cape Recife (34°02' S 25°42' E) on the western side of the bay (Figure 3.1). The coastline is approximately 135 km long, consisting mainly of sandy beaches, with rocky shores along Cape Recife. Three main rivers enter the western half of Algoa Bay: the Sundays, Swartkops and Coega Rivers. The Alexandria Dunefield borders about 50 km of the northern part of the bay from Sundays River to Cape Padrone. Port Elizabeth city is situated on the south-western corner of Algoa Bay.

Boat-based surveys were undertaken along the coastal waters between the two headlands of Algoa Bay, as well as along key geographical features within the bay. This includes two island groups and a shallow reef on the south-west border, which will be discussed in the following section.

3.2. GEOGRAPHICAL FEATURES

The coastline west of Algoa Bay consists of several relatively shallow bays with maximum depths of 100 m and average depths of less than 50 m; of which Algoa Bay is the largest with an area of approximately 3100 km² (Bremner, 1979a; Bremner, 1983; Bremner *et al.*, 1991). This bay has a planimetric shape with the appearance of a near perfect clockwise logarithmic spiral (Figure 3.1) (Bremner, 1979b; Ross, 1984). Approximately 45 km south of Cape Recife lies the continental shelf break at a depth of 150 m (Bremner *et al.*, 1991).

The Alexandria Dunefield borders the north and north-eastern boundaries of the bay, forming an extended sandy shore stretching from Sundays River to Woody Cape (Figure 3.1) (McLachlan, 1983; Illenberger, 1986). This dunefield is the largest in South Africa with an area of 120 km²

extending across 50 km, with an average width of 2.2 km and consists mostly of unvegetated mobile dunes (Illenberger, 1986).



Figure 3.1. Features of Algoa Bay on the south Eastern Cape coastline of South Africa.

The Swartkops and Sundays Rivers are two large perennial systems with permanently open estuaries into Algoa Bay (Figure 3.1) (Bremner, 1983). The Sundays River has a catchment area of 20 729 km², with the estuary extending approximately 21 km inland, bordering the Alexandria Dunefield (Harrison and Whitfield, 1990). The Swartkops River has a catchment size of approximately 1 555 km² and is situated 10 km north-east of Port Elizabeth (Melville-Smith and Baird, 1980). The estuary is 16 km long and has evidence of elevated levels of heavy metals which are deposited in the sediments of the surrounding ocean floor (Melville-Smith and Baird, 1980; Binning and Baird, 2001). The seasonal Coega River is the next largest river entering the bay, supporting a salt-extraction works, with the Port of Ngqura (Coega Port) situated at the mouth of this non-perennial river (Figure 3.1) (Bremner, 1983). The Papkuils River and the

Baakens River are also small non-perennial streams, the latter, flowing into PE Port (Figure 3.1) (Bremner, 1983).

A reef and two island groups distinguishes Algoa Bay from other bays in South Africa (Figure 3.1). Riy Banks is a shallow reef situated approximately 20 km east-southeast of PE Port (Bremner, 1979a; Bremner *et al.*, 1991). The reef rises from a depth of approximately 50 m to depths between 12 and 15 m, forming a rough ovoid-shaped plateau of 24 km², making it a popular diving and spear-fishing site (Bremner, 1979a; Bremner *et al.*, 1991). The northern and western flanks of the outcrop are steep, with more gently sloping topography on the south and eastern flanks (Bremner *et al.*, 1991).

The Bird Island group, comprising of Bird Island, Black Rocks, Seal Island and Stag Island, is situated approximately eight kilometres south-southeast of Woody Cape and two to three kilometres south of Cape Padrone (Bremner *et al.*, 1991; World Bank, 2004). The Islands of the Cross (commonly reffered to as the St Croix Islands) consist of St Croix Island, Jahleel Island and Brenton Island, all of which border the eastern side of Coega port.

These features, along with the existence of large deposits of unconsolidated sand on land (from the Alexandra Dunefield), the introduction of sediment from the two largest rivers entering the bay, the prevailing wind patterns, groundswell and the Agulhas Current, have all contributed to the present shape of Algoa Bay (Bremner, 1979b). These geographical features have also influenced the nature and distribution of sediments on the sea-floor (Figure 3.2) (Bremner, 1979b). The sea-floor mainly consists of fine sand to sand sized sediment (Figure 3.2) (Bremner, 1979b). The two larger rivers deposit coarse sediment at the river mouths, and finer clay to silt-sized sediments across the central bay area, covering the predominantly sand-based sea-floor (Figure 3.2) (Bremner, 1979b).

Bremner (1979b) and Bremner *et al.* (1991) discuss the bathymetry of Algoa Bay in detail. In general, the sea-floor slopes towards the south-southeast at about 0.15° with several islands, depressions and ridges disturbing this gradient locally (Bremner, 1979a; Karczmarski, 1996). St Croix Island, Jahleel Island and Brenton Island are three isolated outcrops forming a south-westward ridge known as the 'St Croix Ridge', and are surrounded by the generally smooth-sloping sea-floor (Bremner, 1979a). Between the two island groups lie a number of hill-like features that form shallow depressions and small uneven ridges along the north-east shoreline of

the bay (Bremner, 1979a). Cape Recife, Riy Banks and the Bird Island group are three areas of extremely rough morphology that form a steep, sea-ward dipping, discontinuous ridge of exposed bedrock along the outskirts of the Algoa Bay (known as the 'Recife Bird Ridge'), where water depth quickly exceeds 50 m (Bremner, 1979a).



Figure 3.2. Interpolated sediment map of Algoa Bay derived from SANHO point sample data using Thiessen polygons (note: sparse data in the south-west region has led to the construction of large Thiessen polygons).

3.3. CLIMATE AND WEATHER PATTERNS

This large, relatively exposed bay is situated in the subtropical climate of the Eastern Cape Province of South Africa and faces the south-west Indian Ocean. The weather of Algoa Bay is predominantly controlled by high pressure systems as well as cold fronts and coastal lows (Goschen and Schumann, 1988). These fronts and coastal lows are associated with high winds, cloud cover and rainfall (Goschen and Schumann, 1988).

Algoa Bay falls in the transition zone between winter and summer maximum rainfall regions which are found on the west and east coast respectively, and experiences an overall winter maximum rainfall (Stone *et al.*, 1998). Raw weather data were obtained from the South African Weather Service (SAWS). Port Elizabeth receives approximately 545 mm of rainfall a year, with large amounts of intra and inter-annual variability (Figure 3.3). During the summer, Algoa Bay receives an average of 8.5 to nine hours of sunshine a day, while in winter this decreases to approximately 6.9 hours per day (Karczmarski, 1996; Stone *et al.*, 1998). Average daily air temperatures range from 13 to 23 $^{\circ}$ C (Figure 3.3).



Figure 3.3. Mean monthly temperatures for all three stations in Algoa Bay, and mean monthly precipitation for Port Elizabeth. Temperatures recorded at PE airport, Coega Port and Bird Island.

Algoa Bay is dominated by west south-westerly winds, as well as south-westerly and westerly winds throughout the year, which coincides with the general orientation of the coastline (Figure 3.4) (Illenberger, 1986; Goschen and Schumann, 1988). Land and sea breezes also constitute an important component of local winds in the bay, generally intensifying in the afternoons (Beckley, 1977; Beckley and McLachlan, 1979; Karczmarski, 1996). During summer, easterly winds become more prominent and the average wind velocity increases (Goschen, 1991). Goschen

(1991) calculated that maximum wind speeds occurred between September and November, although this is thought to have shifted forward into January and February.



Figure 3.4. Wind rose for PE between 2005 and 2010. Data obtained from SAWS.

3.4. HYDROLOGICAL FEATURES

Wave direction and height in the bay is determined by local winds and distant storms combined with the underlying topography (Karczmarski, 1996). Swell height in the inshore zone of the bay is decreased by the dominant west south-westerly winds (Karczmarski, 1996). Conversely, due to Algoa Bay being exposed on the east, the easterly winds bring deep-sea waves which are unhindered by a headland and thus increase the swell height (Karczmarski, 1996). Taking into account that the sea state generally worsens throughout the course of the day, Karczmarski (1996) determined that the average monthly sea state corresponds to a Beaufort sea state of three.

The dominant oceanic feature associated with Algoa Bay is the warm Agulhas Current which flows between 44 and 60 km offshore of Cape Padrone and Cape Recife respectively (Goschen

and Schumann, 1988). The current begins alongside Mozambique and flows close inshore down the east coast of South Africa along the continental shelf (Lutjeharms, 1981; Lutjeharms *et al.*, 2001). Around Algoa Bay the current begins to diverge from the coast as the continental shelf widens towards Cape Point (in the Western Cape) where it turns around and flows eastwards as the Agulhas Return Current (Ross, 1984; Lutjeharms *et al.*, 2001; Griffiths *et al.*, 2010). The current flows quickly at a velocity of approximately one metre per second (and can exceed speeds of 2.5 m.s⁻¹), with an average temperature of between 22 and 27 °C in August and March respectively (Ross, 1984). This current plays an important role in determining temperatures and smaller currents within Algoa Bay (Goschen and Schumann, 1988).

A cooler, more nutrient-rich counter current also affects the inshore waters of the bay (Beckley and McLachlan, 1979; Griffiths *et al.*, 2010). The water temperature in Algoa Bay is generally well mixed in winter, with little or no thermocline apparent, resulting in a vertical temperature difference of approximately one degree Celsius between bottom and surface water (Beckley, 1983; Schumann and van Heerden, 1988). The average water temperature in the bay is approximately 18.1 °C¹, but can fluctuate between 10.8 and 26.5 °C during upwellings and inshore meanders of the Agulhas Current respectively (Christensen, 1980; Lasiak, 1982).

Upwellings in Algoa Bay are generated by strong winds and the bottom boundary layer of the Agulhas Current (Schumann *et al.*, 1982; Beckley, 1983). Westerly winds produce upwellings north-east of Cape Recife, while easterly winds generate upwellings on the southern shores of the two headlands (Beckley, 1983; Goschen and Schumann, 1988). These upwellings bring up cold, nutrient-rich water that result in local changes to the physical environment, having distinct affects on the biota (Karczmarski, 1996). This is enhanced due to the lack of a thermocline, allowing the upward mixing of these nutrients (Karczmarski, 1996).

Semi-diurnal tides occur in Algoa Bay, with tidal ranges of 1.61 m and 0.51 m at spring and neap tide respectively (Beckley, 1977; Karczmarski, 1996). These tides form current speeds of over 30 cm.s⁻¹, insignificant to the general circulation pattern due to the overall bathymetry of the bay (Goschen and Schumann, 1988).

¹ Measured at Humewood Beach, Port Elizabeth (1973 – 1999) and Pollock Beach, Port Elizabeth (1999 – 2010). Data obtained from the South African Weather Service (SAWS).

3.5. THE BIOTA ASSOCIATED WITH HABITATS IN ALGOA BAY

3.5.1. Sandy beaches

The sandy shores of the Alexandria Dunefield produce a significant freshwater discharge into the corresponding surf zone providing a habitat for an important diatom, *Anaulus australis*, which accounts for over 95% of the primary production along the Alexandria Dunefield coastline (McLachlan, 1983; Illenberger, 1986; Webb and Wooldridge, 1990). This diatom is preyed upon by a wide variety of invertebrates and fish, forming a crucial producer base in the associated food web, which includes odontocetes as the apex predators (Webb and Wooldridge, 1990; CSIR, 2004; CSIR, 2007).

Several fish species are also common in the surf-zone associated with sandy beaches in the bay; namely, *Pomadasys olivaceum* (gorrie), *Liza richardsoni* (South African mullet), *Sarpa salpa* (streepie), and *Lithognathus mormyrus* (sand steenbras) (Lasiak, 1983). Fish biomass fluctuates daily, with two peaks in abundance, associated with low tide and twilight (Lasiak, 1982). Spawning takes place near the shoreline during the spring and summer months and these inshore waters provide a nursery ground for juvenile fish (Lasiak, 1983; Harrison and Whitfield, 1990). The permanently open Sundays River estuary also provides an important breeding and nursery ground for many fish species that have both recreational and commercial value along the Eastern Cape coastline (Harrison and Whitfield, 1990; Pattrick and Strydom, 2008).

3.5.2. Rocky shores

The south-western component of Algoa Bay consists of several small rocky reefs (Karczmarski, 1996). Rhodophyte seaweeds such as *Plocamium corallorhiza* and *Amphiroa ephedrae* predominantly cover shallow sub-tidal reefs of between zero to eight metres (Beckley and McLachlan, 1979; Beckley and Buxton, 1989; Karczmarski, 1996). The deeper reefs are primarily covered by Ascidians, Octocorals, Hydrozoans and Sponges, with *Pyura stolonifera* (red bait) dominating reef crests (Beckley and McLachlan, 1979; Beckley and Buxton, 1989; Karczmarski, 1996).

3.5.3. Reefs

The most recent survey on reef fish was done by Beckley and Buxton (1989) who identified 49 fish species during reef surveys conducted in the 1980s. The species composition of the most abundant ichthyofauna varies little intra-annually, with most species endemic to South Africa (Beckley and Buxton, 1989). Sparidae are the most dominant species on sub-tidal reefs (Beckley and Buxton, 1989). Several schooling species such as *Boopsoidea inornata* (dikoog), *Diplodus sargus capensis* (blacktail), *Spondyliosoma emarginatum* (steentjie), *Pachymetopon aeneum* (hottentot) and *S. salpa* were the most abundant fish on these sub-tidal reefs, with *B. inornata* observed on 98% of the dives (Beckley and Buxton, 1989). The inshore reefs, like tidal pools, estuaries and sandy shores, are an important nursery ground for juvenile fish (Beckley and Buxton, 1989).

The variety of habitats within Algoa Bay supports a diverse range of fish species which have resulted in Algoa Bay supporting a large commercial and recreational fishing industry, the main industry being the chokka squid fishery (Klages *et al.*, 1992; Pattrick and Strydom, 2008). This potentially plays a key role in determining foraging areas for cetaceans (Kenney *et al.*, 1997; Bearzi *et al.*, 2006; Nelson *et al.*, 2008).

3.5.4. Marine Protected Areas

St Croix Island supports the largest population of the endemic African penguin (*Spheniscus demersus*) (Shelton *et al.*, 1984; Coastal and Environmental Services, n.d.). A significant proportion of the total Cape gannet population (*Morus capensis*) breed on Bird Island, and many roseate terns (*Sterna dougallii*) and African black oystercatchers (*Haemotopus moquini*) utilise both island groups, feed in the surrounding waters during their breeding season (Klages *et al.*, 1992; Watson *et al.*, 1997; World Bank, 2004). This area is also well known for its vast Cape fur seal (*Arctocephalus pusillus*) population that inhabits Black Rocks, and the great white shark (*Carcharadon carcharias*), which is found in the adjacent waters (Randall *et al.*, 1988; Oosthuizen, 1991).

Several important geographical features make the proposed GAENP MPA a key habitat for several marine species, including the Sundays River and the Alexandra Dunefield. The dunefield is associated with a long stretch of sandy shores bordering the Algoa Bay coastline, providing a

variety of habitats for marine animals, making this area an important breeding and nursery ground for numerous fish species and many endemic species (for example, the South African mullet and Cape gannet (*Morus capensis*)) (Illenberger, 1986; Pattrick and Strydom, 2008). Over 100 species of macroalgae and 86% of the endemic marine invertebrates are found in this area, as well as various 'vulnerable' and 'threatened' species such as the African penguin (*Spheniscus demersus*), great white shark (*Carcharadon carcharias*), and the diatom *A. australis* (Webb and Wooldridge, 1990; Klages *et al.*, 1992; World Bank, 2003; CSIR, 2007; Pattrick and Strydom, 2008; Coastal and Environmental Services, n.d.).

The GAENP MPA will potentially protect 95% of primary production in Algoa Bay, 38% of the endemic macroalgal species, 86% of South Africa's endemic marine vertebrates, 45% of endemic marine invertebrate species, 34% of endemic fish species, and 23 species of seabirds, half of which are 'vulnerable', 'threatened' or 'near-threatened' and 15 of which breed in the bay (World Bank, 2003; Pattrick and Strydom, 2008; Coastal and Environmental Services, n.d.). The GAENP MPA will also set out to rebuild the collapsed fish stocks as well as protect nursery and spawning areas for other fish species and the chokka squid (Pattrick and Strydom, 2008).

3.5.5. Sardine Run

Sardines (*Sardinops sagax*) are a major contributor to pelagic fisheries off the coast of southern Africa (Beckley and van der Lingen, 1999; van der Lingen *et al.*, 2010). They are found in temperate, coastal and shelf waters along the South African coastline, including Algoa Bay (Beckley and van der Lingen, 1999). Most sardines spawn in spring and summer along the Agulhas Bank and migrate eastwards as large shoals during winter, moving inshore as the Agulhas Bank narrows in a phenomenon known as the Sardine Run (Beckley and van der Lingen, 1999; van der Lingen *et al.*, 2010). This migration occurs with the expansion of their temperature range, due to cooling air temperatures and upwellings along the coast as the warm Agulhas Current moves further offshore, and is thought to begin around Bird Island in Algoa Bay (Baird, 1971; Armstrong *et al.*, 1991; Beckley and van der Lingen, 1999). This run is known to attract a large number of marine predators, including cetacean species such as Bryde's whales, bottlenose dolphins and common dolphins (O'Donoghue *et al.*, 2010a; O'Donoghue *et al.*, 2010b).

3.6. ANTHROPOGENIC ACTIVITIES

This study comes at a crucial time with several large scale industrial developments that have been proposed or are under construction, along with a general increase in anthropogenic activities. The role and influence of these developments and conservation strategies on cetaceans in Algoa Bay was reviewed on a broad-scale in Chapters One and Two, and will be further discussed in relation to the findings of this study in Chapter Six. A brief summary of the current activities in Algoa Bay are outlined below.

3.6.1. Industry

3.6.1.1. Port of Port Elizabeth

PE Port was established in 1825 serving local industries (especially agriculture and farming) and offers an alternative for container ships when Cape Town or Durban harbours are congested (Ports and Ships, 2010). The port has three container berths, two bulk berths and a tanker berth, with additional jetties available for tug, fishing, trawling and recreational vessels (Ports and Ships, 2010). However, all vessels are limited by the 14.5 m channel depth (Ports and Ships, 2010). The port has a total area of 115 HA with a breakwater length of 1.2 km (Ports and Ships, 2010). The shipping channel and anchoring areas for both PE and Coega Ports are illustrated in Figure 3.1^2 .

Principal products handled include container shipping, manganese ore, petroleum products and cars (Ports and Ships, 2010). The regional commercial and recreational fishing industries also make extensive use of this port (Ports and Ships, 2010). Over 1 200 ships entered the port in the 2008/ 09 financial year, with over 10 million tons of cargo being handled during the 2005/ 06 financial year (Ports and Ships, 2010).

3.6.1.2. Port of Ngqura (Coega)

The deep-water Coega Port is situated approximately 20 km north-east of PE Port at the mouth of the Coega River and forms part of the 12 000 hectare Coega IDZ (Figure 3.1 illustrates the area)

² The delineation of the ship anchoring area and shipping lanes were obtained from the Algoa Bay Navigational Chart INT 7531 SAN 1024 (updated in August 2009).

(Ports and Ships, 2010). Harbour size, channel depth and the associated infrastructure in the PE Port have been unable to keep up with the increasing demands of shipping in South Africa, which motivated the building of this deep-water port. The port was completed and began commercial operations in October 2009 (Ports and Ships, 2010). The eastern breakwater, 2.7 km in length, is laid out such that it avoids the St Croix Islands exclusion zone situated on the east of the port (Ports and Ships, 2010). This port has a maximum channel depth of 18 m and a total of 32 berths, which will serve to re-direct vessels carrying manganese and other bulk ore for export, away from the city (Ports and Ships, 2010). At present, this port serves both dry and liquid bulk carriers and cellular container vessels (Ports and Ships, 2010).

Blasting, dredging and drilling were carried out to create the deep-water channel and port in the early 2000s. The effects of noise on marine mammals is described in Richardson *et al.* (1995) and Nowacek *et al.* (2007), with some of the proposed impacts being hearing loss (temporary or permanent), discomfort, displacement from habitat, stress, energetic consequences and various affects on an individual behaviours. Destruction of the physical habitat during the development of the Coega Port could have resulted in a decline or migration of prey species that are eaten by odontocetes (Clapham *et al.*, 1999; Friedmann and Daly, 2004). Other long term anticipated effects from an increase in shipping activity may be an increase in ship strikes, higher ambient noise levels from increased shipping traffic, increased environmental pollution (for example oil), and the disturbance of benthic habitats from the increased number of ships anchoring (Clapham *et al.*, 1999; Nowacek *et al.*, 2007).

3.6.1.3. Future developments in the bay

Several reports have suggested that there is a proposed waterfront development on the south side of PE Port and Kings Beach Precinct, which would have a construction period of approximately 20 years (SRK Consulting, 2007; Cull, 2010; NMBM, 2010). This development will be comprised of a number of key features including an international conference centre, leisure and tourism services, a number of residential and retail areas, and possibly, an expansion of the allocated area for auto and container terminals, and for the fishing industry (SRK Consulting, 2007; Cull, 2010; NMBM, 2010). Alongside the many profitable aspects of these project (job creation and commodity value), there are a number of potential well-known environmental risks and impacts involved in the various stages of construction and implementation of these projects, which were delineated earlier (section 3.6.1).

An oil refinery is planned for the Coega IDZ, with construction occurring between 2012 and 2015, and operations beginning in 2016 (de Bruyn, 2010; Kernohan, 2010). The refinery will be approximately 1 000 ha in size and will produce 400 000 barrels of oil a day at peak capacity (de Bruyn, 2010; Kernohan, 2010). There are several risks involved with the establishment of an oil refinery, which start from the exploration phase, to the various stages of oil collection (the drilling and extraction phase) and the transporting and refining of the oil (O'Rourke and Connolly, 2003; Nowacek *et al.*, 2007). These risks and impacts include: oil spills (which have significant environmental consequences), the physical alteration and disruption of the water (O'Rourke and Connolly, 2003; Nowacek *et al.*, 2007).

3.6.2. Commercial exploitation of marine resources

Algoa Bay supports a variety of west and east coast fish species which have been heavily exploited over the last 50 years, with 19 of the 27 commercially exploitable line fisheries having collapsed (CSIR, 2004; Pattrick and Strydom, 2008). Thus, fishing for recreational, commercial or illegal purposes, has had a major impact on fish diversity, abundance and distribution of these prey species in the bay (CSIR, 2004; Pattrick and Strydom, 2008). The chokka squid (*Loligo vulgaris reynaudii*) fishery in Algoa Bay is an important industry in the Eastern Cape, possibly the fourth largest fishing industry in the country at present, illustrating the significance of these fishing activities (Karczmarski *et al.*, 1998; Coastal and Environmental Services, n.d.). Overfishing could results in the loss of a prey base, and consequently potentially increases the competition for prey species (Friedmann and Daly, 2004).

3.6.3. Recreational and tourism activities

A number of anthropogenic activities take place along the beachfront of Port Elizabeth which pose potential threats to cetaceans, namely swimming, snorkelling, scuba diving, paddling (surf skiing and kayaking), surfing, kite surfing, windsurfing, shore angling, urban and industrial pollution, the development of the coastal zone, and destruction of coastal habitats (for example, through the establishment of ports and dolosse) (Karczmarski *et al.*, 1998; Friedmann and Daly, 2004; SRK Consulting, 2007).

There has been a rise in recreational boating activities, especially jet skis which are used in the surf zone, an important area for several cetacean species (Karczmarski *et al.*, 1998; Friedmann and Daly, 2004; SRK Consulting, 2007). Boats have been reported following cetaceans at a very close range, which also poses a potential threat. Some operators do not have permits and are not properly educated as to the potential effects of boating activities around these animals, which increases the potential impact (Karczmarski *et al.*, 1998). This has been observed in several studies which have indicated that fast approaching boats, loud engines and improper manoeuvring of a vessel can affect group dynamics and behaviour (Karczmarski *et al.*, 1998; Van Parijs and Corkeron, 2001; Constantine *et al.*, 2004; Lusseau and Higham, 2004).

Marine eco-tourism is also on the rise in South Africa, and in Algoa Bay, where a whalewatching permit set to be released in 2010/ 2011 (Government Gazette, 2009a; Government Gazette, 2009b). Whale-watching is defined as boat- or land-based tourism that is informally or formally organised, with some commercial aspect, in order to observe and/ or swim with any of the cetacean species (Hoyt, 2001). Two benefits of whale-watching are generating revenue for the local community (some of which could be fed back into research), and changing people's attitudes towards the environment (Karczmarski *et al.*, 1998; Hoyt, 2001). However, these benefits also need to be weighed up against the potential impact that this activity has on cetaceans, both in the short- and long-term, in order for this activity to be sustainable in the future (Wilson *et al.*, 1997; Constantine *et al.*, 2004; Hoyt, 2005; Bejder *et al.*, 2006).

3.6.4. Conservation strategies

3.6.4.1. Marine Protected Areas

Two MPAs currently exist in Algoa Bay around the two island groups: the Bird Islands, which lie two to three kilometres south of Cape Padrone, and St Croix Islands, which lie north-east of the Coega River (Figure 3.1) (World Bank, 2004). Both of these groups of islands have an exclusion zone of 100 to 300 m (Bremner *et al.*, 1991; World Bank, 2004).

The proposed GAENP MPA will adjoin the existing Addo Elephant National Park to create a contiguous MPA incorporating these two island groups (Figure 3.1). This area will extend from the Coega River to beyond Cape Padrone and seawards along the shoreline, to create a MPA of approximately 120 000 ha (CES, n.d.; World Bank, 2004). The GAENP aims to prevent further

ecosystem degradation by creating a mega-biodiversity conservation area where terrestrial, aquatic and marine ecosystems are protected, while creating employment and contributing to poverty reduction (World Bank, 2004). Although there has been no recent development on the implementation of the MPA in Algoa Bay, SAN Parks are still planning to expand the GAENP MPA in the near future (Oosthuizen, 2010: Pers. Comm.).

3.6.4.2. Proposed humpback dolphin marine sanctuary

After a population study on the humpback dolphins in Algoa Bay in the early 1990s, it was proposed that a humpback dolphin marine sanctuary be established in order to protect this species along the Eastern Cape coastline (Karczmarski, 1996; Karczmarski et al., 1998). This is due to their low population numbers and their inshore distribution, which make this species particularly vulnerable to the effects of anthropogenic activities prominent in this region (Karczmarski et al., 1998). Humpback dolphins are known to prefer inshore, shallow reefs, such as those found on the south-west corner of Algoa Bay (Karczmarski, 1996). This area is actively used by these animals, especially as a primary feeding ground (Karczmarski, 1996). They are therefore susceptible to the destruction of this important habitat and vulnerable to the potential threat of increasing anthropogenic activities in this area (Karczmarski, 1996; Karczmarski et al., 1998). As mentioned previously, the main threat identified is the inshore powerboat traffic, including jet skis and inflatables (rubber ducks, zodiacs) (Karczmarski et al., 1998; Klages, 2006). The proposed humpback dolphin sanctuary therefore aims to protect this species by establishing an exclusion zone for these inshore powerboats, extending southwards from Hobie Beach to Cape Recife (Karczmarski et al., 1998; Klages, 2006). The demarcated area would be approximately 800 m wide (roughly following the 10 m isobath) and seven kilometres alongshore (Klages, 2006).

3.7. CONCLUSION

This chapter provided a detailed overview of the prominent geographical features of Algoa Bay, and a synopsis of the current activities in the bay, with proposed developments and conservation strategies. A thorough knowledge of the geography of the study area is required to accurately interpret biological data (Chapter Six), and to produce relevant and more precise recommendations and guidelines (Chapter Seven).

4.1. INTRODUCTION

The research design, methods, the associated techniques, and the tools used to achieve objectives one to five, are examined in detail in this chapter. The framework from which this was established has been outlined in Chapter One. This is based on the aim of the research which is to determine the spatial and temporal distribution, and habitat preference of the cetaceans in Algoa Bay. The research matrix outlined below (Table 4.1) indicates how the five objectives, and their associated methods, have been used to achieve this aim. The primary data were collected during dedicated boat-based surveys, with GIS and statistics being used to conduct the data analysis.

4.2. OBJECTIVE ONE (A): DEDICATED BOAT-BASED SURVEYS

4.2.1. Introduction

Dedicated boat-based surveys were performed within Algoa Bay, including within the proposed MPA, between March 2009 and July 2010. The location of various cetacean species, individual and group dynamics and behaviour, were all recorded. The surveys were carried out on a monthly basis on an 8.5 m semi-rigid boat with two outboard Evinrude engines (115 HP each). A predetermined course was followed using a bearing, or by following the coastline behind the surf zone (150 to 250 m offshore) with GPS points being taken at the beginning and end of a course change (see Appendix One for the Effort Form). Approximately four survey days (or alternatively, up to eight half days) were carried out in a month depending on weather conditions. Surveys were only carried out during daylight hours with favourable weather conditions and the tracks were recorded using a Garmin GPSMAP 76CSx GPS, which computed the boat's position every five minutes.

Table 4.1. Research Matrix.

Objectives	Research question	Data required	Data acquisition tools	Data source	Method of analysis
1. To collect data on cetacean location, behaviour and other attribute information	Which cetaceans are found in Algoa Bay?	Spatial data and associated attribute information	Dedicated boat- based surveys	Primary	Data input and preparation into a database for further analysis
		Spatial data and associated attribute information	Opportunistic sightings	Secondary	
2. To acquire secondary spatial and non-spatial data on geographical and anthropogenic variables	Can these geographical and anthropogenic variables be used to explain cetacean distribution?	Spatial, geographical and anthropogenic data	Dedicated boat- based surveys Relevant	Primary	Data input and preparation into a database for further analysis
			literature, maps and other secondary sources	Secondary	
3. To determine the spatial and temporal distribution, and behaviour of the different cetacean species in Algoa Bay	a) Where and when are cetaceans located in Algoa Bay?b) Does the behaviour of the animals help explain this distribution?	Results from previous analysis			Data analysis in Excel and spatial analysis in ArcGIS
4. To relate the distribution and behaviour of cetaceans to geographical and anthropogenic variables	a) Are there associationsbetween cetacean distributionand these variables?b) Where are potential keyhabitats for each of thecetacean species?	Relevant literature and results from previous analysis			Data analysis in Excel, ArcGIS, and R. Compare data to relevant literature
5. To produce a set of recommendations and guidelines for improved management, conservation and research	How can the outcomes of this research be used to form a baseline for future research, conservation and development strategies?	Relevant literature and results from previous analysis			Addressing existing and potential problems through reviewing results and existing literature

4.2.2. Survey design

In general, biological surveys collect data on organisms to assess the status of the resource (Mann, 1999). Marine mammal surveys are defined as encounters with groups of animals or individuals along a predetermined trackline, and staying with these animals for a brief period of time (less than 30 min) before continuing along the track (Mann, 1999). Thus, surveys provide a snapshot view of the animals in question in order to answer population level questions such as distribution (Mann, 1999). In this study, a boat-based survey was conducted in Algoa Bay, in order to obtain information on the cetaceans in the area.

4.2.2.1. Survey field procedures

Survey outline

A systematic design with equal coverage is necessary to assess absolute cetacean abundance with unbiased distribution patterns (Dawson et al., 2008). Logistical constraints such as unfavourable wind and sea conditions, prohibited this form of survey, a common problem in cetacean research (Cañadas et al., 2005; Dawson et al., 2008; Hammond, 2009). Instead, the relative abundance (a measure of the relative number of animals in an area), and more importantly, the baseline spatial and temporal distribution patterns of the cetaceans, were determined. Thus, a stratified search effort (as used by Aragones et al. (1997) and Dawson et al. (2008)) was carried out during the dedicated boat-based surveys, focusing on the coastal areas of the bay, with three main tracks being carried out monthly (Figure 4.1). The first track went from PE Port to St Croix Island, around the island, to the Sundays River mouth, and coastwise back to PE Port (going around Coega Port) (Figure 4.1). The second track followed the coastline from the PE Port to Cape Recife. Weather permitting, after completing this track, the boat continued along different bearing towards Riy Banks (a shallow reef on the outskirts of the bay) or headed straight back to the port (Figure 4.1). The third track begun at Black Rocks, passing through the Bird Islands, to the headland and then coastwise along Woody Cape/ Alexandria Dunefield to Sundays River (Figure 4.1). Consequently, the entire coastline of Algoa Bay was surveyed in this study. The offshore tracks were harder to carry out as sea conditions generally declined with increasing distance from shore. Therefore, if weather permitted, priority was given to more exposed tracks.


Figure 4.1. Outline of survey tracks performed in Algoa Bay. Tracks 1, 2 and 3 are described in the text. Dashed lines represent opportunistic tracks covered during surveys.

Survey observation technique

Cetacean detection is influenced by a number of variables that need to be controlled as far as possible. These variables include: Beaufort sea state, speed of the vessel, number of observers, observer experience and eye height of observer (Evans and Hammond, 2004; Dawson *et al.*, 2008; Hammond, 2009). Four to five crew members, with a minimum of two trained observers, acted as observers, scanning the sea 360° around the survey platform from the horizon to the boat, perpendicular to the trackline, and in front and behind of the line. Trained observers were those who have a background in marine biology and were experienced in dedicated boat-based surveys.

Sighting probability was greatly reduced when conditions deteriorated to a sea state higher than Beaufort four; therefore surveys were conducted in calm sea conditions only. Suitable conditions were Beaufort sea states of four or less, good visibility, a swell of less that two meters (and preferably less than 1.5 m), and no rain. The surveys in this study were carried out at a speed of six knots to ensure that the search effort was relatively consistent and that all individuals along a trackline were spotted, including species with longer dive times (see: Wilson *et al.*, 1997; Davis

et al., 2002; Dawson *et al.*, 2008). If the sea state declined to a high Beaufort four/ five, search effort was stopped until the vessel moved to a more protected area, or the survey was called off (in accordance with: Selzer and Payne, 1988; Dawson *et al.*, 2008). In good weather conditions and calm seas (Beaufort zero or one), speed was sometimes increased to a maximum of 8.9 kn. This speed is still below other survey speeds reported by Weigle (1990), Reilly and Fiedler (1994), Wilson *et al.* (1997), and Bearzi *et al.* (2009).

4.2.3. Cetacean sightings data

Term	Definition	Citation
Group	A group is all individuals within a 100 m radius, moving in the same general direction and performing similar behaviours, i.e. with some form of interaction between the individuals.	Campbell <i>et al.</i> , 2002; Cañadas <i>et al.</i> , 2002
Sighting	A sighting is a group of cetaceans, including any solitary animals observed. Two separate sightings were recorded if the individuals in an area showed no signs of interacting, were performing different behaviours, and/ or had distance greater than ten metres separating the two groups. However, if the individuals interacted (including inter-species interactions), or participated in similar behaviours, this was noted as one sighting.	Karczmarski <i>et al.</i> , 1999; Chilvers and Corkeron, 2003, Balmer <i>et al.</i> , 2008
Group size	Group size is estimated through counting and recounting the number of individuals observed throughout the duration of the sighting, with three final figures: maximum, minimum and best estimate.	
Group composition	The number of calves, juveniles and adults.	
Calf	A calf has a small body size (less then half of adult size), lighter colouration, the occasional presence of foetal folds (if newly born), immature swimming patterns, and the constant association with an adult female i.e. swimming in the 'calf position'.	Weigle, 1990; Smolker <i>et al.</i> , 1992; Campbell <i>et al.</i> , 2002
Juvenile	Juveniles range in size from large calves to small adults, and do not consistently travel in close association with an adult female.	Smolker <i>et al.</i> , 1992
Adult	Adults are full-sized individuals.	Smolker et al., 1992

 Table 4.2. Definitions for cetacean sightings.

4.2.3.1. Field procedures for sightings

The terms used in the collection of sighting data are explained in Table 4.2. When an individual animal or group was 'sighted' (either directly or through a cue such as a blow, splash or feeding seabirds) within 1.5 nm of the trackline, the track was stopped and the boat headed towards the sighting (Dawson *et al.*, 2008). This was recorded as 'time seen'. If cetaceans were sighted further than two nautical miles away from the trackline, the boat did not head towards the sighting, as search effort along the track would have been compromised (Barlow, 1995). Therefore, only the approximate position, and any other relevant information that could be collected from a distance, were recorded for these sightings before continuing with the survey.

Upon arrival at a sighting, the 'time closed' was noted. During the sighting, the time and position were recorded using the onboard GPS (M52i S/GPS Compact Fish finder). Several environmental variables were also documented including sea surface temperature (SST), bottom depth, Beaufort sea state, wind direction and cloud cover.

The group size, composition and surface behaviour, along with any other relevant information, was observed and recorded (further detail given below). Cetaceans were identified to the lowest taxonomic level based on descriptions in field guides and literature such as Best (2007), Perrin *et al.* (2009), and Skinner and Chimimba (2005). Bottom depth and SST were also noted using the onboard GPS. Upon leaving, the 'time left', GPS position as well as other standard environmental variables were recorded (see Appendix Two for the Sighting Form).

The boat approached the sighting slowly and was manoeuvred carefully in order to reduce any potential impact on the animals' behaviour (Weigle, 1990; Constantine *et al.*, 2004). To avoid variation in the data collected, all final decisions regarding species, group size, composition, and behaviour were made by the researchers. If there were any discrepancies on the validity of the data (especially the number of individuals and group composition in very large groups), this was noted. Very brief sightings, where the individual could not be found again (i.e. no 'time closed'), were marked as 'unconfirmed' sightings. Unconfirmed sightings were not used in further, more-detailed analysis. At the end of a sighting, the boat went back to the position at 'time seen', and continued with the track (Dawson *et al.*, 2008).

4.2.3.2. Behaviour

 Table 4.3.
 Behavioural terms for mysticetes and odontocetes.

Term	Definition	Citation
Socialising	High levels of 'playful' activity, and is recorded in both mysticetes and odontocetes. Socialising in odontocetes includes surfing waves, breaching, jumping, chasing and tail- slapping. In mysticetes calf-suckling was considered a 'socialising' behaviour.	Shane <i>et al.</i> , 1986; Balance, 1992; Karczmarski <i>et al.</i> , 2000a
Mating	Belly-to-belly contact between two individuals of the same species.	Karczmarski <i>et al.</i> , 1997
Foraging	Any effort to capture and consume prey which can be seen through direct evidence (prey in the mouth), or indirect observations. Foraging in mysticetes is indicated by lunging or circular, horizontal movements just below the surface of the water (in order to trap prey). In odontocetes, chasing prey at the surface of the water, frequent and asynchronous dives in one location with loud exhalations, or rapid sharp turning/ circular swimming on the surface, are indicative signs of foraging. There is usually no contact between individuals, although they are known to feed cooperatively.	Constantine <i>et al.</i> , 2004; Shane <i>et al.</i> , 1986; Balance, 1992; Karczmarski <i>et al.</i> , 2000a; Friedmann and Daly, 2004; Best, 2007
Slow travel	Persistent one-directional movement of the whole group at speeds of less than three knots.	Constantine <i>et al.</i> , 2004
Travelling	Persistent one-directional movement of the whole group at speeds of at least three knots.	Balance, 1992; Constantine <i>et al.</i> , 2004
Fast travel	Fast travelling involves porpoising, where the dolphins leap clear of the water while moving in a particular direction.	Constantine <i>et al.</i> , 2004
Milling	Non-directional, relaxed movements in a confined area. This behaviour is frequently seen in conjunction with other behavioural states such as foraging and socialising.	Shane <i>et al.</i> , 1986; Constantine <i>et al.</i> , 2004
Resting	In odontocetes, this was identified by the dolphins engaging in extremely slow movements (and almost no forward movement), while surfacing very close together (clumped). Odontocetes will surface together then sink slowly as a group, and at times the group appears to be stationary (floating on the surface). In mysticetes this was characterised by slow surfacing and then sinking, afterwards re-surfacing at the same place a few minutes later.	Shane <i>et al.</i> , 1986; Balance, 1992; Karczmarski <i>et al.</i> , 2000a; Constantine <i>et al.</i> , 2004
Other	In mysticetes: breaching, fluking, flipper slapping, sailing and spy hopping, as it is not unequivocally known why they perform these activities. It is thought that these activities could be for 'fun', to remove parasites, or for thermoregulation. 'Other' was also used for ambiguous activities in both odontocetes and mysticetes.	Cummings, 1985a

Field procedures for behavioural data

The definitions for the different behaviours are outlined in Table 4.3. The method used to determine behaviour in this study was *ad libitum* sampling, where observations of interest were noted, as well as the predominant behaviour (Mann, 1999). An additional method was used 10 times for each species, in order to determine whether the research boat was having an impact on the animals in a sighting (Acevedo, 1991; Constantine *et al.*, 2004). This method is defined as 'focal group' sampling by Mann (1999), where the predominant group activity (greater than 50% of the individuals) was recorded at three minute intervals, in order to better define the behavioural patterns (Mann, 1999). If the animal/ group were submerged at the three minute interval, no behaviour was recorded, and upon surfacing the behaviour was noted. Throughout the study, an effort was also made to determine the behaviour of the animals as the boat approached the group ('initial behaviour'). In addition, the behaviour upon the arrival of the boat (at 'time closed') as well as any changes in behaviour during the sighting was noted.

4.3. OBJECTIVE ONE (B): OPPORTUNISTIC DATA

Several studies have indicated that shore-based monitoring of coastal cetaceans provides valuable information on the species such as recording habitat use, and determining the temporal variation in occurrence at study sites (Saayman *et al.*, 1972; Bristow *et al.*, 2001; Pierpoint *et al.*, 2009). Although this study does not include formal shore-based monitoring of cetaceans, the value of this data were realised and therefore collected on an 'opportunistic' basis (Table 4.1).

Data from opportunistic shore-based sightings were collected by both trained and untrained individuals throughout the duration of the study period. Trained observers noted the time, approximate location, species, group size and behaviour, while untrained observers provided the same information excluding behaviour. GPS positions were acquired from Google Earth © using landmarks and 'distance from land' estimates provided by the observer. Opportunistic sightings were often reported by members of the general public, using the beaches for recreation.

Boat-based opportunistic sightings provided a platform for data collection and for supplementing the data collated from dedicated boat-based surveys. As with shore-based sightings, boat-based opportunistic sighting data were provided by both trained and untrained individuals, collecting information on time, species, group size and behaviour (depending on level of training). The GPS position, SST and depth were captured using the onboard GPS/ navigator which all vessels are

required to have. As sightings were generally brief; group size, composition and behaviour were not as detailed as dedicated surveys. Opportunistic boat-based sightings were supplied by recreational boat users in the bay such as fishermen, divers and yachtsmen. All sightings were carefully checked for inconsistencies and accuracy before being included in the final dataset.

4.4. OBJECTIVE TWO

Environmental and anthropogenic variables that could potentially affect sighting effort or the distribution and behaviour of cetaceans were identified through observations, literature and interviews. Landmarks and other key geographical features including harbours, rivers, important coastal features (such as a dunefield), and bathymetry, were downloaded or taken from topographical maps and other existing maps of the region, and used as data layers (Table 4.4).

Spatial data file name	Туре	Manipulation	Source of data
Algoa Bay (coastal) outline	Line vector file	No	STEP database
Coastal vegetation, developed areas	Polygon vector file	No	Topographical maps 3325 DC, DA, DB and 3326 CA, CB and CD
Bathymetry	Line vector file	Yes	SAEON
Existing MPAs and proposed GAENP MPA	Polygon vector files	No	SAN Parks
Shipping areas	Polygon and line vector files	No	Algoa Bay Navigational Chart INT 7531 SAN 1024 (updated in August 2009)
Sea-floor substrate	Point vector files	Yes	SANHO
Effort and sighting data	Point vector files	Yes	Boat-based surveys and opportunistic sightings

Table 4.4. Description of the spatial data files used in this research. 'Manipulation' indicates whether further processing was done on the file in preparation for spatial analysis.

4.5. OBJECTIVES THREE AND FOUR

4.5.1. Introduction

GIS was used to integrate sighting information with environmental data. Using the positions and other attribute data obtained during dedicated boat-based surveys, maps were created illustrating the spatial and temporal distribution of each species. The overall distribution as well as the spatial and temporal distribution of the cetaceans were displayed and contrasted to the surrounding environmental and anthropogenic variables using a range of data layers which were identified through objective two.

Spatial analyses were done using a Geographic Information System (GIS), ArcMap 9.3 with the spatial analyst extension (ESRI® Inc., 2008). Further spatial analysis required the use of the Geospatial Modelling Environment (GME) tool (Beyer, 2010), which requires two statistical packages to run, StatConn and R (Baier, 2009; The R project for statistical computing, 2010). Statistical analyses were conducted in both R and Microsoft Excel 2003 (Microsoft® Office Excel, 2003; The R project for statistical computing, 2010). The tracklines and sighting positions as well as the associated attribute data acquired from the dedicated boat-based surveys were entered into Excel and imported into ArcMap 9.3. In order to conduct the spatial analysis in ArcMap, the data was projected to Transverse Mercator central meridian 25 and referenced to the Hartebeesthoek 94 datum.

4.5.2. Overview of the environmental parameters

For the purposes of this study, conventional austral seasons were used. Seasons were therefore grouped into three month periods, with autumn between the months of March and May, winter between June and August, spring between September and November, and summer between December and February. Prefixes 'early', 'mid' or 'late' were added to the season to indicate the month (for example, early summer = December).

Several environmental parameters were also collected during boat-based surveys. Beaufort sea state, wind direction, cloud cover, SST and bottom depth were collected every half an hour, and/ or at the beginning and end of each sighting, depending on the length of time between sightings. As an indicator of the survey conditions throughout the study, these variables were averaged for

each month and graphed so that the conditions throughout the year could be viewed, including data collected when search effort was 'off'. The display of effort data is discussed in the GIS analysis section.

4.5.3. Search effort

To determine the search effort throughout the duration of the study, the total time spent 'on effort' (where all criteria were met for suitable survey conditions) was calculated for each day of the study (Barlow, 1995; Cañadas and Hammond, 2008). The total length of tracks covered 'on effort' for each survey, was also totalled across the months.

A search effort grid of the total length of tracklines within each grid cell (one square kilometre) was calculated using the Intersect Analysis tool in ArcMap. All the tracks were "intersected" with the grid and "dissolved" using the cell ID field, in order to determine the sum of the track lengths for each one square kilometre cell in Algoa Bay. This dissolved effort layer was then "joined" to the grid such that the grid layer contained information on the total length of tracks within each cell, representing a measure of effort per grid cell.

4.5.4. Generating a grid for Algoa Bay

No formal method of choosing an appropriate grid size was found in an extensive search of the cetacean literature. Most cetacean studies used cell sizes ranging from 25 km² to 2 860 km², depending on the size of the study area (for example: Cañadas and Hammond, 2008; Gómez de Segura *et al.*, 2008; Weir *et al.*, 2009; Becker *et al.*, 2010; Santora *et al.*, 2010). Thus, a combination of factors taken from the available literature, were considered.

A paper by Hengl (2006) discussed the methodology in selecting an appropriate grid resolution for output maps for land-based vegetation and hydrological datasets, based on some of the intrinsic properties of the data points. This includes an inherent factor that a larger resolution grid results in the aggregation of data (and visa versa), and as a result, affects the accuracy and display of the output map (Hengl, 2006; Seo *et al.*, 2009). Therefore, there is an optimum grid size that best represents spatial variability while taking into account spatial autocorrelation (Qi and Wu, 1996; Guisan and Thuiller, 2005; Hengl, 2006; Seo *et al.*, 2009). Spatial autocorrelation is

defined as values that are less or more alike (negative or positive correlation) than would otherwise be expected from random processes (Legendre, 1993).

Various studies place importance on these aforementioned aspects where the models are used for predicting present and future patterns of species distributions (Qi and Wu, 1996; Guisan and Thuiller, 2005; Seo *et al.*, 2009; Becker *et al.*, 2010). As the grid in this study was used to illustrate track distances and density of sightings, it was deemed more important to choose a grid size that was cartographically suitable (easy to visualise spatial patterns), and one that represented highly mobile animals, rather than for the use of statistical analysis and predictive modelling of distribution patterns. Several grid sizes were tested (taking into account spatial autocorrelation) with the final result being largely subjective. Thus, a grid size of one square kilometre was created (using the Geospatial Modelling Environment (GME) tool), which incidentally has also been used to illustrate other land-species distributions (Socioeconomic Data and Applications Centre, 2010).

To determine whether the grid size was suitable, a Moran's I test for spatial autocorrelation was conducted using R (Azzellino *et al.*, 2008; Paradis, 2010). Values range from '-1' to '+1', indicating perfect dispersion to perfect correlation, with a value of '0' indicating a random spatial pattern (Booth, 2004; Azzellino *et al.*, 2008). Significant p-values signify very weak or no spatial autocorrelation, which would indicate whether the grid chosen was a suitable size (Booth, 2004; Azzellino *et al.*, 2010). The Moran's I test for spatial autocorrelation is defined as:

$$I = \frac{N\sum_{i}\sum_{j}W_{i,j}(X_i - \overline{X})(X_j - \overline{X})}{(\sum_{i}\sum_{j}W_{i,j})\sum_{i}(X_i - \overline{X})^2}$$

where N is the number of units, X_i is a variable at a particular location, X_j is a variable at another location, \bar{x} is mean of the variables and W_{ij} is a matrix of spatial weights.

4.5.5. Overview of sightings

The total number of sightings for each species was calculated and displayed for each month (Bearzi *et al.*, 2009). The mean number of sightings per survey and sighting frequency (Ballance, 1992; Bearzi *et al.*, 2009) was calculated using the two equations below:

$$S_{Ave} = S_T / S_u$$

where S_{Ave} is the mean number of sightings per survey, S_T is the total number of sightings for each species and S_u is total number of surveys, and

$$S_F = S_T / H_T$$

where S_F is the sighting frequency $(S.hr^{-1})$ and H_T is the total search effort time in hours.

The mean and range of group sizes were determined for each species throughout the duration of the study, using the minimum, maximum and best estimates for the sizes of each group (Campbell *et al.*, 2002). The total time spent with each species throughout the duration of the study was quantified by subtracting 'time closed' from the 'time left' in each sighting, and totalled for all species (Bearzi *et al.*, 2009). The mean time spent with each sighting was also calculated for each species by dividing the total time by the total number of sightings.

4.5.5.1. Spatial distribution of sightings

An initial map provided an overview of all the sightings recorded in Algoa Bay throughout the duration of the survey, including those collected opportunistically. A suite of maps were then created to represent the spatial distribution of each species separately, with any opportunistic sightings included. This was done to provide a clearer picture of where each species was found. A map was also created with the locations of humpback whale and southern right whale mother-calf pairs to determine whether there were any differences in the spatial distribution between sightings with and without calves present.

4.5.6. Spatial sightings per unit effort

Sightings per unit effort (SPUE) is defined as the number of sightings observed during 100 km of search effort. SPUE provides an indication of the number of animals observed in an area or during a particular time-frame (see next section), taking into account the search effort in that area (or during a period of time) (Barendse *et al.*, 2010). The same process used in calculating the total effort in each grid cell was used to establish the number of sightings for each species per cell. The table formed was then "joined" to the effort grid layer so that the number of sightings could be

compared to the effort in each one square kilometre cell. The cells were then grouped into four classes: search effort with no sightings, low SPUE, medium SPUE and high SPUE. These classes were established using geometric intervals which are designed to handle and display continuous data (ESRI Inc., 2007). Classes contain approximately the same number of values and have intervals that are relatively constant (ESRI Inc., 2007). Four classes were also defined for individuals per unit effort (IPUE). Maps for each species, depicting these categories, were then created to determine whether there were key areas with a high number of sightings that have been corrected for the intensity of search effort.

4.5.7. Temporal sightings per unit effort

In order to determine whether SPUE exhibited changes throughout the year, the total number of sightings (overall and for each species) per 100 km effort, per month, was calculated using the following equation:

$$SPUE = (S_T / TL_{MT}) \times 100$$

where TL_{MT} is the total track length for a month.

This index provides a graphical representation of the relative abundance overall, and for a particular species in each month of the year (Findlay *et al.*, 1994; Mullin *et al.*, 1994; Keiper *et al.*, 2005). This was used to clarify whether species were more prevalent in the bay in certain months of the year (Elwen *et al.*, 2009). The overall mean SPUE was calculated by taking the average of all the SPUE for each month of the year and was displayed as a horizontal line on the graph (Keiper *et al.*, 2005; Barendse *et al.*, 2010).

4.5.8. Group dynamics and relative density

The total number of calves, juveniles and individuals were calculated for each species. Frequency of individuals was determined using the same equation as sighting frequency, to provide an indication of how often an individual from a particular species was observed (Baumgartner *et al.*, 2001; Bearzi *et al.*, 2009; Gilles *et al.*, 2009). The relative density of each species in Algoa Bay was also calculated using the following equation:

$RD = I_T / TL_T$

where RD is the relative density (Indiv.km⁻¹), I_T is the total number of individuals observed throughout survey and TL_T is the total length of tracks completed throughout survey (km).

4.5.9. Observed behaviour of the cetaceans

The predominant behaviour for each sighting was used to establish the behavioural budget for each species observed, both overall and per season. This was done by enumerating the number of times a particular behaviour occurred and dividing it by the total number of sightings for that species, to indicate how much time a species spends doing a particular activity compared to other behaviours (Bearzi *et al.*, 2009). As behavioural observations of 'mating' in bottlenose dolphins were often intermingled with other socialising activities, no differentiation was made, and the overall behaviour was assigned as 'socialising'. The behavioural budget was also extrapolated for each species per season.

4.5.10. Cetacean relative densities

Kernel density is an interpolation function which fits a smooth surface to each point (sighting) in order to visualise the density of sightings per unit area (Silverman, 1986; Fotheringham *et al.*, 2000). It is therefore an effective tool for determining key areas/ critical habitats, and to visualise trends over large areas (Bailey and Gatrell, 1995; Fotheringham *et al.*, 2000). Kernel density was estimated for each species using the Kernel density tool in ArcMap. This feature took into account the number of individuals in each sighting. An output cell size of approximately 130 map units and a search radius of 4 000 m were used to create the final raster features. These maps were then visually compared to the SPUE maps in order to determine whether certain areas of the bay had higher relative densities compared to others.

4.5.11. Sightings and environmental variables

Boxplots were created for several environmental parameters to determine the relationship between each species and a habitat type/ environmental parameter (Baumgartner *et al.*, 2001; Elwen *et al.*, 2009; Weir *et al.*, 2009). The bottom depth and SST recorded at each sighting was graphed as a boxplot in R with the median, upper and lower quartiles displayed as the box.

Whiskers represented the data within 1.5 times the inter-quartile range, and points were used to indicate any outliers. To determine the extent to which the results for the bottom depths were a construct of the depths surveyed, the distribution of depth classes covered during the survey, were calculated. This was done by overlaying the survey grid with the bathymetry. The number of cells within each depth class was totalled and multiplied by the total length of tracks within the depth class. The proportion of surveys conducted within each depth class was then graphed.

A boxplot was also created to elucidate the median and mean distance from land and from the nearest river, for each species. The distances were measured using the Spatial Join Analysis Tool in ArcMap by creating a spatial join between the species sightings and the Algoa Bay coastline. The match option "closest" was used to calculate the closest distance of each sighting, from the land or river (in kilometres). The relevant data was then tabulated and a boxplot was created.

A one-way ANOVA was then conducted on these environmental variables in R, to establish whether there was a significant difference in means of each of these variables, among the different species (Townend, 2003; McKillup, 2006). If a significant difference was found, a posthoc Tukey HSD test was conducted to indicate which species had significant differences between their means (Moore *et al.*, 2000; Redfern *et al.*, 2006). The mean bottom depth and distance from land for humpback whale and southern right whale mother-calf pairs were also calculated and compared to the rest of the sightings for that species, using the Student's T-test.

Substrate data for Algoa Bay was only available in point data form. Thus, in order to determine what substrate each sighting was associated with, the point data layer was 'coded' by giving each substrate type a number, and then converted into thiessen polygons. Thiessen polygons contain one point within each polygon, with all points within the polygon being closer to that point than to the point of any other polygon (ESRI Inc., 2007). As the results were based on the best available data (with some large data gaps), the final sediment map was potentially not the most accurate representation of the sediment distribution on the sea floor.

The Spatial Join function was used to determine which sightings were associated with the different sea-floor substrates. The proportion of each type of bottom substrate for each species was then graphed and compared to the other species, to examine these associations (Bearzi and Politi, 1999; Allen *et al.*, 2001).

4.5.12. Sightings and anthropogenic variables

Four areas were defined for the shipping zones in Algoa Bay (see Figure 3.1). The two shipping approaches (channels) for each port were delineated and named the Port Elizabeth and Coega channels. Two anchoring areas also exist for these two ports: an inshore anchoring area and an offshore anchoring area, which have also been illustrated in Figure 3.1. To determine the number of sightings for each species that were found in each of these zones (and thus indicate potential areas of conflict), the Intersect Analysis tool was used to establish which observations were attributed to each of these zones. The proportion of sightings for each species within a zone was then extrapolated and graphed.

As mentioned previously, mysticetes are known to select certain areas along the subtropical and tropical coastline during austral winter, to breed and calve (Shane *et al.*, 1986; Ingram and Rogan, 2002). Thus, there are potentially certain areas within Algoa Bay that are particularly suited to these groups of animals. The number of mysticete calves and juveniles found both within and outside the proposed GAENP MPA was therefore calculated using the Intersect Analysis tool, between the sightings layer and the proposed MPA polygon layer. The purpose was to illustrate the potential importance of this proposed MPA in protecting these more 'vulnerable' individuals in the cetacean population.

4.6. SIGHTING RATE AND RELATIVE ABUNDANCE MODELLING

Statistical models are a set of probability distributions (in the form of mathematical equations) that describe the relationship between a set of independent and dependent variables (McCullagh, 2002). Thus, these models can be used to explain animal population distribution patterns (Barry and Welsh, 2002). In addition to GIS techniques which were used to display spatial and temporal distribution patterns, statistical models were used to further explain the relationship of the different cetacean species to their surrounding environment. The first step to understanding these relationships was to perform exploratory data analysis using descriptive statistics and visualising the data. After which, several variables were further analysed using Generalised Additive Models (GAMs), Generalised Linear Models (GLMs) and Zero-altered conditional models, also known as hurdle models.

4.6.1. Exploratory data analysis

Pairwise-scatterplots were constructed in R to investigate possible relationships between the presence of the different cetacean species and the geographical/ environmental factors in Algoa Bay (Swartzman *et al.*, 1994; Fotheringham *et al.*, 2000; McKillup, 2006). Variables included group size, Beaufort sea state, SST, bottom depth, seasonality, distance from land, and distance from the nearest river. Seasonality, being cyclic, was included in the models using the angular transformation of the month of the year (January = 0, December = 11) in the form sin ($\pi \times$ month/12) and cos ($\pi \times$ month/12). Both trigonometric identities were required to determine the peak of the cycle (the month of the year).

4.6.2. Generalised models

Statistically significant correlations identified in the scatterplots were further analysed using both GAMs and GLMs to gain ecological insight into the relationship between the response variable and the selected explanatory variables (Becker *et al.*, 2010).

GLMs are an extension of linear models with greater flexibility, which allows for non-constant variance structures in the data and any error structure within the exponential family of distributions (Nelder and Wedderburn, 1972; Redfern *et al.*, 2006; Becker *et al.*, 2010). In GLMs, the linear model is to be related to the response variable, y, via a link function, $\eta(\cdot)$, and is modelled as a sum of the explanatory variables $(x_1, x_2, ..., x_p)$ each corresponding to linear coefficient $(\beta_1, \beta_2, ..., \beta_k)$, such that:

$$\eta(y) = \mu = \alpha + \sum_{i=1}^{k} \beta_i x_i$$

GAMs are non-parametric extensions of GLMs (Hastie and Tibshirani, 1990) in that the linear coefficients are replaced by any parametric or non-parametric function, denoted as $f_i(\cdot)$, of the explanatory variables such that:

$$\eta(y) = \mu = \alpha + \sum_{i=1}^{k} f_i(x_i)$$

A distinct advantage of using GAMs over GLMs is that GAMs have the ability to deal with nonlinear and non-parametric relationships between the response and explanatory variables (Hastie and Tibshirani, 1987; Hastie and Tibshirani, 1990; Guisan *et al.*, 2002; Agenbag *et al.*, 2003). In both GLMs and GAMs, the coefficients are estimated by minimising the appropriate negated loglikelihood function, $\ln L$. Akaike's Information Criterion (AIC: Akaike, 1973) was used in the GLMs and GAMs as the basis for selecting the most parsimonious model that explained the most variance with the fewest number of parameters. Thus, linear, polynomials (second through forth order) denoted as *poly*(·), and smoothing splines denoted as *s*(·), were included in a stepwise forward/backward variable selection procedure. The most parsimonious model had the lowest AIC. A χ^2 test, indicated the importance of the effect of each of the explanatory variables in the final model (McKillup, 2006). To determine the fit of the model, a pseudo-coefficient of determination was calculated using the following equation:

Pseudo $-R^2 = 1 - \frac{\text{Residual deviance}}{\text{Null deviance}}$

where residual deviance = $\sum_{i=1}^{n} [\ln L(\mu_i; y_i) - \ln L(y_i; y_i)]$ and null deviance = $\sum_{i=1}^{n} [\ln L(\mu_i; \overline{y}) - \ln L(y_i; y_i)]$

with y_i being the observed response, \overline{y} the mean response and μ_i the model predicted response.

This value, which falls between zero and one, is used as a substitute for the classical R^2 , with a good fit defined by a small overall residual deviance (Swartzman *et al.*, 1994; Millar, 2000; Maunder and Punt, 2004). Due to the large number of zeros in count data, a Poisson distribution was used in the GAM models (Millar, 2000; Becker *et al.*, 2010).

Five explanatory variables were considered for all the analyses in order to construct parsimonious models. The selection of variables was based on existing literature, data availability and data limitations. Consequently, the explanatory variables used were Beaufort sea state, SST, depth and seasonality in the form of the two trigonometric identities.

4.6.3. Hurdle models

A characteristic feature of animal abundance/ occurrence data, is that it is often zero-inflated when the data contains more zeros than would be expected (Barry and Welsh, 2002; Zuur *et al.*,

2009). Thus, to empirically model such data, it has been suggested that the data are modelled using zero-altered conditional models (Mullahy, 1986; Welsh *et al.*, 1996; Zuur *et al.*, 2009).

Zero-altered conditional models, also known as hurdle models, are two-component GLMs. The first component models the presence/ absence of a species using the whole dataset with a binomial GLM (Mullahy, 1986; Welsh *et al.*, 1996). The second component determines the magnitude of the presence data from the positive values in the dataset (Gurmu, 1997; Dalrymple *et al.*, 2003; Cunningham and Lindenmayer, 2005). The hurdle model defines each observation's contribution to the likelihood as:

$$\Pr(Y = y) = \begin{cases} \pi_0 & y = 0\\ (1 - \pi_0) \frac{f(y)}{1 - f(0)} & y > 0 \end{cases}$$

where f(y) and f(0) are the probabilities, from an appropriate probability density function for a positive and zero observation, respectively. The denominator, 1 - f(0), conditions the probability to observing at least one animal. In this study, a negative binomial probability density function was used to account for possible over-dispersion of the data (Cunningham and Lindenmayer, 2005; Bilgic and Florkowski, 2007; Zuur *et al.*, 2009).

Hurdle models have the advantage that they can take two ecological processes into account that could influence cetacean distribution and relative abundance in the bay. The first process is presence-absence, and the second, if cetaceans are present, what is the mean abundance. A stepwise procedure, as applied to a GLM or a GAM, was used to determine which predictor variables were significant.

4.7. DEFINING KEY HABITATS FOR CETACEANS IN ALGOA BAY

The fifth objective is to produce a set of recommendations and guidelines for management. To achieve this objective it is necessary to define potential key habitats for the cetaceans in Algoa Bay. The term 'key habitats' was used to identify areas that were *potential* critical habitats, however, more long-term data would be needed to verify this. Important locations and activities have already been identified as mother-calf pair sightings for the mysticetes, and key behaviours (mating, resting and foraging) for all cetacean species. Thus, this data was combined to determine the location of the key habitats.

Each point (representing a location of one of these important locations and activities) was assigned a value of 'one'. This data were then converted into a raster feature, with a cell resolution of one kilometre for both mysticetes and odontocetes. The value of each cell was then determined by the total number of key behaviours, or the number of mother-calf pairs, in each cell. These cells were then combined using the Raster Calculator in Spatial Analyst (with equal weightings). In order to display areas where these habitats were located, the raster layer was converted back into a shape file. A map was then created illustrating the positions of these key habitats over a kernel density map for mysticetes and odontocetes. The two kernel density layers were reclassified into six classes, with the top two classes representing areas of highest density, in order to easily visualise these areas. As a result, the final maps provided an indication of areas where there was a high density of sightings, as well as indicating whether these areas were places that were utilised for important activities such as foraging, resting or calving.

4.8. CONCLUSION

This chapter examined the various methods applied in conducting this research, and the analyses applied to the data. The approaches used in this study provided the means of achieving the objectives set out in Chapter One, the results of which are detailed in the following chapter. Based on these results, key habitats in Algoa Bay are defined in Chapter Six, which will be used to formulate appropriate recommendations for future research, conservation and management strategies in Chapter Seven.

5.1. INTRODUCTION

The findings of this research, conducted in Algoa Bay between March 2009 and July 2010, are presented in this chapter. An overview of the baseline data collected in the study, which was used for further data analysis, is addressed at the outset. A large portion of the chapter is dedicated to the spatial and temporal distribution of the cetaceans in Algoa Bay, and is represented in graphical and cartographic format. The sightings per unit effort (SPUE) and behaviour is provided, in order to further explain the spatial and temporal disfribution for the sightings are subsequently compared to some geographical and anthropogenic variables, forming a foundation for the explanatory modelling of cetacean distribution. These results are then used to establish key habitats for each of the cetacean species in the bay.

5.2. OVERVIEW OF ENVIRONMENTAL PARAMETERS RECORDED

The mean sea state for the study period was Beaufort two ('on' and 'off effort'), with no overall trend observed in the sea state over the study period (Figure 5.1). An increase in cloud cover was generally associated with calmer conditions, and therefore a slight decrease in sea state, this being most common in the summer months (Figure 5.1).

Fluctuations in sea surface temperatures (SST) corresponded with changes in seasons, with 22.5 ^oC recorded during March (after summer) and the lowest mean temperature of 16.8 ^oC during July (during winter) (Figure 5.2). The mean SST for the duration of the study was 19.2 ^oC.

5.3. SEARCH EFFORT

Dedicated boat-based surveys were conducted in Algoa Bay on 57 days between March 2009 and July 2010. Overall, 365 hours were spent at sea, of which 291 hours were spent 'on effort'. A sum total of 2 866 km of tracks were completed during this period. These were predominantly carried out in the western half of the bay between Cape Recife and St Croix Island (Figure 5.3). In addition, surveys were carried out along the Alexandria Dunefield coastline (between Bird Island and Sundays River) (Figure 5.3).



Figure 5.1. Beaufort sea state and cloud cover measured at sea for the duration of the study. Dashed lines represent the mean sea state (Beaufort 2) and cloud cover (5/8).



Figure 5.2. Sea surface temperature (SST) recorded at sea for the duration of the study. Dashed line represents the overall mean SST of 19.2 °C.



Figure 5.3. Survey tracks completed during boat-based surveys. A one square kilometre grid was used, displaying the total distance of tracks covered within each block, to illustrate intensity of effort. Track distance also includes the paths covered in sightings. Moran's I test detected no spatial autocorrelation (p < 0.05).

5.4. CETACEAN SIGHTINGS

5.4.1. Overview of sightings

Cetaceans were sighted 353 times in Algoa Bay; six sightings were of unidentified whales, and one sighting of an unidentified dolphin, a total of 346 confirmed sightings (Table 5.1). Six species were observed over 67 hours of search effort (Table 5.1). The most common species observed were the bottlenose dolphin (odontocete) and the humpback whale (mysticete) (Table 5.1). The other cetaceans observed in the bay were two mysticete species, the southern right whale and the Bryde's whale, and two odontocete species, the humpback dolphin and common dolphin (Table 5.1).

Opportunistic sightings were collected during the same time period as the dedicated boat-based surveys. This data came from both shore-based and boat-based observations with a total of 308

sightings comprised of the same six cetacean species (Table 5.1). The most common species observed opportunistically was the bottlenose dolphin followed by the humpback whale (Table 5.1).

Species	Total opportun- istic. sightings	Total survey sightings	Sightings per survey	Sighting frequency (S.hr ⁻¹)	Mean group size (min/max)	Total time (hh:mm)	Mean time per sighting (hh:mm)
Southern right whale	35	37	0.65	0.13	2.4 (1/8)	09:01	00:15
Humpback whale	44	64	1.12	0.22	2.3 (1/6)	14:10	00:14
Bryde's whale	7	27	0.47	0.09	1.2 (1/2)	05:41	00:18
Bottlenose dolphin	192	183	3.21	0.63	57.8 (1/500)	28:13	00:09
Humpback dolphin	22	28	0.49	0.10	3.3 (1/23)	07:58	00:17
Common dolphin	8	7	0.18	0.03	343.0 (1/800)	02:02	00:17
Total (Mean)	308	346	6.12 (1.02)	1.20 (0.20)		67:05	(00:15)

Table 5.1. Summary of sightings for the six cetacean species observed in Algoa Bay. The results are based on the dedicated boat-based surveys unless otherwise stated.

The mean number of sightings, the sighting frequency, and the total number of sightings per survey, provide an indication of the prevalence of each of the species (Table 5.1). Bottlenose dolphins were sighted most frequently, with a high number of sightings also recorded for humpback whales. The least common species observed was the common dolphin which was seen approximately once every eight surveys.

Group sizes varied extensively among the different species, with common dolphins having the largest groups, with a mean of 343 individuals per sighting (Table 5.1). Groups as large as 500 to 800 animals were observed in bait balls. On two occasions a single common dolphin was seen within a group of bottlenose dolphins, and thus was not included in this group average. Bottlenose dolphins had widely ranging group sizes between one and 500 individuals (Table 5.1). Small groups of one to eight humpback dolphins were observed, with the exception of one sighting consisting of 23 individuals in June 2010 (Table 5.1).

Mysticetes had smaller group sizes as compared to the odontocetes (Table 5.1). Bryde's whales were generally observed singularly, with groups of two to three individuals seen in humpback whales and southern right whales (Table 5.1). Larger groups of up to eight southern right whales were recorded during mating or surface active groups (SAGs). Maps of the spatial pattern of group sizes for each species were created. However, these maps did not show any discernable patterns and are therefore found in Appendix Three.



5.4.2. Spatial distribution of sightings

Figure 5.4. Sightings in Algoa Bay between March 2009 and July 2010. Sightings observed opportunistically are also displayed.

Cetaceans were sighted throughout the survey area (Figure 5.4). Opportunistic sightings illustrate similar distribution patterns, with mysticetes also observed in areas not covered by the survey, i.e. offshore of the Alexandria Dunefield (between PE Port and Bird Islands) (Figure 5.4). Cetacean sightings were associated with all major geographical features of the bay, including rivers/

estuaries, the two ports, Riy Banks, and the two island groups. Moreover, a large number of sightings were located in the area that corresponds with the proposed GAENP MPA (Figure 5.4). There was a notable lack of sightings along a stretch of coastline west of the Sundays River (Figure 5.4).

Southern right whales were sighted in the sheltered, inshore areas of Algoa Bay between PE Port and past the Sundays River (Figure 5.5). Only one sighting was observed in the rocky shore area of Cape Recife, and no southern right whales were seen beyond 26.05 °E. These animals appeared to avoid the areas around Swartkops and Sundays estuaries, with the majority of sightings found south of the Swartkops River, especially around PE Port.

Humpback whales were recorded throughout the survey area, including Riy banks, and the two island groups (Figure 5.5). A number of inshore sightings were observed between PE Port and Cape Recife, and off the Alexandria Dunefield coastline. In contrast to southern right whales, no sightings were seen in the inshore area between the two ports. The opportunistic sightings of humpback whales further illustrated that this species occurred in the offshore area between PE Port and the Bird Islands.

Bryde's whales were predominantly observed in the offshore areas of the western half of the bay, with opportunistic sightings recorded offshore of the Alexandria Dunefield and the Bird Islands (Figure 5.5). Unlike humpback whales, Bryde's whales were not closely associated with the island groups or Riy Banks. Four inshore sightings of Bryde's whales were recorded between Swartkops River and Coega Port, and a large portion of offshore sightings were located seawards of the same area.

Bottlenose dolphins were observed extensively in the inshore areas of Algoa Bay (Figure 5.5). The majority of sightings were recorded between PE Port and Cape Recife, and along the Alexandria Dunefield (beyond Sundays River). Two sighting gaps were observed between PE Port and Swartkops River, and between Coega Port and Sundays River. However, some opportunistic sightings were recorded between Coega Port and north of St Croix. Sightings were also observed around the reef and islands.

Humpback dolphins showed a similar distribution pattern to southern right whales. However, all sightings were exclusively located in the shallow, inshore waters of the bay (Figure 5.5). The

majority of sightings were observed in the rocky shore area between PE Port and Cape Recife. In contrast to other species, humpback dolphins were also frequently sighted on either side of the Swartkops River.



Figure 5.5. Distribution of sightings for each species observed in Algoa Bay March 2009 to July 2010. Data collected during dedicated boat-based surveys is represented by circles and opportunistic data with stars.

Common dolphins were not often sighted in Algoa Bay. This species was observed in the offshore and exposed areas of Algoa Bay, with opportunistic sightings recorded around St Croix and in the offshore areas of the eastern half of the bay (Figure 5.5). Two sightings of a single

common dolphin were briefly observed off the Alexandria Dunefield and east of Coega Port within a large group of bottlenose dolphins. Due to the brevity of these sightings, and without genetic sampling, it was not established whether these two animals were common dolphins or hybrids, and for this reason, were excluded from further data analysis.

5.4.3. Spatial sightings per unit effort (SPUE)

The following maps illustrate the spatial distribution of sightings, corrected for search effort, providing a more meaningful indication of potential key areas for the cetaceans in Algoa Bay. Maps of the number of individuals per unit effort (IPUE) were also created, but as they illustrated similar trends to the SPUE maps, they are not shown in the result but are included in Appendix Four.

Overall, there was no significant pattern of the cetacean SPUE, although a slightly higher cetacean SPUE was found along the rocky shores south of PE Port and along the coast east of 26°12' E (Figure 5.6). High SPUE were found in some offshore areas, where search effort was comparatively lower than the coastal areas. Despite a relatively high search effort along the coastline between the PE Port and Sundays River, the cetacean SPUE was low in this area, especially between Coega Port and Sundays River (Figure 5.6). Riy Banks, St Croix Island and the Bird Islands all had a medium SPUE.

The highest SPUE of southern right whales was recorded south of the Swartkops River (Figure 5.7). Humpback whales showed no distinct pattern of high and low SPUE areas. However, lower SPUE areas did appear to be located further inshore, while higher SPUE areas were observed offshore (Figure 5.8). Higher SPUE areas were also recorded closer to land along the Alexandria Dunefield coastline compared to the western side of Algoa Bay (Figure 5.8). Bryde's whales, like humpback whales, had a higher number of SPUE in the offshore survey areas (at bottom depths greater than 30 m), with the highest SPUE for Bryde's whales illustrated in a group of three high cells located offshore of St Croix Island and Swartkops River (Figure 5.9). Low Bryde's whale SPUE areas were observed along the coastline between PE and Coega Port (Figure 5.9).



Figure 5.6. Spatial distribution of cetacean SPUE. Four levels of cells were used: empty cells, where search effort was carried out without any sightings recorded, and three coloured cells to represent low, medium and a high SPUE. Note: low, medium and high SPUE is a relative value defined for each species and is therefore not directly comparable across the different species.



Figure 5.7. Spatial distribution of southern right whale SPUE.



Figure 5.8. Spatial distribution of humpback whale SPUE.



Figure 5.9. Spatial distribution of Bryde's whale SPUE.



Figure 5.10. Spatial distribution of bottlenose dolphin SPUE.



Figure 5.11. Spatial distribution of humpback dolphin SPUE.

Bottlenose dolphins were seen in relatively high numbers in the coastal areas of Algoa Bay, with the highest SPUE for this species observed east of Sundays River (Figure 5.10). The higher SPUE for humpback dolphins were found south of PE Port, with low SPUE seen throughout the rest of their range in Algoa Bay (Figure 5.11). Due to the low number of common dolphin sightings, a SPUE map for this species was not created.

5.4.4. Temporal sightings per unit effort (SPUE)

Search effort varied during the study period depending on the environmental conditions, with a mean distance of 169 km per month, ranging from 314 km in April 2009 to only 41 km in July 2009 (Figure 5.12). On average 8.6 sightings were observed every 100 km, with a distinct peak in sightings occurring in October 2009 (21.1 SPUE) and June 2010 (27.3 SPUE), with July 2009, November 2009 and February 2010 also having an above average SPUE (Figure 5.12). The lowest number of sightings observed occurred in March and June 2009.

The seasonality of sightings was displayed per month so that the arrival and departure of the two migratory whale species could be more accurately displayed. Southern right whales had an annual mean of 1.3 SPUE (Figure 5.13). This species was first sighted in the bay in winter (August 2009 and June 2010), with a peak in sightings in October (spring) of 9.6 SPUE. No sightings of southern right whales were observed in summer and early autumn.

Humpback whales showed a similar seasonal distribution pattern to southern right whales. They were first observed in Algoa Bay from late May, with an average winter peak SPUE of 2.8 in June (Figure 5.13). After which there was a decline in numbers for the rest of winter/ early spring. A sharp rise in numbers was seen in November and December, with a SPUE of 10.2 and 10.7 respectively. The mean SPUE for humpback whales was 1.9 throughout the year.

Bryde's whales were seen year-round in Algoa Bay, with the exception of late winter and spring (July to November) (Figure 5.13). High numbers were observed in autumn, with the highest SPUE of 3.4 and 3.3 occurring in May 2009 and June 2010 respectively. Bryde's whales had a mean SPUE of one for the duration of the study.



Figure 5.12. Sightings per 100 km search effort (SPUE) for all species, and search effort (SE) per month, for the duration of the study. Dashed line represents the mean SPUE (8.6) and the mean SE (169 km).

Bottlenose dolphins were the only species recorded in every month of the year with a mean of 6.6 SPUE (Figure 5.13). This species illustrated bimodal annual peaks in the SPUE, with the highest peaks occurring in October 2009 with 14.7 SPUE and in June 2010 with 22.3 SPUE. Lower sighting numbers were recorded during autumn in both 2009 and 2010.

Humpback dolphins had a mean sighting rate of 1.2 SPUE (Figure 5.13). The highest number of sightings was observed in February with a SPUE of 3.5. In autumn, very low numbers of humpback dolphins were recorded, with no sightings observed in May in either year. No sightings were recorded in September 2009.

Common dolphins were not frequently observed in Algoa Bay, with an overall average of 0.3 SPUE (Figure 5.13). Sightings were recorded once in September, May and June and twice in February. No sightings of common dolphins were recorded in late winter and early spring, or in early summer. Due to the low number of sightings, it was difficult to determine any general trends in occurrence.



Figure 5.13. Sightings per 100 km effort (SPUE) for each species, and search effort per month. Dashed line represents the mean SPUE.

5.5. GROUP DYNAMICS AND RELATIVE DENSITY

Bottlenose dolphins and common dolphins constitute over 97% of the overall number of cetacean individuals observed, contributing significantly to the overall relative density of cetaceans in Algoa Bay (Table 5.2). Considerably lower relative densities were detected in the other species, with the lowest observed in Bryde's whales. Across all six species, a mean of 31.47 indiv.hr⁻¹ and 3.22 indiv.km⁻¹ was recorded (Table 5.2).

Bryde's whales were the only species where no calves or juveniles were observed (Table 5.2). Calves were present in 40.5% of southern right whale sightings and 67% humpback whale sightings, throughout their observed ranges in Algoa Bay (see Figure 5.14 for locations). On two

occasions a calf or juvenile (possibly a yearling) humpback whale were observed alone. A large number of calves and juveniles were identified in bottlenose dolphin groups compared to common dolphins, where only 11 calves and juveniles were seen in total. No seasonality was observed in the number of calves and juveniles for bottlenose dolphins and humpback dolphins.

Species	Total number calves	Total number of Juveniles	Total number of individuals	Individual frequency (indiv.hr ⁻¹)	Relative individual density (indiv.km ⁻¹)
Southern right whales	15	2	87	0.22	0.02
Humpback whales	43	6	143	0.35	0.04
Bryde's whales	0	0	32	0.08	0.01
Bottlenose dolphins	464	275	10 170	25.14	2.57
Humpback dolphins	6	7	91	0.22	0.02
Common dolphins	8	3	1718	4.25	0.43
Total (Mean)	536	293	12 241	31.47 (5.25)	3.22 (0.54)

Table 5.2. Group dynamics and relative density of the cetaceans in Algoa Bay.



Figure 5.14. The locations of mother-calf pairs for southern right whales and humpback whales.

5.6. OBSERVED BEHAVIOUR OF THE CETACEANS

Each cetacean species utilised Algoa Bay differently (Figure 5.15). Southern right whales spent the majority of their time during winter and spring either milling (49%) or mating (24%), the latter was only recorded during spring (Figures 5.15 and 5.16). This activity was almost exclusively observed around PE port (Figure 5.17).

Humpback whales spent a larger proportion of their time travelling (48%), as well as milling (16%), the latter was observed in increasing measures as the number of calves and juveniles increased towards the end of their wintering season (late spring to early summer), before migrating back to their polar feeding grounds (Figures 5.15 and 5.16). Breaching, flipper slapping and other behaviours defined as 'other' were also frequently observed. The large proportion of 'other' behaviour in autumn is based on two sightings, where breaching and flipper slapping were recorded. Humpback whales were observed resting in the inshore areas of the remote Alexandria Dunefield coastline, especially around Cape Padrone (Figures 5.17). Four instances of foraging occurred during spring, where sharp, fast turns (in a 'zigzag' fashion) were observed just below the surface of the water or at the surface. In three of these sightings, gannets and dolphins were seen foraging in the same area.



Figure 5.15. Behavioural budget for each cetacean species. Note: 'mating' behaviour in bottlenose dolphins was classed as socialising, as these behaviours were closely related in the field.

Bryde's whales were predominantly sighted travelling (45%), milling (28%) and foraging (20%) (Figures 5.15 and 5.16). During one sighting, the behaviour could not be determined as the whale appeared to be attracted to the research vessel. It would approach the vessel and pass sideways underneath, turning around to repeat the behaviour. This continued for the entire duration of the sighting. Most foraging behaviour consisted of horizontal lunges below the surface of the water, with only one instance of vertical lunge feeding observed in a single animal in autumn.



Figure 5.16. The seasonal behavioural budget for the different cetacean species.

Bottlenose dolphins spent the majority of their observed time travelling (43%) and foraging (23%), with little change in their behaviour between seasons (Figures 5.15 and 5.16). Foraging took place along the entire shoreline especially east of Sundays River and south of PE Port, as well as being observed by Riy Banks reef and the two island groups (Figure 5.18). Socialising activities recorded were: tail slapping, breaching, jumping, mating, surfing waves, chasing another individual, fluking (including some observed sailing), aerial somersaults, playing with food (throwing fish into the air), and on one occasion it appeared that a mother was teaching a calf how to forage for food. Socialising activities were often observed in conjunction with travelling or foraging behaviour.

Humpback dolphins were primarily seen foraging (33%) and travelling (41%) throughout the study period (Figures 5.15 and 5.16). Foraging was mainly observed south of PE Port and around the Swartkops River (Figure 5.18). Milling and slow travel were recorded during autumn and winter, along with socialising. Socialising activities observed were: tail-slapping, jumping and spy-hopping. Mating was observed once during December south of PE Port (Figure 5.18).

Common dolphins were predominantly observed foraging in bait balls (70%), or fast travelling (20%) to or from a foraging area (Figures 5.15 and 5.16). Bait balls consisted of gannets circling and diving in the area, and sometimes other marine birds (such as African penguins and roseate terns) and Cape fur seals, as well as other cetacean species such as Bryde's whales and bottlenose dolphins.

5.7. CETACEAN RELATIVE DENSITIES

Several high density areas for cetaceans were recorded in Algoa Bay (Figure 5.19). The highest was along the inshore areas of the Alexandria Dunefield, around 26°10' E, and east of Sundays River. A high number of cetaceans were also observed around St Croix Islands and PE Port.

A high density area for southern right whales was seen around PE Port (Figure 5.20), compared to humpback whales which were found on Riy Banks, around both headlands, and east of $26^{\circ}10^{\circ}$ E along the Alexandria Dunefield coastline (Figure 5.21). The high density area for Bryde's whales was offshore of the Swartkops River (Figure 5.22).


Figure 5.17. Key behaviours for the three whale species observed in Algoa Bay.



Figure 5.18. Key behaviours for the three dolphin species observed in Algoa Bay.

Bottlenose dolphin kernel density mapping revealed one key area located approximately 26°15' E on the Alexandria Dunefield coastline (Figure 5.23). Other relatively densely populated areas were recorded along the entire coastline, with lower densities around the two estuaries and Coega Port. A high density area for humpback dolphins was observed along the coastline of the southwest corner of the bay (south of PE Port) (Figure 5.24) in between the two high density areas for southern right whales and humpback whales. Kernel density mapping was not carried out for common dolphins due to the low number of sightings.

The kernel density maps illustrate similar key areas as the SPUE maps for the respective cetacean species. These two sets of maps take into account both the number of individuals in a sighting as well as the search effort in the area, and thus provide a good indication of key areas for each cetacean species within Algoa Bay.



Figure 5.19. Kernel density map of the six cetacean species observed in Algoa Bay. Density estimates take into account the number of individuals in a sighting (i.e. values represent the number of individuals per square kilometre).



Figure 5.20. Kernel density map of the southern right whales in Algoa Bay. Density estimates take into account the number of individuals in a sighting.



Figure 5.21. Kernel density map of the humpback whales in Algoa Bay. Density estimates take into account the number of individuals in a sighting.



Figure 5.22. Kernel density map of the Bryde's whales in Algoa Bay. Density estimates take into account the number of individuals in a sighting.



Figure 5.23. Kernel density map of the bottlenose dolphins in Algoa Bay. Density estimates take into account the number of individuals in a sighting.



Figure 5.24. Kernel density map of the humpback dolphins in Algoa Bay. Density estimates take into account the number of individuals in a sighting.

5.8. SIGHTINGS AND ENVIRONMENTAL VARIABLES

Sightings were recorded in waters with SST between 13.5 and 24.8 °C, the full range of SST recorded during surveys (Figure 5.25). However, most sightings were observed between 17 and 21 °C. Overall, there was no significant difference between the mean SST for the six cetacean species (ANOVA, df = 5, F = 1.195, p > 0.1).

Sightings were observed in waters with bottom depths of 2.3 to 90.6 m (Figure 5.26). Southern right whales were mostly recorded in waters between 11 and 15 m deep, compared to humpback whales and Bryde's whales which were observed in deeper waters of 18 to 45 m. Bottlenose dolphins were predominantly sighted in bottom depths of zero to 15 m however, they were also observed in bottom depths of 30 m. Humpback dolphins were recorded in shallow waters of less than 15 m deep, which is in contrast to common dolphins which were observed in waters between 11 and 44 m deep. Despite the majority of boat-based surveys being conducted in waters less than 15 m deep (60%), some of the species were still predominantly observed in deeper waters (Figures 5.26 and 5.27). This indicates that the 'inshore' species were well surveyed, while the

'offshore' species with median bottom depths above 15 m (humpback whales, Bryde's whales and common dolphins), were possibly not as well covered during boat-based surveys.



Figure 5.25. The median sea surface temperature (SST) and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles. Environmental (ENV) data was data collected both 'on' and 'off' effort during boat-based surveys.



Figure 5.26. The median bottom depth and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles. Environmental (ENV) data was data collected both 'on' and 'off' effort during boat-based surveys.



Figure 5.27. The proportion of depth classes surveyed in Algoa Bay.

There was a significant difference in the mean bottom depths among the cetacean species (ANOVA, df = 5, F = 59.622, p < 0.05). A Post-hoc Tukey HSD test illustrated highly significant differences in the mean depths between several species (see Table 5.3). The significant difference in depths suggest that there are two groups of cetacean species in Algoa Bay, 'offshore' species (humpback whales, Bryde's whales and common dolphins) that are found in deeper waters, and 'inshore' species (southern right whales, bottlenose dolphins and humpback dolphins). These trends are also illustrated spatially in the distribution maps (Figures 5.4 and 5.5).

	Southern right whale	Humpback whale	Bryde's whale	Bottlenose dolphin	Humpback dolphin	Common dolphin
Southern right whale		p < 0.001	p < 0.001			
Humpback whale	p < 0.001			p < 0.001	p < 0.001	
Bryde's whale	p < 0.001			p < 0.001	p < 0.001	
Bottlenose dolphin		p < 0.001	p < 0.001			p < 0.05
Humpback dolphin		p < 0.001	p < 0.001			p < 0.001
Common dolphin				p < 0.05	p < 0.001	

Table 5.3. The significant differences in mean depths (white blocks) and mean distances from land (grey blocks), between the different cetacean species.

The distance from land data was log-transformed to remove heteroscedasticity (uneven scatter). The distance from land for each species (Figure 5.28) illustrated a similar trend to their associated depths (Figure 5.26). Sightings ranged between 0.1 km and 16.4 km from land (Figure 5.28). A one-way ANOVA revealed a highly significant difference between the log mean distance from land for each species (df = 5, F = 11.171, p < 0.05). A post-hoc Tukey HSD test revealed highly significant differences in the log distances from land between several species, the same significant differences as the mean depths (Table 5.3). These trends are also illustrated spatially in the distribution maps (Figures 5.4 and 5.5).

The mean distance from land and bottom depths were also calculated for mother-calf pairs, for both southern right whales and humpback whales (Figures 5.29 and 5.30). This was compared to sightings without the presence of calves. Southern right whales were observed significantly closer to land and at shallower depths compared to non mother-calf pairs (t(35) = 4.78, p < 0.05 and t(35) = 2.21, p < 0.05). However, there was no significant difference in mean bottom depths and distance from land between humpback whale mother-calf pairs and non-calf groups (t(62) = 0.736, p > 0.1 and t(62) = 1.16, p > 0.1).



Figure 5.28. The log distance from land, and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles.



Figure 5.29. Mean bottom depth (m) for two mysticete species comparing sightings of mothercalf pairs and non mother-calf pairs.



Figure 5.30. Mean distance from land (km) for two mysticete species comparing sightings of mother-calf pairs and non mother-calf pairs.

The distance to rivers data was log-transformed to remove heteroscedasticity. The log distance to rivers for each species illustrated a similar trend to the log distance from land (Figure 5.28), with sightings ranging from 500 m and 23.6 km to the nearest river (Figure 5.31). A one-way ANOVA revealed a significant difference between the log distance to rivers for each species (df = 5, F = 7.897, p < 0.05). A post-hoc Tukey HSD test indicated that humpback whales were observed

significantly further from rivers than southern right whales, bottlenose dolphins and humpback dolphins (p < 0.05).



Figure 5.31. The log distance from rivers and standard deviation for each species observed in Algoa Bay. Outliers are represented by open circles.



Figure 5.32. The proportion of sea-floor substrate associated with each species. Substrate data was made available by SANHO.

There were few noteworthy trends between each cetacean species and the type of sea-floor substrate (Figure 5.32). Five cetacean species were predominantly associated with a mud and sand substrate, with southern right whales utilising the largest proportion of mud substrate (Figure 5.32). Humpback dolphins were the only species which were predominately associated with a rocky substrate (57%) (Figure 5.32). Furthermore, coral and shells, which are linked to reef habitats, were associated with most species, with the exception of humpback and common dolphins (Figure 5.32).

5.9. SIGHTINGS AND ANTHROPOGENIC ACTIVITIES

Southern right whales were seen in three of the four shipping zones, with three sightings in the PE channel (Figure 5.33). No mother-calf pairs for this species were sighted in any of the shipping zones. One humpback whale sighting was recorded in each of the four shipping zones, with calves seen in the all of the zones, except in the offshore anchoring zone (Figure 5.33). Over 20% of Bryde's whales were observed within the shipping areas, more than any other species (Figure 5.33). Only nine of the 183 bottlenose dolphin sightings were seen within a shipping zone, the most (six sightings), being recorded in the PE channel (Figure 5.33). No humpback dolphin sightings were observed within any of the shipping areas, and one common dolphin sighting was recorded in the inshore anchoring area (Figure 5.33).

Southern right whales and humpback whales were recorded in Algoa Bay during austral winter and spring, using the coastal areas as calving and nursing grounds. Both species were seen with calves and juveniles in the bay during this period (Figure 5.14). A significant proportion of these sightings were observed within the proposed GAENP MPA, with 76% of the southern right whales and 63% humpback whales sighted in this area (Figure 5.34).



Figure 5.33. Proportion of cetacean sightings within the shipping zones.



Figure 5.34. Proportion of mother-calf pairs inside and outside the proposed GAENP MPA (numbers in histogram represent actual number of sightings).

5.10 SIGHTING RATE AND RELATIVE ABUNDANCE MODELLING

5.10.1. Introduction

This next section describes the results of statistical analyses conducted on the data. Exploratory data analysis (EDA) in the form of scatterplots was constructed for each of the cetacean species, using several explanatory variables. Further analysis was conducted using GAMs and GLMs to determine cetacean distribution in relation to Beaufort sea state, SST, bottom depth and seasonality. Stepwise forwards/ backwards procedures were conducted on both sets of models, with the final model (lowest AIC) being reviewed. All GAMs performed better than the GLMs (Table 5.4). Hurdle models were then constructed to ascertain whether these variables are also affecting the relative abundance of the different cetacean species in Algoa Bay. As this dataset was restricted by sample size and the limited time-scale of the study, the results should be explored in more detail in future studies. Due to the extremely small number of common dolphin sightings, this species was excluded from the statistical analysis.

5.10.2. Southern right whales

The scatterplots presented in Figure 5.35 reveal significant relationships between southern right whales and several environmental parameters. This species was primarily observed in spring. An increase in sightings were observed in water temperatures ranging between 17 and 19 °C, and at depths of approximately 15 m. Larger groups were observed in colder water temperatures of less than 17 °C. Most sightings were also recorded within two kilometres from land. There was no clear relationship between group size and their proximity to the rivers entering the bay.

The most parsimonious GAM indicates the relationship between southern right whales and their environment and is best described by *Group size* = s(Beaufort) + s(SST) + Depth + Season. No variables were statistically significant (p > 0.05) (Table 5.4 and Figure 5.36). Nonetheless, the model did explain 54% of the variability within the data. Southern right whales were primarily associated with winter and spring, with no sightings observed in the first half of the year (Figure 3.56). The range in Beaufort sea state of one to four and SST between 16 and 20 °C, had an equal influence on the occurrence of southern right whales in the bay (Figure 5.36). This species was more prevalent in shallow waters below 20 m, with no sightings found above 25 m (Figure 5.36). The hurdle model indicated that seasonality and SST strongly influenced the presence of these



animals in the bay, with bottom depth (p < 0.05) and seasonality (p < 0.05), significantly influencing the estimated number of animals observed (Table 5.5).

Figure 5.35. Scatterplots showing the relationship between southern right whale group size (BestEst) and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river. The lower panel contains the scatterplots, and the upper panel provides the results of a Spearman correlation, which illustrates the degree of statistical dependency between two variables.



Figure 5.36. The results of a GAM for southern right whales as a function of Beaufort, SST, bottom depth and seasonality. Results are shown for the best-fit model after a stepwise-selected GAM was completed. Dashed lines represent the two standard error bands. Tick marks (rugs) on the x-axis show the location of data points.

5.10.3. Humpback whales

Humpback whales revealed significant relationships with several environmental parameters (Figure 5.37). This species illustrated a slight preference for warmer waters over 18 °C (Figure 5.37). They were observed in a wide range of bottom depths (mostly less than 40 m) and generally within 10 km from land (Figure 5.37). However, several sightings were noted further offshore, in deeper waters. Two peaks in sightings were observed in Algoa Bay, one in winter and also at the beginning of summer (July and December).

The GAM for humpback whales demonstrated that the relationship between these animals and the environment is best described by *Group size* = p(SST, 4) + s(Depth) + Season (Figure 5.38). This model explained 49% of the variability within the data (Table 5.4). Most sightings were observed between 15 and 21 °C, with a slight increase in sightings in warmer waters, which was also seen in the scatterplot (Figures 5.37 and 5.38). Depth was a highly significant variable in the model (Table 5.4, χ^2 = 29.891, df = 3, p < 0.05), with a steep increase in sightings with increasing depth, and a peak of observations between 20 and 30 m (Figure 5.38). Seasonality also played a

significant role in the presence of humpback whales in the bay (Table 5.4, $\chi^2 = 17.856$, df = 3, p < 0.05), with a decrease in sightings in late spring, and the highest sighting rates in both winter/ early spring and late spring/ early summer, and no sightings observed between late summer and mid-autumn (Figure 5.38). The hurdle model indicated that bottom depth and seasonality also had a highly significant influence (p < 0.05) on the estimated number of humpback whales observed (Table 5.5).



Figure 5.37. Scatterplots showing the relationship between humpback whales and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river.



Figure 5.38. The results of the GAM for humpback whales as a function of SST, bottom depth and seasonality. Results are shown for the best-fit model after a step-wise-selected GAM was completed.

5.10.4. Bryde's whales

The scatterplots for Bryde's whales (Figure 5.39) illustrate the relationships between sightings and environmental parameters. This species was primarily observed in late autumn/ early winter with no sightings recorded between July and November (Figure 5.39). This species was observed in a wide range of SST and depths, with most Bryde's whale sightings occurring in water less than 60 m deep (Figure 5.39).

The results of the GAM indicate that the relationship between Bryde's whales and the environment is best described by *Group size* = p(SST, 4) + p(Depth, 4) + Season (Figure 5.40). Like southern right whales, none of the environmental variables for Bryde's whales were statistically significant (p > 0.05) (Table 5.4). Bryde's whales were observed across all depths and SST, with an apparent increase in sightings in waters above 23 °C (Figure 5.40). Sighting rates appeared to be highest during early-winter and early-summer, similar to humpback whales (Figure 5.40). In contrast to the other two mysticete species that were found in Algoa Bay, Bryde's whales were not generally observed in the second half of the year (Figure 5.40). The hurdle model indicated that SST and depth played a significant (p < 0.05) role in determining the estimated number of animals observed (Table 5.5).



Figure 5.39. Scatterplots showing the relationship between Bryde's whales and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river.



Figure 5.40. The results of a GAM for Bryde's whales as a function of SST, bottom depth and seasonality. Results are shown for the best-fit model after a step-wise-selected GAM was completed.

5.10.5. Bottlenose dolphins

The scatterplots presented in Figure 5.41 reveal several relationships between bottlenose dolphins and the environmental parameters. This species was observed year-round in the bay, with a peak in sightings during winter and spring (Figure 5.41). They were recorded in a wide range of SST, with a slight preference for warmer waters over 18 °C (Figure 5.41). Bottlenose dolphins were generally found in shallow waters less than 20 m deep and within two kilometres from land (Figure 5.41).

The most parsimonious GAM for bottlenose dolphins indicates the relationship between bottlenose dolphins and their environment and is best described as *Group size* = s(Beaufort) + p(SST, 4) + p(Depth,4) + Season, with 66% of the variability explained by the data (Figure 5.42, Table 5.4). Seasonality had a highly significant influence on this model (Table 5.4, χ^2 = 16.135, df = 3, p < 0.05), with an increase in sightings during the second half of the year (Figure 5.42), which coincides with the prevalence of southern right whales and humpback whales in the bay. To some extent, more sightings were observed at a Beaufort sea state of zero or one, however the number of sightings at a sea state of 4 was only fractionally smaller than Beaufort zero and one (Figure 5.42). There also appeared to be an increase in occurrence of bottlenose dolphins with an increase in temperatures, between 16 and 22 °C (Figure 5.42). Figure 5.39 illustrates that the distribution of bottlenose dolphins was significantly influenced by bottom depths ($\chi^2 = 6.315$, df = 3, p < 0.05), with these animals only being observed between three and 30 m, despite surveys being conducted in deeper waters (Figure 5.42 and Table 5.4). The results of the hurdle model demonstrated that depth and seasonality played a highly significant role (p < 0.05) in the overall estimated relative abundance of bottlenose dolphins in the bay, with higher numbers seen during the second half of the year and in waters less than 20 m deep (Table 5.5).



Figure 5.41. Scatterplots showing the relationship between bottlenose dolphins and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river.



Figure 5.42. The results of a GAM for bottlenose dolphins as a function of Beaufort, SST, bottom depth and seasonality. Results are shown for the best-fit model after a step-wise-selected GAM was completed.

5.10.6. Humpback dolphins

The last set of scatterplots illustrates the relationship between humpback dolphins and several environmental parameters (Figure 5.43). More humpback dolphins were observed between 18 and 22 °C (Figure 5.43). Most of the sightings were also found in extremely shallow waters (less than 10 m deep) and close to shore, within 600 m (Figure 5.43).

The results of the GAM indicate that the relationship between humpback dolphins and their environment is best described by *Group size* = p(Beaufort, 3) + p(Depth, 4) + Season (Figure 5.44). Like southern right whales and Bryde's whales, none of the environmental variables for humpback dolphins were statistically significant in the most parsimonious GAM (Table 5.4). However, the GAM did explain 59% of the variability within the data (Table 5.4). No significant trend was seen between the presence of humpback dolphins and Beaufort sea state, with a slight decrease in sightings observed at Beaufort zero (Figure 5.44). This species was only observed in waters less than 15 m deep, and in this narrow band of shallow water, there was no upward or downward trend of humpback dolphin occurrence (Figure 5.44). Humpback dolphins were seen throughout the year, with the lowest numbers in late autumn and highest numbers during winter (Figure 5.44). The hurdle model indicated that depth was a very good explanatory variable for

both the presence of humpback dolphins (p < 0.05) as well as the relative abundance (p < 0.05) of these animals in Algoa Bay (Table 5.5).



Figure 5.43. Scatterplots showing the relationship between humpback dolphins and Beaufort sea state, SST, bottom depth, seasonality, distance from land and distance from the nearest river.



Figure 5.44. The results of a GAM for humpback dolphins as a function of Beaufort, bottom depth and seasonality. Results are shown for the best-fit model after a step-wise-selected GAM was completed.

Table 5.4. Comparison of the variables included in the final sighting rate models (GAMs and GLMs) for each species. Linear fits are represented by 'L 1', smoothing splines by 'S #', and polynomial fits by 'P #', where '#' indicates the associated degrees of freedom. The pseudo R^2 value represents a goodness-of-fit measure. P-values were displayed if some level of significance was shown: 0 '***' 0.001 '**' 0.01'*' and 0.05 '.'.

	Model	Explanatory variables						
Species		Beaufort	SST	Depth	sin	cos	Pseudo R ²	AIC
Southern right whale	GAM	S 3 0.066 ⁻	S 3 0.064 ⁻	L 1	L 1	L 1	0.54	150.57
	GLM	-	-	0.007**	0.000***	0.000***	0.62	152.14
Humpback whale	GAM	-	P 4	S 3***	S 3 ***	L 1	0.49	194.05
	GLM	0.000***	-	-	0.065	0.000***	0.72	250.47
Bryde's whale	GAM		P 4	P 4	S 3 0.080 ⁻	L 1	0.61	121.05
	GLM	-	0.002**	0.008**	-	-	0.91	144.28
Bottlenose dolphin	GAM	S 1 0.097 [.]	P 4	P 4	-	S 3 0.001**	0.66	430.05
	GLM	0.003**	0.015*	0.000***		0.000***	0.78	476.01
Humpback dolphin	GAM	P 3	-	P 4	L 1	P 3	0.59	138.02
	GLM	0.143	0.028*	0.000***	-	0.003**	0.69	144.08

Table 5.5. Outcomes of the hurdle models for the different cetacean species. Only variables that were significant in the two stages of the model were noted with their appropriate levels of significance (0 '***' 0.001 '**' 0.01'*' and 0.05 '.').

Hurdlos	Stage	Explanatory variables					
11ul ults		Beaufort	SST	Depth	sin	cos	
Southern right whale	1	-	0.0195*	-	-	0.05 [.]	
	2	-	-	0.008**	0.000***	0.000***	
Humpback whale	1	-	-	-	-	-	
	2	-	-	0.000***	-	0.000***	
Bryde's whale	1	-	-	-	-	-	
	2	-	0.007**	0.008**	-	-	
Bottlenose dolphin	1	-	-	-	-	0.063 [.]	
	2	0.003**	0.026*	0.000***	-	0.000***	
Humpback dolphin	1	-	0.329*	0.007**	-	-	
	2	-	0.050*	0.000***	-	0.017*	

5.10.7. Summary of the statistical models

The results of the exploratory data analysis, GAMs, GLMs and the hurdle models indicated that bottom depth and seasonality were the key explanatory variables (that having the greatest effect on the mean response) in determining sighting rate and group size of cetaceans in Algoa Bay, confirming statistically what was illustrated in previous analyses in this chapter. In summary, more mysticetes and odontocetes were observed in the second half of the year, from mid-winter to early summer (July to December). Cetacean species also appeared to utilise and favour different depth classes, with distinctions between inshore and offshore species. Beaufort sea state did not appear to significantly influence the number of sightings for each species directly. However, the inclusion of Beaufort sea state into most of the final models suggests that this variable is correlated with other factors that influence cetaceans, such as seasonality.

5.11. DEFINING POTENTIAL KEY HABITATS FOR THE CETACEANS IN ALGOA BAY

Key habitats (potential critical habitats) were defined by looking at high density areas, places where key behaviours occurred (foraging, resting and mating), and the location of mother calf pairs. As the data for this research was only captured over a relatively short-period of time, these maps (Figures 5.45 and 5.46) provide only a preliminary insight into the key habitats for the mysticetes and odontocetes in Algoa Bay.

For mysticetes, five areas (A, B, D, E and G) were identified as key habitats (Figure 5.45). Areas C and F have lower densities, but were considered relatively important in two variables (Figure 5.45). This map can be interpreted further when taking into account all the geographical variables that have been compared to cetacean distribution in this chapter, including bottom depth, which was an important factor in the cetacean distribution in the statistical models. Area B (in bottom depths less than 20 m) was considered to be the key habitat for southern right whales (Figure 5.45). This area also possibly extends north-westwards towards the Swartkops River, with a high number of sightings and mating activities observed here (Figure 5.45). A secondary, smaller key habitat for southern right whales falls within area E, where a high number of mother-calf pairs for this species was located (Figure 5.45). In contrast to this, key habitats for humpback whales were located along the more exposed edges of the bay in A, F and G, with area A being the farthest offshore (Figure 5.45). Areas D and E were also moderately utilised by this species (Figure 5.45). As the total number of Bryde's whales was comparatively low compared to the other two mysticete species, along with the lack of calves present in sightings, it was more difficult to establish a key area for Bryde's whales. However, Bryde's whales were seen relatively often in area C, while foraging and resting (Figure 5.45).

The high number of sightings for bottlenose dolphins and the significantly lower number of sightings for common dolphins, make it more difficult to distinguish key areas for the odontocete species (Figure 5.46). Nevertheless, several locations are more extensively used by odontocetes compared to other areas in Algoa Bay (Figure 5.46). The coastal nature and shallow habitats for both bottlenose dolphins and humpback dolphins has been revealed in various ways. Although area H is not the most densely populated area, a number of key activities were located here, all of which were associated with bottom depths of less than 15 m (Figure 5.46). In addition, a significant proportion of humpback dolphin sightings were located in area H. This, as well as the presence of important activities such as mating and foraging along these rocky shores, makes this area very important for humpback dolphins (Figure 5.46). Bottlenose dolphins extensively utilised the sandy, shallow waters along the Alexandria Dunefield coastline (east of Sundays River). Additionally, two key areas (K and L) where a very high number of sightings, as well as foraging activities, were located within this region (Figure 5.46). These areas were also

associated with shallow waters; primarily within bottom depths of less than 20 m (Figure 5.46). Areas J (St Croix Island), and to a lesser extent, I, have also been considered important as a relatively large amount of foraging was recorded there (Figure 5.46).



Figure 5.45. Potential key habitats (A – G) for the mysticetes in Algoa Bay.



Figure 5.46. Potential key habitats (H - L) for the odontocetes in Algoa Bay.

5.12. SUMMARY OF FINDINGS

Six cetacean species were observed in Algoa Bay during dedicated boat-based surveys and opportunistic data collection between March 2009 and July 2010. A total of 346 sightings were recorded, with over 12 000 individuals observed, including calves and juveniles. Mother-calf pairs for the two migratory whale species were seen throughout their range in the bay, with significantly higher numbers in the proposed GAENP MPA. A number of behaviours were recorded in the different species, including behaviours such as foraging and mating, which are indicative of potential key habitats. Some sightings were recorded within the various shipping zones.

Southern right whales were only recorded during winter and spring, with a peak in sightings during spring, where mating was the most observed behaviour. This species was recorded in the sheltered, inshore (shallow) areas between the two ports, with the associated substrate primarily consisting of sand and mud.

Humpback whales were the second most observed species in the bay, with all sightings recorded between June and December. Two peaks of observations were noted in June and November/ December. This species was seen throughout the survey area, and was recorded opportunistically offshore of the Alexandria Dunefield. Humpback whales were the least closely associated with rivers compared to the other species, and were generally found over sand, rock or coral substrate. These animals were frequently observed breaching, with some opportunistic foraging observed.

Bryde's whales were observed in Algoa Bay during the austral summer and autumn, with a peak of sightings occurring in May. These animals were recorded in the inshore waters between the two ports, and in the offshore zone, where sand, rock and mud substrate dominates. Within these two zones, they were found intermittently within the Coega channel and the two anchoring areas. Bryde's whales, in comparison to the other species, were located in the deepest waters of Algoa Bay, with no calves and juveniles recorded. This species was predominantly observed travelling, milling and foraging.

Bottlenose dolphins were the most common species recorded in Algoa Bay, with the highest number of individuals, calves and juveniles observed. This species was recorded year-round, with peaks in sightings during winter and spring, and a decrease in sightings during autumn. They were sighted throughout the study area, with a greater inshore distribution compared to humpback whales. The highest number of sightings for this species was recorded in the coastal zone of the Alexandria Dunefield, which is primarily comprised of sandy substrate. Bottlenose dolphins were predominantly observed travelling, milling and foraging.

Humpback dolphins were recorded in the coastal zone of the western half of the bay, with the majority of sightings observed along the rocky shores south of PE Port. This species was recorded closest to the shore out of the six cetacean species, with all sightings occurring in depths of less than 15 m. The lowest number of sightings was observed during autumn, with increased numbers noted in winter and early summer. Group sizes consisted of approximately four individuals, with one sighting of 23 humpback dolphins. This species was most often observed travelling and foraging, with socialising occurring in autumn and winter.

Common dolphins were only recorded seven times in Algoa Bay, two sightings of which were of single individuals within a group of bottlenose dolphins. However, this species contributed to the second highest number of individuals in the bay due to their large group sizes. These dolphins were predominantly observed foraging in bait balls with other marine species in the offshore and exposed areas of the bay.

In conclusion, this chapter presented the findings through the use of tools such as GIS and statistics. These results address objectives three and four, which were to map and investigate the spatial and temporal distribution and behaviour of the cetacean species, and to relate this distribution to the environmental and anthropogenic variables. The following chapter will further develop and explain these findings.

6.1. INTRODUCTION

Six cetacean species were observed throughout the surveyed area in Algoa Bay, displaying a number of different spatial and temporal distribution patterns. These distribution patterns, together with the influence of group dynamics such as the predominant behaviour, will be discussed in order to meet objective three, which was to determine the spatial and temporal distribution and behaviour of the cetacean species. These patterns indicate habitat preferences for each species within the bay. The relationship between these habitat preferences and the surrounding physical environment is addressed through the outcomes of objective four, which relates the spatial and temporal distribution and behaviour of these normal distribution and behaviour of these environmental and anthropogenic variables.

Existing literature is referred to, to demonstrate the present state of knowledge, and compare the measured biological and geographical variables, to cetacean distribution within Algoa Bay. Each species is discussed separately, with an overview of the general spatial and temporal distributions of the different cetacean species occurrences in the bay. This is followed by the discussion of more specific group dynamics and behaviours which influence their distribution patterns. The final section for each species builds on these two previous aspects in order to identify their habitat preferences within Algoa Bay and how this relates to general habitat preferences within their range. Anthropogenic activities that have affected, or may potentially affect the distribution of cetaceans are also discussed for mysticetes and odontocetes. This discussion is intended to form the foundation of a more comprehensive framework for future management, conservation and research decisions.

6.2. SOUTHERN RIGHT WHALE

6.2.1. General spatial and temporal distribution patterns

Previous data on the southern right whale's utilisation of the South African coastline was obtained from aerial surveys conducted between 1970 and 2000 (Best, 2000; Elwen and Best, 2004). These surveys established that this species was primarily recorded on the south and west coasts of the country, especially between Walker Bay and Plettenberg Bay, with very few

sightings east of Cape Padrone (Best, 1990; Elwen and Best, 2004). The results of this study indicate that this situation has changed over the last decade, with a greater number of southern right whales recorded in Algoa Bay. However, as this is a local study, it is not evident whether the presence of these animals in the bay is a result of a shift in distribution away from other areas, or whether it is a result of their large population growth recorded by Best *et al.* (2001), that has resulted in an expansion of their range along the coast (Figure 6.1) (Best, 2010: Pers. Comm.).



Figure 6.1. Diagram illustrating the distribution of southern right whales in southern Africa. The grey area represents the previously known limits (Baia dos Tigres to Maputo Bay). The blue and orange areas illustrate the previous core distribution and the current range expansion, respectively. The arrows represent the eastward and westward movement of this species at the beginning and end of their wintering season.

During the 'whaling' period, southern right whales were caught along the entire coastline of South Africa, within their winter range which extended from Baia dos Tigres, Angola, in the west, to Maputo Bay, Mozambique, in the east (Figure 2.1) (Best, 2007; Reilly *et al.*, 2008b). Therefore, the recent increase in numbers at the end of their modern day range suggests that, with the population growth over the last three decades, this species is extending beyond its present day range to its previously known limits (Figure 6.1) (Belgrano *et al.*, 2008; Mate *et al.*, 2010).

Patterns of range expansion for this species have already been observed up the west coast of South Africa (Best, 1990), and off the Península Valdés in Argentina (Rowntree *et al.*, 2001).

In this study, southern right whales were recorded in Algoa Bay between June and December, throughout the duration of their previously known wintering months off the coast of South Africa (Figure 5.13) (see: Best, 1990; Best *et al.*, 2003). This is comparable to other subtropical breeding grounds in the southern hemisphere such as Argentina, Uruguay, Australia and New Zealand, where sightings are recorded in winter and spring (Burnell and Bryden, 1997; Rowntree *et al.*, 2001; Patenaude, 2003; Costa *et al.*, 2005; Belgrano *et al.*, 2008).

The sighting frequency for southern right whales was 0.13 s.hr⁻¹. This frequency is relatively high compared to the other species, particularly since this species is only being observed five to six months of the year (Table 5.1). A intensive shore-based study in Saldanha Bay (which had more hours 'on effort'), calculated a sighting frequency of 0.37 s.hr⁻¹ (Barendse *et al.*, 2010), which is almost three times higher than what was recorded in Algoa Bay. The results of these two studies could again indicate that the number of sightings in this region have increased in recent years, which is indicative of range expansion and an increase in numbers within this range.

6.2.2. Group dynamics and behaviour

The mean group size for southern right whales was 2.4 individuals (Table 5.1), slightly larger than those recorded by Best (1990a), who estimated an average of 1.51 individuals per sighting. Groups of two or more individuals are common in the subtropical breeding grounds (Best, 1990; Costa *et al.*, 2005), which is in contrast to their Antarctic feeding grounds where only 15.4% of sightings consist of more than one individual (Best *et al.*, 2003).

Aerial surveys between 1969 and 1987 recorded no mother-calf pairs in Algoa Bay, only adults (Best, 2000). This is in sharp contrast to what was observed in the bay during the present study, which recorded mother-calf pairs in 40% of the sightings (Table 5.2). In this study, a peak in sightings occurred approximately two months after the previously estimated peak birthing time (August) for southern right whales observed off the South African coastline (see: Best, 1994; Best, 2007) (Figure 5.13). A large portion of these sightings during October were mother-calf pairs, which suggests that these animals move into the bay after giving birth. This was confirmed by field observations, where slightly larger calves (six to seven metres in length) were recorded.

Studies in the region have indicated a westward movement along the south coast (Figure 6.1) (Best, 2000; Mate *et al.*, 2010), which would imply that these animals were born east of Algoa Bay, and slowly moved westwards during the season, before returning to their summer feeding grounds at the beginning of summer. However, three extremely small calves were also seen in the bay in June and July 2010 (approximately four to five metres in length), one with neonatal folds, indicating that some of the observed calves were probably born in Algoa Bay.

A large proportion of socialising behaviour observed in spring was a result of an increase in mating activities, often linked to 'surface active groups' (SAGs) (Figure 5.16). SAGs are groups of between two and ten animals that perform a variety of behaviours including splashing, spy-hopping, rolling, and exposing of their bellies and flippers at the surface of the water (Best *et al.*, 2003; Patenaude, 2003). Such activities were seen five times in the bay, with a maximum group size of eight individuals. The seasonality of these larger groups was also shown to be significant in the final hurdle model (p < 0.05; Table 5.5). Although these socialising/ mating behaviours are relatively common on their wintering grounds, this is in sharp contrast to the activities of these animals in their polar feeding grounds, where almost no socialising has been observed (except between a mother and calf) (Best *et al.*, 2003).

6.2.3. Description of the habitat preference within Algoa Bay

Southern right whales were recorded in the sheltered and sandy areas of Algoa Bay (Figures 5.32 and 3.2). This is comparable to findings in other studies which also noted the favouring of similar habitats along the South African coastline, with this species avoiding the more exposed and rocky shores (Figure 5.5) (Best *et al.*, 2001a; Elwen and Best, 2004). The lack of sightings along the rocky shore region on the south-west corner of the bay could also be attributed to the steeper bathymetry in this area, as these animals tend to prefer a gentle sloping/ flat sea-floor (Best *et al.*, 2001a). The high proportion of sightings associated with sand and mud could also explain why southern right whales, out of all the species, were the most closely associated with river mouths (Figure 5.31), which tend to have finer sediments and gentle gradients around them. Most sightings were found near the two ports, both of which have rivers that enter them (Figure 5.5). The long harbour walls create a buffer to the ocean, and result in a sheltered area from the wind and swell, providing an ideal habitat for this species.

Best (2000) suggested that 'nursery' and 'mating' areas existed in separate patches along the south coast of South Africa, and previous data did not indicate that Algoa Bay served as either (Best, 1990; Elwen and Best, 2004). The results of this study suggest that this bay now has both nursery and mating areas (Figure 5.17), with a mating area existing around the PE Port, and the presence of mother-calf pairs recorded throughout their observed range in Algoa Bay (Figure 5.14). Most of the mother-calf pairs (76%) were observed in the proposed GAENP MPA (Figure 5.34). Recent shifts in nursery and mating grounds have also been seen in other regions, such as off the coast of Valdés, Argentina (Rowntree *et al.*, 2001). However, the results of this study does not suggest a well defined segregation between a nursery and mating area in the bay as suggested in previous studies (Payne, 1986; Best, 1990; Elwen and Best, 2004), with mother-calf pairs in Algoa Bay overlapping with the core mating area (Figures 5.14 and 5.17).

The results of this research were similar to Elwen and Best (2004), with mother-calf pairs being located both closer to shore and in shallower depths compared to sightings consisting of only adults (p < 0.05; Figures 5.29 and 5.30). One hypothesis is that the mother-calf pairs utilise the shallow surf zone to avoid predation by killer whales (which are occasionally observed along the coastline of South Africa), as the waves mask the noise of these animals, and the shallow water prevents an attack from below (Thomas and Taber, 1984; Sironi *et al.*, 2008). Southern right whales could also utilise the more shallow areas to avoid interactions with other animals of the same species, which could resulting in possible injury to the calf or interruption of suckling (Elwen and Best, 2004).

Southern right whales were mainly observed at depths of ten to 15 m (Figure 5.26), and approximately 700 m from land (Figure 5.28). This is similar to other findings for this species in the region (Payne, 1986; Best, 1990; Elwen and Best, 2004). These habitat preferences were explored further with the use of a hurdle model which indicated that depth did play a significant role in the relative abundance of southern right whales in Algoa Bay (p < 0.05; Table 5.5).

6.3. HUMPBACK WHALE

6.3.1. General spatial and temporal distribution patterns

Humpback whales are known to use subtropical coastlines as a migratory corridor to their main wintering areas, which exist from Angola and Mozambique northwards (Dawbin, 1966; Best,

2007). Although these animals have a cosmopolitan distribution (Figure 2.2) (MacLeod *et al.*, 2005), humpback whales exhibited a distinct seasonality in Algoa Bay, similar to the southern right whales, with a slightly longer season extending from May to December (Figure 5.14). Studies from different regions in the world have also shown a strong seasonality in humpback whale sightings (see: Winn and Reichley, 1985; Jefferson and Schiro, 1997). In South Africa, a shift in peak sightings, was recorded along the different latitudes (Best *et al.*, 1998; Barendse *et al.*, 2010). These peaks occurred in August at Cape Vidal (east coast), October at Saldanha Bay, and December at Knysna (south coast), for the southward migration of these whales (Figure 6.2) (Best *et al.*, 1998). The tropical limit of their east coast wintering ground is thought to be around central Mozambique (Findlay *et al.*, 1994; Barendse *et al.*, 2010).

Humpback whale sightings increased towards the end of their wintering season, with a lower peak of sightings in June, and the highest number of sightings recorded during November and December (Figure 5.13). This trend is possibly due to the initial arrival of this species at the beginning of the season, resulting in a winter peak of abundance in June, before they continue up the coastline to the tropical areas (Best *et al.*, 1998; Best, 2007). The continuation of sightings throughout the season mainly consisted of adults, which indicates that they might not travel the full extension of their migratory route, and instead, remain along the South African coastline. The sharp increase in sightings in November and December (ten times the SPUE recorded during July and October), mark the end of the winter breeding season, and the southward movement of these animals (Figure 5.13) which coincides with previous studies in the region (see: Best *et al.*, 1998; Best, 2007).

A sighting frequency of 0.22 s.hr⁻¹ was recorded for humpback whales in this study (Table 5.1), which is higher than the total frequency of 0.12 s.hr⁻¹ recorded by Best *et al.* (1998) in coastal and offshore surveys off southern Africa between 19 and 53 °E (Knysna to East Madagascar). In contrast to Algoa Bay, humpback whales in Saldanha Bay were also observed less frequently (0.16 s.hr⁻¹) (Barendse *et al.*, 2010). This shows that they are found in varying numbers around South Africa and possibly indicates that Algoa Bay is an important site.

The mean SPUE for humpback whales (1.9) was high (Figure 5.13), considering they were not observed in the bay between January and late May. The SPUE for these whales ranged from 0.7 to 10.2. In contrast, the SPUE measured for humpback whales in Mozambique was approximately 1.25 between August and September (Findlay *et al.*, 1994). When taking into

account the actual number of individuals observed and the distance covered during the survey, the relative density of humpback whales is low (0.04 indiv.km⁻¹; Table 5.2) compared to a relative density of 0.185 indiv.km⁻¹ recorded for this species in Madagascar by Best and Sekiguchi (1996). However, the high SPUE in Algoa Bay does indicate that this area is potentially important for these animals along the migratory route.



Figure 6.2. Migratory route of humpback whales on the east coast of South Africa. The grey and black arrows represent the north and southward migratory route, respectively. Months where peak sightings occur are displayed in the diagram.

6.3.2. Group dynamics and behaviour

Group sizes were between one and six individuals, with a mean of 2.4 (Table 5.1). This is similar to records in the wintering grounds of 1.4 to 2.2 animals (Mignucci-Giannoni, 1998; Best *et al.*, 1999; Barendse *et al.*, 2010). Group sizes at the beginning of the season were between one and three individuals with very few mother-calf pairs. This is in sharp contrast to the end of the season, where groups were larger (up to six individuals), and most of which contained at least one calf. Barendse *et al.* (2010) also established that group sizes change with season, with larger groups observed in mid-spring and early summer. These observations were further clarified by

the results of the hurdle model which illustrated that seasonality played a significant role in both the occurrence and the number of animals in the bay (p < 0.05; Table 5.5).

This change in group size and composition could be a result of the sequence of migration that these animals undertake. Humpback whales are believed to migrate between their feeding and breeding grounds in a particular order, with females with yearlings generally migrating northwards first, followed by young and mature adults, and lastly, the pregnant females (Dawbin, 1966; Best, 2007). A similar sequence is thought to occur on the southwards migration, with the females with new calves leaving last (Dawbin, 1966; Barendse *et al.*, 2010). This could explain the changes in group composition throughout the austral winter.

However, this does not explain the high number of mother-calf sightings observed in Algoa Bay (over 60%; Table 5.2). This was much higher than studies in Puerto Rico, where calves were present in 12% of the sightings, and 3.7% of the sightings in Madagascar (Best and Sekiguchi, 1996; Mignucci-Giannoni, 1998). Two sightings in Algoa Bay were also of very small calves (five to six metres), one of which had neonatal folds. This suggests that some humpback whale births could be occurring off the south coast of South Africa.

The presence of humpback whale mother-calf pairs in Algoa Bay (Figure 5.14) suggests that they utilise the relatively protected habitat that a bay provides, which are not found on the east coast of the country (Bremner, 1983), until their calves are large or strong enough to migrate back to their summer feeding grounds in Antarctica. It is not known whether these animals would then continue travelling westwards along the south coast of the country, or whether they migrate directly south (Best, 2007).

Alongside the change in group sizes and composition, they appear to change their behaviour on the return route along South Africa (Figure 5.16). At the beginning of their wintering season, breaching, flipper slapping and other similar 'play' behaviours were dominant at sightings. Towards the end of the season, greater time was spent travelling, milling and resting. The decrease in highly active behaviours and the increase in milling and resting (especially in groups with small calves), is possibly due to these animals conserving energy before migrating back to their polar feeding grounds.
Three instances of foraging were observed during November (all with calves), two of which were in large groups (four to six individuals) (Figure 5.16). Dolphins and gannets were also recorded foraging in the area with the latter two groups. Laws (1977) suggested that opportunistic foraging could be due to appropriate food being available, and is therefore consumed to store up extra blubber reserves for suckling and migrating. All groups showed similar behaviours with spurts of travelling in a zigzag fashion (sharp, sudden turns), with milling/ slow travel in between, which according to Best, P.B. (2010: Pers. Comm.) is suggestive of feeding. Similar observations have been recorded in Saldanha Bay and Cape Columbine (Findlay and Best, 1995; Best, 2007; Barendse *et al.*, 2010).

6.3.3. Description of the habitat preference within Algoa Bay

Humpback whales were sighted throughout Algoa Bay, in both coastal and more offshore areas, with a preference for certain habitats (Riy Banks, along Woody Cape and at Cape Padrone) (Figures 5.5. 5.8 and 5.21). However, this preference for certain habitats within the bay was not as noticeable as other studies, where only one or two key areas were occupied within the study area (Mignucci-Giannoni, 1998). In Algoa Bay, this species was associated with the major reef (Riy Banks) and the two island groups. They were also sighted at the two headlands bordering the bay. These areas have relatively steep bottom topography (i.e. greater changes in bathymetry). The oceanic ridge ('Recife Bird Ridge') extends between the two headlands and Riy Banks could provide a path for navigation as the humpback whales migrate along the coastline (Figure 6.3). Looking at the behaviour of the animals along the ridge confirms this idea, as a number of animals were observed between Cape Recife and Riy Banks, travelling in the direction of the reef. The observed behaviour of humpback whales on the reef was also different, with most animals seen milling and breaching, before moving away.

Unlike humpback whales in Puerto Rico which were primarily recorded in sheltered areas (Mignucci-Giannoni, 1998), a high number of sightings in this study were observed along the exposed Woody Cape coastline. A significant proportion (63%) of the observed mother-calf pairs were located on the eastern coastline of the bay (inside the GAENP MPA) (Figure 5.34), an area which experiences some of the largest waves in the bay (Pers. Obs.). The hypotheses as to why mother-calf pairs utilise shallow, surf zones was outlined previously with regard to southern right whales.



Figure 6.3. Postulated movement of humpback whales in the Algoa Bay region.

The distance from land recorded for humpback whales ranged from 1.5 to 7.8 km (Figure 5.28). This is similar to the ranges observed at Cape Vidal, and Saldanha Bay (Best *et al.*, 1998; Barendse *et al.*, 2010). Humpback whale mother-calf pairs, like southern right whales, were also found slightly closer inshore than the other sightings of the same species. However, this was not significant (p > 0.1; Figure 5.30). Mother-calf pairs have been observed closer to land in other areas such as the São Tomé and Príncipe archipelago and Puerto Rico (Mignucci-Giannoni, 1998; Picanço *et al.*, 2009).

The distance from land also relates to the depth at which this species was sighted. In Algoa Bay, most sightings were detected in waters greater than 20 m deep, with a mean bottom depth of 27.6 m (Figure 5.26). This is similar to the distribution of humpback whales in Mozambique which were generally observed in waters greater than 18 m deep (Findlay *et al.*, 1994). Depth was also considered a significantly important variable for presence of humpback whales in the bay (p < 0.05; Table 5.4) and their relative abundance (p < 0.05; Table 5.5) in the GAM and hurdle models. These results suggest that perhaps the coastal nature of this survey has only recorded a relatively small percentage of the total population that are utilising the Algoa Bay region.

6.4. BRYDE'S WHALES

6.4.1. General spatial and temporal distribution patterns

Bryde's whales were predominantly observed in the western half of the bay during boat-based surveys, with a few opportunistic sightings off the Bird Islands. This included offshore sightings between Cape Recife and Riy Banks, and more inshore sightings between Swartkops River and St Croix Island (Figure 5.5). These animals are very elusive in nature (O'Callaghan and Baker, 2002), and it was difficult to observe their head or pectoral flippers. This would have confirmed whether all the sightings were Bryde's whales, or whether some consisted of minke whales (*B. acutorostrata*) or sei whales (*B. borealis*), which are thought to occur in the region, and are very similar morphometrically (Best, 1985; Best, 2001; Kato and Perrin, 2009). However, the few rostrums that were observed had the characteristic three ridges for Bryde's whales and like-Bryde's whales were grouped together.

The inshore form of Bryde's whales is thought to be located along the west and southern coastlines of South Africa within the 200 m isobath (Best et al., 1984; Best, 2001). This species occupied Algoa Bay in summer and autumn/ early winter (Figure 5.13). Bryde's whales along the west coast of South Africa are observed during autumn and winter, before migrating downwards towards the south coast during spring (proposed migratory pattern is illustrated in Figure 6.4) (Best et al., 1984; Best, 2001). Small-scale migrations to warmer waters along either the west or the east coast of South Africa, including migrations with the annual Sardine Run, have previously been proposed by Best (2001) and O'Donoghue et al. (2010b). The lack of sightings in Algoa Bay during late winter and spring indicates that the animals observed in this study could undergo similar migrations (Figure 6.4). Although little information has been scientifically recorded for the Sardine Run until recently, it is thought to begin east of Algoa Bay during winter, and move up the east coast, continuing into spring (van der Lingen et al., 2010). Other marine mammals such as bottlenose dolphins and common dolphins have been associated with this phenomenon (O'Donoghue et al., 2010b; van der Lingen et al., 2010). This small-scale seasonal movement of Bryde's whales is probably governed by the distribution of their prey, which undergo migrations at a similar spatial scale along the coast of South Africa (Omura, 1959; Best, 1977; O'Callaghan and Baker, 2002; Penry, 2010).



Figure 6.4. Proposed movement of Bryde's whales along South Africa.

The absence of sightings during late winter and spring could be a result of the short time-span over which this study took place, although, numbers are expected to be considerably lower during this period. A similar trend in the seasonal fluctuations of Bryde's whale sightings was observed in Plettenberg Bay, where highest numbers occurred during autumn, with a significant drop in sightings during winter (Penry, 2010). Similar patterns were also observed along the Brazilian coastline and in the Hauraki Gulf in New Zealand (Zerbini *et al.*, 1997; Wiseman, 2008).

Bryde's whales had a low number of sightings compared to the other two mysticete species observed in this study (Table 5.1). The sighting frequency for these animals was 0.09 s.hr⁻¹, higher than the 0.03 s.hr⁻¹ that was recorded in the Hauraki Gulf, New Zealand (O'Callaghan and Baker, 2002). In Algoa Bay, this species had a mean SPUE of one, which is higher than the SPUE of 0.02 recorded in the north-central Gulf of Mexico (Mullin *et al.*, 1994). The relatively small number of sightings for this species in several surveys (including the present study) could be attributed to the coastal nature of most studies which only cover a small portion of the Bryde's whale habitat. Consequently, if Bryde's whales are utilising the more inshore areas of Algoa Bay, for activities such as foraging, this would explain the higher sighting frequencies and SPUE.

6.4.2. Group dynamics and behaviour

The absence of Bryde's whale calves in this study is similar to other studies of these animals along the coastline (Table 5.2) (Penry, 2010). The low numbers of calves could reflect a low birth rate, or that there are key areas for mother-calf pairs which have not been identified (Penry, 2010). In general, little is known about the reproductive seasonality of this species and whether

they prefer certain areas for mating and calving (Kato and Perrin, 2009; Penry, 2010). However, calves have been observed year round in places such as the Hauraki Gulf, New Zealand and it is therefore thought to be a nursery ground for that local population (Wiseman, 2008).

Few sightings of more than one Bryde's whale were recorded in Algoa Bay (Table 5.1). This is comparable to other studies in the Hauraki Gulf, New Zealand, and in South Africa (Best *et al.*, 1984; O'Callaghan and Baker, 2002; Penry, 2010). Most of the sightings which consisted of two animals were described as foraging in bait balls, with several marine bird species, and on two occasions, with bottlenose dolphins and common dolphins. This has been recorded in other regions with up to five animals seen foraging in bait balls with other marine mammals (Zerbini *et al.*, 1997; Wiseman, 2008; Penry, 2010).

6.4.3. Description of the habitat preference within Algoa Bay

Bryde's whales were observed furthest from land (2.3 to eight kilometres) and in the deepest waters (mean 31.9 m) compared to other cetacean species in Algoa Bay (Figure 5.26 and 5.28). This is typical of this species which have also been recorded in waters greater than 50 m deep in the Hauraki Gulf (O'Callaghan and Baker, 2002; Wiseman, 2008). SST over 23 °C (p < 0.05) and depth (p < 0.05) were considered important explanatory variables when modelling the relative abundance of this species (Figure 5.40 and Table 5.5). A study conducted in Plettenberg Bay also established that SST was a significant variable in the statistical models (Penry, 2010). This was thought to be due to the relationship between SST, Chlorophyll-a concentration and seasonality, all of which affected primary short- and long-term productivity in the region and therefore resulted in an increased occurrence of prey (Penry, 2010). Groups of two animals were primarily associated with foraging in Algoa Bay, which according to Wiseman (2008) suggests that prey distribution is affected by SST and depth, and thus determines where and when foraging can occur.

Bryde's whales indicated two areas of preference within Algoa Bay, between Cape Recife and Riy Banks, and offshore of Swartkops River (Figures 5.9 and 5.22). Few studies have specifically looked at physical habitat in relation to Bryde's whale distribution. As these whales forage year-round and undergo small-scale migrations to follow prey distribution (O'Callaghan and Baker, 2002; Penry, 2010), this is likely the determining factor for the choice of habitat in Algoa Bay. According to the sediment data (supplied by SANHO), the area east of Cape Recife consists of an

underlying substrate of coral and rock (see Figure 3.2). This substrate, alongside the relatively steep bottom topography could contribute to higher prey densities in the area (Tynan, 1996; Baumgartner, 1997; Robinson *et al.*, 2009).

The area offshore of the Swartkops River consists of mud and sand sediment (Figure 3.2). Several bait balls were recorded here, including ones where Bryde's whales were foraging. This region was calculated to have the highest density of this species, and therefore should be considered an important area for these animals. It is not clear what causes shoals of fish to enter this area and further more in-depth studies are needed of the water currents in Algoa Bay, as well as the movement of sediment and nutrients from the Swartkops River, in order to establish what factors lead to the formation of these bait balls (Goschen and Schumann, 1988; Binning and Baird, 2001).

6.5. BOTTLENOSE DOLPHIN

6.5.1. General spatial and temporal distribution patterns

The bottlenose dolphins observed in this study were generally thought to be consistent with the inshore form of this species (*T. aduncus*). However, there were two distinct 'forms' observed during dedicated surveys: a smaller and darker form with a contrasting colouration along the side of the body which was observed between autumn and spring, and a larger, lighter form with a longer beak, seen throughout the year. These features would indicate that both *T. aduncus* and *T. truncatus* are present in Algoa Bay (Ross, 1977; Findlay *et al.*, 1992; Peddemors, 1999). A lack of quantitative records on which form was observed during the dedicated surveys and the absence of genetic data, make it difficult to group sightings according to these forms. Thus, for the purposes of this study, all sightings were classed as one species, *T. aduncus* (based on: Ross *et al.*, 1987; Findlay *et al.*, 1992; Peddemors, 1999; Best, 2007).

Bottlenose dolphins were the most prevalent species in Algoa Bay (Table 5.1). They were seen extensively along the inshore areas of the bay, which forms part of the south-west limit of their range around the Indian Ocean, the westward extent of which occurs around Cape Agulhas (according to: Rice, 1998; Wells and Scott, 1999; Wang and Yang, 2009). This species is thought to be part of a large 'open' population with an extensive range along the South African coastline

(Karczmarski, 1996; Reisinger and Karczmarski, 2009). Similar large ranges have been observed in other locations around the world (Ballance, 1992; Wilson *et al.*, 1997; Merriman *et al.*, 2009).

Bottlenose dolphins were sighted six times more frequently (0.63 s.hr^{-1}) than the other cetacean species, and had the highest overall relative density of 2.57 individuals.km⁻¹ (Tables 5.1 and 5.2). This compares with frequencies recorded in Marlborough Sounds, New Zealand (Merriman *et al.*, 2009). However, the dolphins in Algoa Bay had much higher sighting frequencies than what has recorded in other regions by Bristow *et al.* (2001), Bearzi *et al.* (2009) and Picanço *et al.* (2009), emphasising the prevalence of this species in the region.

6.5.2. Group dynamics and behaviour

Bottlenose dolphins live in a fission-fusion society, in which individuals associate in small groups that change frequently in composition and size (Lusseau *et al.*, 2006). In Algoa Bay, these animals were recorded to have a wide range of group sizes from one to 500 individuals, with a mean group size of 57.8 animals (Table 5.1). The wide range in group sizes was similar to other observations recorded in the Eastern Cape, where a mean group size of 140 bottlenose dolphins has been recorded (see: Saayman *et al.*, 1972; Ross, 1984). Significantly smaller groups have been observed in other parts of the world, with an average of two to 27 individuals (Weigle, 1990; Mullin *et al.*, 1994; Cañadas *et al.*, 2005; Pierpoint *et al.*, 2009).

Calves and juveniles were associated with most bottlenose dolphin sightings, and were observed throughout the year (Table 5.2). This is a common occurrence for this species worldwide, with evidence of some seasonal fluctuations in numbers (Wells and Scott, 1999; Campbell *et al.*, 2002; Bearzi *et al.*, 2009). The presence of calves throughout the year in Algoa Bay is indicative of their ability to utilise the wide range of inshore habitats available along the Eastern Cape coast (Cockcroft and Ross, 1990; Best, 2007).

Bottlenose dolphins exhibited a wide variety of behaviours in Algoa Bay (Figures 5.16 and 5.18). These behaviours did not appear to change significantly over the different seasons, with foraging and travelling predominating throughout the year, a common occurrence for this species throughout their cosmopolitan range (Ballance, 1992; Hanson and Defran, 1993; Bristow *et al.*, 2001; Lusseau and Higham, 2004).

Several observations of bottlenose dolphins included other cetacean species. They were twice observed foraging in bait balls with Bryde's whales and common dolphins, and on another two occasions they were observed foraging with humpback dolphins by Cape Recife. In addition, they were also recorded travelling with humpback whales (coming within two metres of these whales). These multi-species interactions have been recorded in other areas, and vary in their function (Saayman *et al.*, 1972; Mignucci-Giannoni, 1998; Deakos *et al.*, 2010). An example of a positive 'playful' interaction between bottlenose dolphins and humpback whales was observed in Hawaiian waters, where the whale appeared to repeatedly lift the dolphin out of the water (Deakos *et al.*, 2010).

6.5.3. Description of the habitat preference within Algoa Bay

Bottlenose dolphins were commonly observed in shallow waters of five to 11 m deep (mean of 9.2 m) and generally 300 to 800 m from land (Figures 5.26 and 5.28). This is comparable to findings in KwaZulu-Natal (KZN) where this species was located within one kilometre from land (Ross et al., 1989). However, aerial surveys conducted in Algoa Bay in the 1980s recorded the majority of sightings within six kilometres of the land, and the majority of these more 'offshore' sightings were located between PE Port and St Croix (Ross et al., 1987), which was also observed to some extent in the present study. The presence of bottlenose dolphins in Algoa Bay, has been described as part of a 'coastal corridor' of sightings up to a depth of 30 m, throughout their range (Defran and Weller, 1999; Stensland et al., 2006; Bearzi et al., 2009). Sightings located further offshore, are associated with canyons and escarpments in other studies (Bearzi et al., 2009), comparable to the observations in this study where sightings were observed in the region of Riy Banks and the two island groups (Figure 5.5). This is probably due to these features, as well as estuaries/ rivers, being optimal locations for a local abundance of prey (Hui, 1979; Ross et al., 1987; Ballance, 1992; Baumgartner et al., 2001). On the whole, depth was considered an important variable in the final hurdle model (p < 0.05; Table 5.5), which is comparable with other findings where depth has been considered significant in determining the distribution patterns of bottlenose dolphins (see: Baumgartner et al., 2001; Cañadas et al., 2005; Cañadas and Hammond, 2006).

In other studies, bottlenose dolphins have illustrated a more clumped distribution, with areas of higher densities and lower densities within their coastal corridor (Ross *et al.*, 1989; Wilson *et al.*, 1997). However, in Algoa Bay they were observed extensively throughout the coastal zone with

two 'gaps' noted in their distribution. The lack of sightings between PE Port and Swartkops River, as well as between Coega Port and Sundays River is worthy of comment (Figure 5.6). A map of the sightings recorded during an aerial survey in the 1980s showed similar observations, although, it was not explicitly commented on by the author (Ross *et al.*, 1987). It was thought that these dolphins travelled across from Cape Recife to PE Port, to the Swartkops/ Coega Rivers and then offshore to St Croix Island, with some dolphins taking more direct routes (Ross *et al.*, 1987). The direction of movement for bottlenose dolphins was not recorded in the current survey but these proposed movements towards established 'foraging areas' would explain the distribution of inshore and more 'offshore' sightings on the western half of the bay (Figure 5.5). Similar foraging areas were observed in this study (Figure 5.18).

The highest density of sightings in Algoa Bay was observed east of Sundays River, especially past 26.17 °E where the groups were dispersed (Figures 5.10 and Figures 5.23). Bottlenose dolphins are known to utilise group size and structure to achieve maximum foraging efficiency (Wells and Scott, 1999; Campbell *et al.*, 2002), and the coastal waters associated with the Alexandria Dunefield (along Woody Cape) are considered to be areas of high productivity (McLachlan, 1983; Illenberger, 1986; Webb and Wooldridge, 1990). In some surveys, dolphins were seen frequently over several kilometres, with no distinguishable beginning or end to a group. Thus, the large group sizes and high relative abundance of bottlenose dolphins observed could be due to an adequate and predictable supply of food in this region, which has been noted by Ross (1984) as a likely explanation. These sightings were most common in winter and spring, which is possibly due to the higher abundance of suitable prey during these months.

Bottlenose dolphins were also most commonly associated with sand (33%) and rock (20%) substrate, which coincides with the dominant substrates along the inshore areas (Figures 5.32 and 3.2). These species utilise the wide variety of inshore habitats to forage opportunistically on a range of prey that is available in the different habitats (Wells and Scott, 1999; Best, 2007).

6.6. HUMPBACK DOLPHIN

6.6.1 General spatial and temporal distribution patterns

Humpback dolphins were predominantly found in the inshore areas of the western half of Algoa Bay (Figure 5.5). This falls within their range which occurs along the east and south coasts of

South Africa, as well as within their entire range which extends around the Indian Ocean (as defined by: Ross, 1984; Karczmarski *et al.*, 2000b; Jefferson and Karczmarski, 2001; Kaschner *et al.*, 2006b).

No seasonal distribution patterns were obvious (Figure 5.13). The absence of sightings in September and the low number in January could be a result of the limited time period over which this study took place, and more long-term studies would confirm if these changes in sighting frequency are an annual pattern. However, the near-absence of these dolphins during autumn is noteworthy. Boat- and shore-based surveys on the western half of Algoa Bay were conducted in the early 1990s, when this species was found in the bay throughout the year, with distinct seasonal variations (Karczmarski, 1996; Karczmarski *et al.*, 1999). Both of the previous and present studies recorded a greater number of sightings during late summer (Karczmarski, 1996). Similar observations have been made along the Eastern Cape coastline (Saayman *et al.*, 1972). Another lower peak of sightings was observed in spring during the early 1990s (Karczmarski, 1996), and the current study suggests that this peak has shifted into winter.

The sighting frequency for humpback dolphins (0.1 s.hr^{-1}) was similar to Bryde's whales (Table 5.1). This is relatively low considering the coastal nature of this survey and these animals. For example, in Cleveland Bay and the Great Barrier Reef Marine Park, Australia, the sighting frequency for humpback dolphins was approximately 0.2 s.hr⁻¹ (Parra, 2006; Parra *et al.*, 2006). A higher sighting frequency of 0.34 s.hr⁻¹ was recorded in Richards Bay (Atkins *et al.*, 2004). These surveys recorded more than double the frequency of sightings compared to what was observed in Algoa Bay. Overall, this species was also observed less often than bottlenose dolphins (Table 5.1). This is in contrast to shore-based observations along the south Eastern Cape coastline in the early 1970s, where humpback dolphins were sighted more frequently (Saayman *et al.*, 1972).

These animals are thought to immigrate and emigrate into and out of the bay across the seasons, possibly even moving extensively along the coastline, depending on prey availability (Karczmarski, 1996; Karczmarski *et al.*, 1999; Keith *et al.*, 2002). However, it is difficult to determine the scale at which this occurs. Nonetheless, identified animals have been observed both in Algoa Bay and St Francis Bay, suggesting movements of greater than 100 km, and less than 1 000 km (Karczmarski, 1996). It is unlikely that distribution of humpback dolphins is continuous in the Indian Ocean; it is more expected that their distribution consists of many

subpopulations that should be treated as separate management units (Durham, 1994; Karczmarski, 1996; Keith, 1999; Karczmarski, 2000).

6.6.2. Group dynamics and behaviour

In Algoa Bay group sizes of humpback dolphins were much smaller compared to the other two delphinid species (Table 5.1). The mean group size of 3.3 individuals was similar to that observed in Hong Kong waters (Jefferson, 2000), half the size of what had been observed in the bay previously (Karczmarski, 1996), and smaller compared to other locations along the southern African coastline (Durham, 1994; Guissamulo and Cockcroft, 2004). Most of the groups observed in the recent survey consisted of less than four individuals, with three larger groups of seven, eight and 23 individuals observed once respectively. This is in contrast to previous observations in the bay which estimated a mean group size of seven individuals (Saayman and Tayler, 1979; Findlay *et al.*, 1992; Karczmarski, 1996; Karczmarski *et al.*, 1999), and significantly different to observations on this species in Maputo Bay where a mean group size of 14.2 individuals was recorded (Guissamulo and Cockcroft, 2004).

Only one calf or juvenile was observed in the majority of sightings, with one exception of a group of 23 humpback dolphins consisting of three calves and four juveniles in June 2010 (Table 5.2). Relatively few calf sightings were observed in summer, which was previously associated with the peak calving season (Karczmarski, 1996). Overall, the lower number of calves present in this study could indicate that the birth rate has decreased.

Humpback dolphins were primarily observed foraging (33%), with mating and socialising seen in summer and autumn respectively (Figures 5.15 and 5.16). These findings are similar to that of Karczmarski (1996) and Parra (2006). However, foraging also occurred south of Swartkops River, and east of Coega Port (Figure 5.18). This behaviour predominantly occurred along the rocky shores on the south-west corner of the bay, which is in accordance with previous sightings where humpback dolphins have frequently been observed foraging along rocky shores in the Eastern Cape, and in large estuarine systems in KZN (Saayman *et al.*, 1972; Durham, 1994; Karczmarski, 1996; Karczmarski *et al.*, 2000a; Atkins *et al.*, 2004).

Two humpback dolphin sightings were associated with bottlenose dolphins, consisting of three and two individuals, respectively. In the first sighting, both species were observed foraging, with one instance of a bottlenose dolphin tail-slapping a humpback dolphin. Non-aggressive interactions between these two species have been occasionally observed in the bay (Saayman *et al.*, 1972; Karczmarski *et al.*, 1997) and aggressive interactions between these two species in the region have also been reported (Saayman *et al.*, 1972; Saayman and Tayler, 1979). An incidental observation of humpback dolphins interacting with southern right whales in August 2009 was also observed, both species were recorded as socialising. This has been observed previously in the region by Saayman and Tayler (1979).

6.6.3. Description of the habitat preference within Algoa Bay

Humpback dolphins were generally observed within 500 m of land (median of 300 m), at a mean depth of 6.6 m, and within ten kilometres of the nearest river mouth (Figures 5.26, 5.28 and 5.31). This is very similar to the findings from the early 1990s, where 87% of sightings were observed within 400 m of land, and almost all the sightings were in waters less than 15 m deep (Karczmarski, 1996; Karczmarski et al., 2000a). Other studies have also established that humpback dolphins commonly occur within one kilometre of the coast in waters less than 50 m deep, and generally within the surf zone (Ross et al., 1994; Atkins et al., 2004; Parra, 2006). In Richards Bay, humpback dolphins were seen at a mean depth of 13.2 m, with a maximum depth of 31.8 m (almost five kilometres from land) (Keith, 1999), which was much deeper and further offshore than what was observed in this study. This is most probably due to the wide, shallow Tugela bank which results in the 15 m isobath being positioned further offshore (Durham, 1994). In northern Queensland, Australia, humpback dolphins were also found further offshore (mostly less than five kilometres from land), in waters less than 15 m deep (Parra et al., 2006). The inclusion of depth as a significant explanatory variable in both stages of the final hurdle model (p < 0.05) further confirms the influence this variable has in determining the limit of humpback dolphin distribution throughout their range (Table 5.5).

The highest densities of sightings were found along the inshore areas between PE Port and Cape Recife (Figures 5.11 and 5.24). This is the same area identified previously as a key habitat for humpback dolphins in Algoa Bay (Karczmarski, 1996; Karczmarski *et al.*, 1998). Whereas previously, almost all of the sightings were found south of PE Port (Karczmarski, 1996), a number of sightings in this study were found between the two ports. An incidental sighting was even observed in the Swartkops Estuary. Sightings have also been recorded in other estuaries in Mozambique and Australia (Parra, 2006; Parra and Ross, 2009).

Humpback dolphins were to some extent more closely associated with estuaries and rivers in the bay compared to other cetaceans (Figure 5.31). Parra *et al.* (2006) also established that humpback dolphins were found within 20 km of the nearest river. In Richards Bay, a high proportion of sightings and foraging occurred around the harbour, a dredged estuary (Atkins *et al.*, 2004). Thus, along with the rocky shores, estuaries could provide an important foraging ground for this species (Ross *et al.*, 1994; Atkins *et al.*, 2004). This would account for the incidental sightings of these animals in the Swartkops Estuary.

Unlike the other cetacean species observed in Algoa Bay, a significant proportion (57%) of humpback dolphin sightings were associated with a rocky substrate found along the south-west corner of the bay (Figures 5.32 and 3.2). This is in accordance with other records of humpback dolphins (Saayman *et al.*, 1972; Ross *et al.*, 1994; Karczmarski *et al.*, 2000a). Overall, this area of the bay also had the highest incidence of foraging behaviour (Figure 5.18). The high density of sightings and foraging observations occurring within this relatively small area indicate that this region is a key habitat for this species (Figure 5.18 and 5.24).

6.7. COMMON DOLPHIN

6.7.1. General spatial and temporal distribution patterns

Common dolphins were the least frequently observed cetacean species in Algoa Bay, and due to the paucity of sightings, it was difficult to determine whether there are any patterns in their spatial and temporal distribution. In both dedicated boat-based surveys and opportunistic sightings, common dolphins were primarily observed in the offshore areas (Figure 5.5). Although, several sightings were observed more 'inshore' around the Swartkops River and around Coega Port. This was in a similar location to where Bryde's whales were observed.

Common dolphins are generally considered to be 'offshore' species and enter the bay opportunistically in search of shoals of fish (Reilly, 1990; Samaai *et al.*, 2005; Best, 2007). Thus, the sightings recorded in Algoa Bay only represented a small sample at the edge of their extensive range. These dolphins were found in the bay in late summer, winter and in spring.

Common dolphins have long been known as the principal predators in the annual Sardine Run (Cockcroft and Peddemors, 1990; O'Donoghue *et al.*, 2010b; van der Lingen *et al.*, 2010). The

peak in sightings in late summer is expected as common dolphins are thought to inhabit the Eastern Cape coastline during this season, moving to the Natal coastline during winter (Ross, 1984; Cockcroft and Peddemors, 1990). Although, aerial surveys carried out between Port Elizabeth and East London in the late 1980s detected common dolphins in low densities throughout the year (Cockcroft and Peddemors, 1990). Thus, the presence of common dolphins in the bay at the beginning of winter (Figure 5.13) is probably closely linked to the distribution and movement of their prey along the Eastern Cape coastline (Cockcroft and Peddemors, 1990). The sightings in both May and June in 2010 were most likely a result of the dolphins moving into inshore waters at the start of the Sardine Run to feed off small and localised bait balls that had formed in the bay, before following the migration of the sardines up the east coast. However, this movement of common dolphins towards KZN does not necessarily involve the entire population (Cockcroft and Peddemors, 1990), which would account for the presence of common dolphins in Algoa Bay during spring 2009.

6.7.2. Group dynamics and behaviour

Common dolphins were observed in large groups of up to 800 animals, with a mean of 343 individuals (Table 5.2). Two smaller groups were observed travelling in shallow areas. This species is known to have extremely large group sizes in comparison to their other delphinid counterparts (Saayman *et al.*, 1972; Barlow, 1995; Weir *et al.*, 2009). These large groups were also consistent with other findings in South Africa, which had mean group sizes of 302 to 619, with one group of approximately 10 000 individuals observed in an aerial survey between Port Elizabeth and Richards Bay (Cockcroft and Peddemors, 1990; Findlay *et al.*, 1992; Best, 2007; O'Donoghue *et al.*, 2010b). The group sizes appear to be slightly larger than other locations such as the Minch in Scotland where common dolphins had a mean group size of 135 (Weir *et al.*, 2009). Common dolphin groups in south Australia were also observed in small groups of approximately five animals (Filby *et al.*, 2010).

Common dolphins were either seen foraging, or travelling (porpoising) to and from foraging areas in Algoa Bay (Figure 5.15 and 5.16). While following the group, they were frequently observed bow-riding, a common behaviour with this species (Best, 2007). However, no other socialising or 'play' behaviours were observed. Foraging activities were observed in 'bait balls', with a number of other marine predators such as Cape gannets, Bryde's whales and bottlenose

dolphins. This is a frequent observation for common dolphins in South Africa (Saayman *et al.*, 1972; Best *et al.*, 1984; O'Donoghue *et al.*, 2010b).

Two sightings of single animals were seen within a group of bottlenose dolphins. However, there were no photos taken of the individuals, which would have assisted in determining whether the animal was a common dolphin or a hybrid (see: Reyes, 1996; Zornetzer and Duffield, 2003). Common dolphins have been associated with other cetacean species such as striped dolphins (*Stenella coeruleoalba*) and Risso's dolphins (*Grampus griseus*) (Cañadas *et al.*, 2002; Frantzis and Herzing, 2002; Cañadas and Hammond, 2008).

6.7.3. Description of the habitat preference within Algoa Bay

Common dolphins were sighted at depths of 16 to 38 m and 4.5 to six kilometres from land, which was similar to the humpback whales and Bryde's whales (Figure 5.26 and 5.28). Other studies have also observed common dolphins at similar depth/ distances from land (Filby *et al.*, 2010; O'Donoghue *et al.*, 2010b), but there are many areas where a number of sightings have been observed on the continental slope (100 to 200 m deep) (Selzer and Payne, 1988; Findlay *et al.*, 1992; Cañadas and Hammond, 2008). The small number of sightings in Algoa Bay suggests that common dolphins in this region prefer more offshore habitats of less than 500 m deep (Findlay *et al.*, 1992; Peddemors, 1999).

6.8. ANTHROPOGENIC INFLUENCES AND IMPACTS ON CETACEANS IN ALGOA BAY

Based on the knowledge gained from this survey on the spatial use of the bay and cetacean behaviour, threats and issues identified in the literature can now be considered for Algoa Bay (objective four). The predominant threat to mysticetes in the past was whaling, which began in 1775, and almost exhausted several large mysticete stocks, including the species observed in this study (International Whaling Commission, 1986; Best, 2007). Now, several organisations have been established to protect these species from commercial whaling (Friedmann and Daly, 2004). However, there are a number of current and potential future threats that could influence the full recovery of these populations. These were discussed in detail in Chapter Two.

Understanding the spatial and temporal distribution patterns as well as identifying key habitats of a cetacean species is imperative if the favoured environment for these animals is going to be correctly managed in the future (Wilson *et al.*, 1997; Hastie *et al.*, 2003; Cañadas and Hammond, 2008). Coastal dolphins are some of the most threatened cetacean species due to their close proximity to anthropogenic activities (Thompson *et al.*, 2000). With the expanding use of the coastal zone for recreational and commercial activities, it is necessary to monitor these impacts.

With the increase in shipping activities in the bay (associated with the new Coega Port), it is important to monitor the associated impacts in order to mitigate any potential negative affects in the future. A brief summary of the potential threats to cetaceans, in the context of the results of this study, are outlined in Table 6.1 below.

6.9. DEFINING POTENTIAL KEY HABITATS FOR THE CETACEANS IN ALGOA BAY

Several potential key habitats were identified for mysticetes and odontocetes as illustrated in Figures 5.45 and 5.46. Possibly the hardest area to monitor is Riy Banks (A) where humpback whales are prominent. This reef is more difficult to manage as it is located more than 20 nm offshore of PE Port, and it is frequently utilised by commercial fisheries. The dynamics of the fishing activities in this area are not well known, and combined with limited data on the movement of animals in this area, make it difficult to implement appropriate conservation measures.

The south-west corner of the bay is a key habitat for both mysticetes and odontocetes. Southern right whales were seen extensively around PE Port (B), where a large proportion of mating was observed (Figure 5.45). Another key habitat lies south of PE Port (H) for both bottlenose dolphins and humpback dolphins (Figure 5.46). This area corresponds with the proposed humpback dolphin marine sanctuary (Klages, 2006). The sanctuary is set to exclude inshore motorised vessel activities, with the exception of two launch corridors within this exclusion zone (Klages, 2006). The sanctuary will create an 800 m wide buffer for the dolphins, which roughly corresponds with the ten metre isobath. Although the key habitat (H) almost extends to the 20 m isobath, the proposed sanctuary is a realistic and achievable plan which takes into account both development and conservation measures, and should therefore be implemented.

Threats	Species	Observations in the field	Supporting observations in the literature	Level of concern for Algoa Bay				
Ship strikes (collisions with vessels) and various shipping/ boating activities	Southern right whales	Slow moving animals which do not show strong boat avoidance behaviour. A moderate proportion of sightings were recorded in shipping areas.	Calves are particularly vulnerable to ship strikes (Klinowska, 1991; Best <i>et al.</i> , 2001a).	A healthy population growth rate suggests that this would not have a significant impact at a population level (Clapham <i>et al.</i> , 1999; Best <i>et al.</i> , 2001a).				
	Humpback whales	Small proportion of sightings within the main shipping areas. Scars and fresh wounds from propellers observed.		Probably not significant in Algoa Bay due to small percentage of sightings observed within key shipping areas and an overall healthy population size.				
	Bryde's whales	High proportion of sightings within shipping areas.	Population is considered 'data deficient' or 'vulnerable' (Friedmann and Daly, 2004; Reilly <i>et al.</i> , 2008a).	Potential threats to Bryde's whales have not been well assessed, therefore it is difficult to determine whether this activity could potentially significantly affect this species (Best, 2007; Penry, 2010).				
	All odontocetes	Small proportion of sightings within the main shipping areas. Scars and fresh wounds from propellers observed (including cut-off dorsal fins).	Humpback dolphins are classified as 'vulnerable' or 'near- threatened' (Friedmann and Daly, 2004; Reeves <i>et al.</i> , 2008)	Odontocetes are highly mobile, and thus ship strikes are not considered to be a major threat. However, it could impact the small humpback dolphin population.				
Entanglement in stationary/ floating objects (e.g. anchor lines)	Southern right whales, humpback whales, bottlenose dolphins and humpback dolphins	Scars and fresh wounds from fishing lines on many animals.	This threat is associated with various fishing and shipping activities that are known to occur throughout the bay (Rice, 1998; Best <i>et al.</i> , 2001a; Friedmann and Daly, 2004).	Scars and fresh wounds suggest that this threat is already having an impact on the animals. However, the extent of the impact is unknown.				

Table 6.1. Outline of the anthropogenic threats facing cetaceans in Algoa Bay.

Table 6.1. Continued

Whale-	Southern right	Changes in behaviour have	Less mobile animals are	Changes in behaviour have been
watching/	whales, humpback	been observed when vessels	particularly vulnerable (such as	observed, indicating a potential
harassment	whales, bottlenose	(including fishing, pilot and	mother-calf pairs). Noise	impact, especially to certain
by vessels	dolphins and	research vessels) approached	associated with vessel	individuals in a population (e.g.
	humpback	the animals.	disturbance is known to result in	mother-calf pairs).
	dolphins		short-term behavioural changes	
			however, little is known about	
			the long-term negative effects of	
			this activity (Constantine et al.,	
			2004; Corkeron, 2004;	
			Friedmann and Daly, 2004).	
Pollution and	Humpback	Not directly assessed or	These animals have a relatively	Possibly very important due to the
the	dolphins	observed during this study	limited distribution (along coastal	small population size in the region.
destruction/			areas) with distinct key areas	
alteration of			where foraging occurs	
coastal zone			(Karczmarski, 1996).	
	Bottlenose	Not directly assessed or		Less of a threat to this species due to
	dolphins	observed during this study		their larger home ranges and use of a
	I I	8		wide variety of habitats within their
				range.
				6

Although prominent behavioural activities and densities are relatively low between Coega Port and PE Port there are a number of important findings related to this area. This includes the foraging activities by humpback and bottlenose dolphins around Swartkops River (especially north of the river) (I), and an offshore area (C) where a number of Bryde's whales were observed foraging in bait balls. The Swartkops River is known to be highly polluted, with elevated levels of heavy metals which are deposited in the sediments on the surrounding sea-floor (Melville-Smith and Baird, 1980; Binning and Baird, 2001). The coastal zone between these two ports is also highly modified with a large amount of industrial activity taking place, and the presence of anti-erosion dolosses. A search through the literature revealed no studies which have determined the impact pollution is having on the surrounding marine fauna (such as a decrease in prey density). Thus, it is difficult to determine whether these anthropogenic activities have already resulted in cetaceans avoiding this area or whether this area has always been less occupied.

The outline for the proposed GAENP MPA potentially protects several of the key habitats that have been described in this study. This incorporates areas where a large proportion of mother-calf pairs were observed for southern right and humpback whales, as well as opportunistic foraging by humpback whales, delineated by areas D to G and J to L in Figures 5.45 and 5.46. Bottlenose dolphins were recorded in very high numbers in the proposed MPA where a large proportion of the foraging behaviour was exhibited by this species (Figures 5.23 and 5.18). Since the exact boundary of the MPA has not been confirmed, it is suggested that the MPA should *at least* extend from Sundays River to Cape Padrone up to the 30 m isobath. This would incorporate most of the southern right whales and bottlenose dolphins, as well as a large proportion of the humpback whales.

The two island groups are already exclusion zones for fishing activities, which protects an important foraging area for the bottlenose dolphin (area J; Figure 5.45). If the MPA includes both island groups, it will incorporate a relatively important area (D) for the mother-calf pairs, which lies between St Croix Island and the coastline.

Therefore, the proposed protected areas for Algoa Bay could play an important role in conserving cetaceans in Algoa Bay, with a large proportion of the key habitats, falling within these protected areas. However, the area offshore of Swartkops River and Riy Banks both should be examined in more detail, in order to determine the degree of importance of these habitats.

6.10. IMPLICATIONS FOR CONSERVATION AND MANAGEMENT

This study provides an important foundation for understanding the spatial and temporal distribution patterns of the cetacean species utilising Algoa Bay. In order to study cetacean distribution and habitat preferences, knowledge on the location, relative abundance/ frequency and behaviour of a species is required. Analytical tools such as GIS and statistical analysis which were used in this study, have aided in defining and quantifying these distribution patterns. To date, little is known about the role cetaceans play in the marine ecosystems around South Africa, and whether the populations and critical habitats of these species are adequately protected under the various national and international legislation (Harwood, 2001; Cañadas *et al.*, 2005). This type of baseline study would provide input to understanding this broader context.

Distribution studies at a variety of spatial scales are useful tools in monitoring the distribution of animals and identifying critical habitats (Hastie *et al.*, 2003). Once these important areas have been established, further development can take into account these areas, and where possible, work around them to establish corridors where key species, such as cetaceans, are able to move to their important habitats to perform essential activities for survival (Hastie *et al.*, 2003). For example, Karczmarski *et al.* (1998) suggested establishing the humpback dolphin marine sanctuary in Algoa Bay. A conservation priority area was marked on the south-west corner of the bay, extending from Humewood Beach to Cape Recife (Karczmarski *et al.*, 1998). Although the full extent of the this sanctuary is unlikely to be implemented, a significant portion of this zone is under review (Klages, 2006). However, if the proposed King's beach development is implemented, this could result in this key habitat for three cetacean species, being impacted. Thus, this study has reiterated the necessity for this sanctuary, as the results of the present study and recent literature indicate that the humpback dolphin population may be under more stress than it was in the early 1990s.

Another example of managing critical habitats would be the establishment of MPAs, which are located at various places along the coastline of South Africa. These are fundamental in the conservation of cetaceans, but research must be done to determine the role these areas play in actually protecting the species concerned as well as the other biota in the marine environment (Rice, 1998; MacLeod, 2009). According to the data collected during this survey, the proposed GAENP MPA in Algoa Bay has the potential to be an important area for a wide variety of species, including southern right whales, humpback whales and bottlenose dolphins. A large

proportion of the most vulnerable individuals (mysticete mother-calf pairs) sighted in the bay were located within the proposed MPA during the study. Consequently, this area would protect these individuals from several threats outlined in section 6.8.

6.11. CONCLUSION

Species	Key findings										
Southern right whale	Mating activities focused around PE Port during spring										
	Potentially a new calving/nursery area as a significant proportion of sightings were of mother-calf pairs, especially within the proposed GAENP MPA										
	Located in shallow (bottom depths of ten to 15 m) and sandy areas										
Humpback whale	High number of sightings										
	Strong seasonality linked to migration patterns along the continental coastline during austral winter										
	High proportion of mother-calf pairs (and therefore potentially an important nursery area)										
	Associated with reefs and island habitats										
	Opportunistic foraging observed along the Alexandria Dunefield										
Bryde's whale	Elusive, offshore species										
	Predominantly sighted offshore of Swartkops River during summer and autumn (foraging in bait balls)										
Bottlenose dolphin	Widespread throughout shallow waters in bay, throughout the year										
	Exhibited a wide-variety of behaviours and group dynamics (sizes)										
	Higher densities located along Alexandria Dunefield										
Humpback dolphin	Inshore species (mean bottom depth of 6.6 m)										
	Limited distribution with most sightings located in the south-west corner of the bay										
	Associated with rocky shores which is probably linked to prey distribution patterns										
	A noteworthy decline in sightings and group size since 1990s										
Common dolphin	Only offshore odontocete species										
	Extremely large group sizes (up to 800 individuals)										
	Distribution strongly associated with prey in Algoa Bay (primarily observed foraging)										

Table 6.2. A summary of the key findings associated with each of the cetacean species.

The key findings of this research are displayed in a simplified map (Figure 6.5) and in a summary table (Table 6.2). Cetaceans in Algoa Bay illustrated distinct and varying patterns in their spatial and temporal distributions, with a number of key habitats identified (objective three). The location of these key habitats was influenced by a number of geographical and biological variables such as relative density, behaviour, the presence of mother-calf pairs, bottom depth, distance from land, sea-floor substrate, SST and seasonality (objective four) (Table 6.2). The aforementioned variables, along with anthropogenic activities and proposed conservation strategies, play an important role in understanding the present distribution patterns of these cetaceans (objectives four and five). Based on the results of this study, some of the threats facing these animals have been outlined. This forms a framework on which the relative importance for each of the key habitats can be defined. This was done to provide recommendations for future research, management, and conservation strategies, which outlined in the following chapter.



Figure 6.5. Summary map of the key habitats for the different cetacean species in Algoa Bay. Red areas are important coastal key habitats, with the yellow areas representing key habitats that are located offshore (and which need to be researched further). A star indicates that the area is potentially the most important key habitat for the species in Algoa Bay.

CHAPTER 7: CONCLUSIONS AND RECOMMENDATIONS

7.1. INTRODUCTION

This final chapter will summarise the key findings of this research and provide conclusions, in relation to the aim and objectives outlined in Chapter One. The caveats of this research are addressed initially, followed by a summary of the most significant findings per species, leading into recommendations for management, conservation and research.

7.2. POTENTIAL CAVEATS

There are several problems that could affect the quantity and quality of the data captured and analysed in cetacean research, including this study. The short time-span of this research made it difficult to extrapolate long-term spatial and temporal distribution patterns and the habitat preference of these long-lived, wide-ranging animals, as these patterns may have been caused by spurious intra-annual patterns that do not represent their true biogeography. Thus, all results and discussion should be treated as preliminary indications of these trends. However, this study has established potential areas of interest (key habitats) that should be further researched and taken into consideration for future management and planning.

Logistical and financial constraints are the most prominent limiting factors affecting cetacean studies, and this study was no exception. Poor weather conditions during this study resulted in irregular sampling during certain times of the year, where all tracks were not covered. This particularly affected surveys in the more exposed areas of Algoa Bay (Riy Banks and Bird Islands). This was compounded by boat availability being, for the most part, restricted to ten days per month. Thus, some boat-based surveys were partially carried out in less then ideal circumstances (Beaufort sea state three and above), which may have affected search effort, with swell and white caps potentially reducing the ability to sight animals.

Opportunistic sightings (land- and boat-based) are useful in providing attribute information to supplement the primary dataset. However, the same level of detail is not obtained and may not be as accurate as survey data (for example, GPS position, group size and dynamics). Therefore, the opportunistic data collected during the study period were not utilised to the same extent as dedicated surveys. Nevertheless, opportunistic sightings in this study did reinforce trends

observed in the boat-based surveys, and assisted in filling 'data gaps' where a high intensity search effort could not be carried out.

This study did not take into account several geographical, biological and anthropogenic variables that could have potentially improved the understanding of the dynamics of the cetaceans in the bay. A suitable resolution of Chlorophyll-a data, which is an important indicator for primary productivity, was difficult to obtain and could not be processed due to logistical limitations. Data on prey distribution was also not available for Algoa Bay, which could provide valuable information on one of the primary factors affecting cetacean distribution.

Pollution, noise, and interactions with vessels (recreational, fishing etc.) are some of the prominent anthropogenic influences that could be affecting the cetaceans in the bay. The inclusion of these variables, as well as other variables measured in this study, could have potentially resulted in the construction of more robust statistical models, which would improve the explanation of the observed distribution patterns. However, parsimonious statistical models are much easier to interpret ecologically and can provide an indication of what analysis should be done on a larger and more long-term dataset in the future.

There are a number of challenges faced when dealing with spatial datasets. There are limitations within the programs used (such as ArcGIS), which then need to be supplemented by other programs (for example, Hawth's tools and R) that use a different interface. This increases the time taken to perform tasks and often leads to overlay problems with the different sources of data. In this study, the lack of remote sensing data and complex analysis of raster data combined with vector data was a result of these limitations in the data and the programmes. However, the increasing compatibility of different programs (that handle spatial data), significantly improves the range of analytical techniques that can be conducted on a wide range of data.

7.3. CONCLUSIONS

The aim of this research was to determine the spatial and temporal distribution and habitat preference, of the cetaceans in Algoa Bay. This included areas within and outside the proposed GAENP MPA. In addition, this study aimed to relate this distribution to various geographical parameters. Five objectives were laid out in Chapter One, in order to achieve this aim. A

summary of the key findings and conclusions in relation to each objective and research question is outlined below.

The first objective was to collect primary data on cetacean location, behaviour and other related environmental variables via different survey and opportunistic techniques. The principal method for obtaining this data came from dedicated boat-based surveys which were carried out monthly in Algoa Bay, between March 2009 and July 2010 (Chapter Four). As no boat-based surveys had addressed all cetacean species in the bay previously, the first research question was to determine which cetacean species were observed in Algoa Bay. In both the dedicated boat-based surveys and opportunistic data collection, six cetacean species were observed: southern right whales, humpback whales, Bryde's whales, bottlenose dolphins, humpback dolphins and common dolphins. These results were described in Chapter Five.

The second objective was to acquire spatial and non-spatial data of geographic and anthropogenic variables in order to determine whether these variables could be used to explain cetacean distribution. Geographical data such as Beaufort sea state, SST, and bottom depth were collected throughout the dedicated surveys, both during and in-between sightings (while 'on effort'). Secondary data came primarily from reviewing the literature (Chapter Two) as well as other marine and coastal studies in the region (Chapter Three). Secondary data were collected on the bathymetry of the bay, sea-floor substrate, shipping areas and MPAs.

The third and fourth objectives were to determine the spatial and temporal distribution and behaviour of the different cetacean species, and to relate these patterns to the geographic and anthropogenic variables delineated in objective two. The results of this analysis show that, the six cetacean species illustrated distinct differences in their spatial and temporal distribution patterns. These patterns were displayed and discussed in detail in Chapters Five and Six. The key findings for each species are represented spatially on a map (Figure 7.1) and can be summarised as follows:

Southern right whales were primarily observed in inshore waters (bottom depths of ten to 15 m) of the western half of the bay, during austral winter and spring, especially around PE Port. These sheltered and sandy areas were used for mating and calving (both in and outside of the proposed GAENP MPA). However, these two areas (for mating and calving) did not appear to be as segregated, as suggested in other studies. Mating activities were primarily observed in spring,

around PE Port, and coincided with the peak in sightings for this species. The presence of calves in a large proportion of sightings, most of which occurred within the proposed MPA, indicates that there has been a notable change in the distribution patterns of this species along the South African coastline.

Humpback whales were seen extensively in Algoa Bay between June and December. Two peaks in sightings were noted, a winter peak in June associated with their northward migration of this species, and a peak at the end of their wintering season, in December, associated with the southward movement of these animals. These animals occupied a wide range of habitats, with high relative densities associated with the coastal waters off the Alexandria Dunefield, and the islands and reefs in the bay, especially Riy Banks. These whales potentially use the 'Recife Bird Ridge', in order to navigate along the coastline. These animals exhibited a wide range of behaviours, including breaching and flipper slapping, with opportunistic foraging observed three times during November, within the proposed GAENP MPA.

Bryde's whales were the most elusive species observed. They were predominantly recorded in summer and autumn in the western half of the bay. The lack of sightings during winter and spring could be due to seasonal migrations along the South African coast, associated with the movement of their prey in a natural phenomenon known as the Sardine Run. A higher relative density of Bryde's whales was observed offshore of the Swartkops River, where a number of foraging activities were recorded.

Bottlenose dolphins were the most prolific species in Algoa Bay. They were observed year round in the inshore waters, with a peak in sightings during winter and spring. This species exhibited a wide variety of behaviours and were associated with most of the geographical features in the bay. These animals also had large and highly variable group sizes. However, higher densities were predominantly observed along the Alexandria Dunefield (in the proposed MPA). These dolphins were also observed interacting with other cetacean species. This included foraging in bait balls with Bryde's whales and common dolphins, as well as foraging with humpback dolphins along the south-west corner of the bay. Social interactions between bottlenose dolphins and humpback whales and humpback dolphins, were also recorded.



Figure 7.1. Annotated map of the potential key habitats for cetacean species. Two offshore key habitats (in yellow) need further research in order to determine the extent of the importance of this area.

Humpback dolphins were sighted in very shallow waters, with most sightings occurring within 500 m of land, with a mean bottom depth of 6.6 m. Despite their inshore distribution, these animals were not observed as frequently as some of the other cetacean species, with the lowest number of sightings observed during autumn. Group sizes were much smaller than what was recorded in a study in the early 1990s (Karczmarski, 1996). However, like the previous study, these dolphins are still strongly associated with the rocky shores along the south-west corner of Algoa Bay.

Common dolphins were the only offshore odontocete species sighted during this study. They were observed in extremely large groups of up to 800 individuals which were generally associated with bait balls, along with other marine fauna. Due to the paucity of sightings (and low number of offshore surveys), it was difficult to determine whether these animals utilise the bay opportunistically, or on a seasonal basis (in search of prey), where their occurrence would be part of an established pattern of movement along the South African coastline. However, it is known that these animals are the principle predators in the annual Sardine Run. Thus, these animals follow the Sardine Run, which is thought to begin east of Algoa Bay during winter, before moving up the coast towards KZN.

The final objective is to establish some recommendations and guidelines. By way of conclusion, through consideration of the information collected on cetacean behaviour, together with identified threats (Chapter Six) and the knowledge of the spatial habitat preferences of these animals, some recommendations are put forward in the following section.

7.4. RECOMMENDATIONS FOR MANGEMENT, CONSERVATION AND RESEARCH

Based on the results of this research, cetacean distribution in Algoa Bay was associated with several environmental parameters, such as sea-floor substrate, bottom depth, SST and other geographical variables. With reference to these findings, the information derived from this study could be considered in future management strategies. The potential management areas where this information might be used are outlined below. These ideas should be considered in conjunction with the annotated map (Figure 7.1) which represents the key cetacean habitats.

1. With regard to shipping lanes and anchoring, in the event that these areas are changed, critical areas for cetaceans can be taken into consideration and avoided. Consequently, the integrity of the associated habitats in these critical areas would be maintained. This is especially relevant as a potential key habitat was identified in a shipping/ anchoring area for Bryde's whales (Figure 7.1).

2. With regard to urban planning, more informed decisions can be made about where to place new infrastructure such as waste pipelines, as they can alter sea-floor characteristics (habitat loss) and could change prey distribution patterns through this loss or alteration in habitat. This would potentially affect the distribution of odontocetes, which is linked to their prey distribution (Bristow *et al.*, 2001). Calving areas for all the cetacean species could be affected/ changed, as the distribution of these calving areas correspond with the optimal habitat conditions associated with this activity, and thus they would be affected by a disruption in the habitat (Bristow *et al.*, 2001). In terms of Algoa Bay, a number of anthropogenic activities are situated in the south-west corner, coinciding with a potential key habitat for three cetacean species (Figure 7.1).

3. With regard to MPA planning, the present and proposed MPAs need to be re-evaluated with a long-term dataset, in order to determine whether the proposed boundaries adequately protect the critical habitats or if they need to have more fluid boundaries that shift across time (Hastie *et al.*, 2003), i.e. that the summer and winter boundaries are slightly different to account for seasonal distribution shifts in a species. This study has illustrated that St Croix Island and the proposed GAENP MPA are both considered potential key habitats for three cetacean species that utilise Algoa Bay, and therefore the proposed MPA should be implemented (Figure 7.1). Further research should also be conducted on the marine fauna that utilise Riy Banks, as this reef is potentially an important habitat for cetaceans and other animals, and should therefore be carefully monitored (Figure 7.1).

4. With regard to legislation, Karczmarski *et al.* (1998) identified inshore powerboat traffic (especially jet skis) as a large disturbance to humpback dolphins. Current legislation outlaws any disturbance to cetaceans (Government Gazette, 2008), however, this is not effectively implemented. Boat traffic should be prohibited in important inshore areas, limiting the negative impact on these dolphins. The establishment of the proposed humpback dolphin marine sanctuary has the potential to help manage these disturbances for a significant proportion of the humpback dolphin habitat situated in the south-west corner of Algoa Bay (Figure 7.1). However, guidelines on approaching these animals and the time spent around a group of animals, need to be carefully

outlined and closely monitored to avoid excessive disturbance of these cetaceans, which could possibly have short- and long-term effects (Richardson *et al.*, 1995; Karczmarski *et al.*, 1998).

5. With regard to future research, Hastie *et al.* (2003) illustrate the need for a multi-scale approach in identifying geographic, biological and anthropogenic factors that determine the distribution patterns of a particular cetacean population. Although this study only provides baseline data on broad-scale patterns on the cetacean species, this is required to direct further studies in asking the 'right' questions, to better conserve and manage these cetacean populations and the marine ecosystem as a whole.

This study falls under the auspices of a three year research project on the cetaceans in Algoa Bay, after which further long-term monitoring needs to be established. This will ensure that the trends observed are characteristic of the species, allowing for more refined/ established key habitats that can be monitored and conserved effectively. This will also aid in determining the effects of large-scale influences, such as climate change, on cetacean distribution (Simmonds and Eliott, 2009).

The shipping zones were the main anthropogenic activity observed in this study. However, there are many other anthropogenic influences that indirectly or directly influence cetacean distribution and their habitat preference, for example, chemical and noise pollution, and the effects of the presence of recreational and commercial (fishing) vessels, on cetacean behaviour. Thus, the extent of these threats needs to be established in further research, to determine how these factors could be controlled.

Future research should also include genetic sampling, prey analyses, larger-scale surveys (covering the entire Algoa Bay up to the continental shelf) and behavioural studies. Genetic sampling, combined with photo-identification work, would aid in determining the population dynamics and further the understanding of the spatial and temporal distribution patterns. This would be especially interesting for the bottlenose dolphins observed in this study where two distinct forms were observed, possibly indicative of two species. Some preliminarily photo-identification analysis has been done on the humpback dolphins in Algoa Bay by the author, and this should be continued, along with more detailed behavioural studies. This would determine the significance of the key habitat (in the south-west corner of the bay) for this species, and mitigate any potential impacts which could negatively affect this population in the future.

A wide range of analytical techniques have been applied in this study. The integration of the variables used along with remote sensing images, and further spatial analysis and statistics, on a more long-term dataset, will refine these trends and allow for better inference from the results. This would form a foundation from which explanatory and predictive statistical models, of the spatial distribution of the cetacean species, could be constructed.

In conclusion, this dissertation has made a valuable contribution to current knowledge of cetaceans in Algoa Bay. This study has provided insight on local-scale cetacean distribution, behaviour and habitat use, making an important contribution to research on cetaceans in the coastal waters of South Africa. The findings of this research demonstrate the value of a site-specific study, which takes into account a multitude of species, and identifies key habitats. Additionally, it forms a foundation from which future research and management decisions, which are beneficial and sustainable in terms of conservation strategies and anthropogenic activities, can be made.

APPENDIX 1: EFFORT FORM

Date:	Observers:	Sightability (1-5)	Surve	ey #
Vessel: Honckenii				
Launch site: <i>PEDSAC</i>				

Weather check: Beaufort scale/ sea state (B1-4), wind direction, cloud cover (x/8), water temp ($^{\circ}C$) & water depth (m), mist/haze? SE = search effort. Course change (cc): destination, speed and bearing/direction.

Tiı	me													B	Wind	/8	°C	m	Comments		
																					Engines on
									S						E						
									s						Е						
									S						Е						
									s						Е						
									S						Е						
									S						Е						
									s						E						
									S						Е						
									s						E						
									S						Е						
									S						Е						
									S						E						

APPENDIX 2: SIGHTING RECORD

SIGHTING RH SHEET - ALG BAY	ECO OA	RD	F	orm	n #		Da	te (da	iy, mo	onth,	year)		General Location Recorder Sighting #
Time seen	Ti	ime	clos	ed	Ι	nitia	al G	PS re	ading	S			Beaufort SST Depth 0 1 2 3 4 °C m
Time left	C	lose	st Di	ist m		Ave	rag	e Dist	n	Pho	otos by]	Frame no. start - end Photo Notes
Final GPS rea	ding	[B	eaufort		SST Depth Wind Dir Cloud cover
			S						E		2 3	4	
									•	_			
Species	Ma	x N	0.	Mi	n No).	Be	st est.	Ca	lves	Juv.	✓	Dominant behaviourGroup dynamics
													Travelling Milling Socialising Foraging Resting Clumped / Dispersed

Time	Behaviour	Response to boat: (avoid / neutral / boat friendly)
Seen		Comments:
Closed		
3		
6		
9		
12		
15		
18		
21		

APPENDIX 3: MAPS OF GROUP SIZES



Figure A3.1. The spatial distribution of southern right whales with the size of the points representing the different group sizes.



Figure A3.2. The spatial distribution of humpback whales with the size of the points representing the different group sizes.



Figure A3.3. The spatial distribution of Bryde's whales with the size of the points representing the different group sizes.



Figure A3.4. The spatial distribution of bottlenose dolphins with the size of the points representing the different group sizes.



Figure A3.5. The spatial distribution of humpback dolphins with the size of the points representing the different group sizes.



Figure A3.6. The spatial distribution of common dolphins with the size of the points representing the different group sizes.
APPENDIX 4: MAPS OF INDIVIDUALS PER UNIT EFFORT (IPUE)



Figure A4.1. Spatial distribution of cetacean IPUE. Cells represent low, medium and high IPUE.



Figure A4.2. Spatial distribution of southern right whale IPUE.



Figure A4.3. Spatial distribution of humpback whale IPUE.



Figure A4.4. Spatial distribution of Bryde's whale IPUE.



Figure A4.5. Spatial distribution of bottlenose dolphin IPUE.



Figure A4.6. Spatial distribution of humpback dolphin IPUE.



Figure A4.7. Spatial distribution of common dolphin IPUE.

- Acevedo, A., 1991. Interactions between boats and bottlenose dolphins, *Tursiops truncatus*, in the entrance to Ensenada De La Paz, Mexico. *Aquatic Mammals*, 17(3): 120-124.
- Agenbag, J.J., Richardson, A.J., Demarcq, H., Fréon, P., Weeks, S. and Shillington, F.A., 2003. Estimating environmental preferences of South African pelagic fish species using catch size- and remote sensing data. *Progress in Oceanography*, 59: 275-300.
- Akaike, H., 1973. Information theory and an extension of the maximum likelihood principle. In:B.N. Petrov and F. Csaki (Editors), *International Symposium on Information Theory*, Budapest, 268-281 pp.
- Allen, M.C., Read, A.J., Gaudet, J. and Sayigh, L.S., 2001. Fine-scale habitat selection of foraging bottlenose dolphins *Tursiops truncatus* near Clearwater, Florida. *Marine Ecology Progress Series*, 222: 253-264.
- Aragones, L.V., Jefferson, T.A. and Marsh, H., 1997. Marine mammal survey techniques applicable in developing countries. *Asian Marine Biology*, 14: 15-39.
- Armstrong, M.J., Chapman, P., Dudley, S.F.J., Hampton, I. and Malan, P.E., 1991. Occurrence and population structure of pilchard *Sardinops ocellatus*, round herring *Etrumeus whiteheadi* and anchovy *Engraulis capensis* off the east coast of southern Africa. *South African Journal of Marine Science*, 11: 227-249.
- Arthur, S.M., Manly, B.F.J., McDonald, L.L. and Garner, G.W., 1996. Assessing habitat selection when availability changes. *Ecology*, 77(1): 215-227.
- Atkins, S., Pillay, N. and Peddemors, V.M., 2004. Spatial distribution of Indo-Pacific humpback dolphins (*Sousa chinensis*) at Richards Bay, South Africa: environmental influences and behavioural patterns. *Aquatic Mammals*, 30(1): 84-93.
- Azzellino, A., Gaspari, S.A., Airoldi, S. and Lanfredi, C., 2008. Biological consequences of global warming: does sea surface temperature affect cetacean distribution in the western Ligurian Sea? *Journal of the Marine Biological Association of the UK*, 88: 1145-1152.
- Baier, T., 2009. StatConn DCOM. Version 2.2.
- Bailey, T.C. and Gatrell, A.C., 1995. *Interactive spatial data analysis*. Longman Scientific & Technical, Essex, England, 413 pp.
- Baird, D., 1971. Seasonal occurrence of the pilchard *Sardinops ocellata* on the east coast of South Africa. *Investigational Report, Division of Sea Fisheries, South Africa*, 96: 1-19.
- Ballance, L.T., 1992. Habitat use patterns and ranges of the bottlenose dolphin in the Gulf of California, Mexico. *Marine Mammal Science*, 8(3): 262-274.
- Balmer, B.C., Wells, R.S., Nowacek, S.M., Schwacka, L.H., McLellean, W.A., Scharf, F.S., Rowles, T.K., Hansen, L.J., Spradlin, T.R. and Pabst, D.A., 2008. Seasonal abundance and distribution patterns of common bottlenose dolphins (*Tursiops truncatus*) near St. Joseph Bay, Florida, USA. *Journal of Cetacean Resource Management*, 10(2): 157-167.
- Barendse, J., Best, P.B., Thornton, M., Pomilla, C., Carvalho, I. and Rosenbaum, H.C., 2010. Migration redefined? Seasonality, movements and group composition of humpback whales *Megaptera novaeangliae* off the west coast of South Africa. *African Journal of Marine Science*, 32(1): 1-22.

- Barlow, J., 1995. The abundance of cetaceans in California waters. Part I: Ship Surveys in summer and fall 1991. *Fishery Bulletin*, 93: 1-14.
- Barlow, J., 2006. Cetacean abundance in Hawaiian waters estimated from a summer/fall survey in 2002. *Marine Mammal Science*, 22(2): 446-464.
- Barry, S.C. and Welsh, A.H., 2002. Generalized additive modelling and zero inflated count data. *Ecological Modelling*, 157: 179-188.
- Baumgartner, M.F., 1997. The distribution of Risso's dolphin (*Grampus Griseus*) with respect to the physiography of the northern gulf of Mexico. *Marine Mammal Science*, 13(4): 614-638.
- Baumgartner, M.F., Mullin, K.D., May, N.L. and Leming, T.D., 2001. Cetacean habitats in the northern Gulf of Mexico. *Fishery Bulletin*, 99(2): 219-239.
- Bearzi, G. and Politi, E., 1999. Diurnal behavior of free-ranging bottlenose dolphins in the Kvarneric (northern Adriatic Sea). *Marine Mammal Science*, 15(4): 1065-1097.
- Bearzi, M., Politi, E., Agazzi, S. and Azzellino, A., 2006. Prey depletion caused by overfishing and the decline of marine megafauna in eastern Ionian Sea coastal waters (central Mediterranean). *Biological Conservation*, 79: 436-438.
- Bearzi, M., Saylan, C. and Hwang, A., 2009. Ecology and comparison of coastal and offshore bottlenose dolphins (*Tursiops truncatus*) in California. *Marine and Freshwater Research*, 60: 584-593.
- Becker, E.A., Forney, K.A., Ferguson, M.C., Foley, D.G., Smith, R.C., Barlow, J. and Redfern, J.V., 2010. Comparing California Current cetacean-habitat models developed using *in situ* and remotely sensed sea surface temperature data. *Marine Ecology Progress Series*, 413: 163-183.
- Beckley, L.E., 1977. A study of the littoral seaweed epifauna of St. Croix Island, Algoa Bay. M.Sc. Thesis. University of Port Elizabeth, South Africa. 168 pp.
- Beckley, L.E., 1983. Sea-surface temperature variability around Cape Recife, South Africa. *South African Journal of Science*, 79: 436-438.
- Beckley, L.E. and Buxton, C., 1989. Underwater observations of reef fish in and around Algoa Bay, South Africa. *Transactions of the Royal Society of South Africa*, 47(1): 29-38.
- Beckley, L.E. and McLachlan, A., 1979. Studies on the littoral seaweed epifauna of St. Croix Island. I. Physical and biological features of the littoral zone. *South African Journal of Zoology*, 14(4): 175-182.
- Beckley, L.E. and van der Lingen, C.D., 1999. Biology, fishery and management of sardines (Sardinops sagax). Marine and Freshwater Research, 50: 955-978.
- Bejder, L., Samuels, A., Whitehead, H., Gales, N., Mann, J., Connor, R., Heithaus, M., Watson-Capps, J., Flaherty, C. and Krützen, M., 2006. Decline in relative abundance of bottlenose dolphins exposed to long-term disturbance. *Conservation Biology*, 20(6): 1791-1798.
- Belgrano, J., Gribaudo, C., Arcucci, D., Kröhling, F. and Iñígues, M., 2008. Recent increase in the number of southern right whales (Eubalaena australis) in Golfo San Jorge, Santa Cruz, Patagonia Argentina. Paper SC/60/BRG4 presented to the Scientific Committee of the International Whaling Commission. Santiago, Chile. 10 pp.

- Best, P.B., 1977. Two allopatric forms of Bryde's whale off South Africa. *Report of the International Whaling Commission*, Special Issue 1: 10-38.
- Best, P.B., 1985. External characters of southern minke whales and the existence of a diminutive form. *Scientific Reports of the Whales Research Institute, Tokyo*, 36: 1-33.
- Best, P.B., 1990. Trends in the inshore right whale population off South Africa, 1969-1987. *Marine Mammal Science*, 6(2): 93-108.
- Best, P.B., 1994. A review of the catch statistics for modern whaling in southern Africa, 1908-1930. *Report of the International Whaling Commission*, 44: 1908-1930.
- Best, P.B., 2000. Coastal distribution, movements and site fidelity of right whales *Eubalaena australis* off South Africa, 1969-1998. *South African Journal of Marine Science*, 22: 43-55.
- Best, P.B., 2001. Distribution and population separation of Bryde's whales *Balaenoptera edeni* off southern Africa. *Marine Ecology Progress Series*, 220: 277-289.
- Best, P.B., 2007. *Whales and dolphins of the southern African subregion*. Cambridge University Press, Cape Town, 338 pp.
- Best, P.B., Brandão, A. and Butterworth, D.S., 2001b. Demographic parameters of southern right whales off South Africa. *Journal of Cetacean Resource Management*, Special Issue 2: 161-169.
- Best, P.B., Butterworth, D.S. and Rickett, L.H., 1984. An assessment cruise for the South African inshore stock of Bryde's whales (*Balaenoptera edeni*). *Report of the International Whaling Commission*, 34: 403-423.
- Best, P.B., Findlay, K.P., Sekiguchi, K., Peddemors, V.M., Rakotonirina, B., Rossouw, A. and Gove, D., 1998. Winter distribution and possible migration routes of humpback whales *Megaptera novaeangliae* in the southwest Indian Ocean. *Marine Ecology Progress Series*, 162: 287-299.
- Best, P.B., Peddemors, V.M., Cockcroft, V.G. and Rice, N., 2001a. Mortalities of right whales and related anthropogenic factors in South African waters, 1963-1998. *Journal of Cetacean Resource Management*, Special Issue 2: 171-176.
- Best, P.B., Reeb, D., Morais, M. and Baird, A., 1999. *A preliminary investigation of humpback whales off northern Angola*. Paper SC51/CAWS33 submitted to the Scientific Committee of the International Whaling Commission. Grenada. 12 pp.
- Best, P.B. and Rüther, H., 1992. Aerial photogrammetry of southern right whales, *Eubalaena* australis. Journal of Zoology, London, 228: 595-614.
- Best, P.B., Schaeff, C.M., Reeb, D. and Palsbøll, P.J., 2003. Composition and possible function of social groupings of southern right whales in South African waters. *Behaviour*, 140: 1469-1494.
- Best, P.B. and Sekiguchi, K., 1996. The distribution and abundance of humpback whales off southern Madagascar, August-September 1994. *Report of the International Whaling Commission*, 46: 323-331.
- Beyer, H.L., 2010. Geospatial Modelling Environment. Version 0.3.3 Beta.

- Bilgic, A. and Florkowski, W.J., 2007. Application of a hurdle negative binomial count data model to demand for bass fishing in the southeastern United States. *Journal of Environmental Management*, 83: 478-490.
- Binning, K. and Baird, D., 2001. Survey of heavy metals in the sediments of the Swartkops River Estuary, Port Elizabeth, South Africa. *Water SA*, 27(4): 461-466.
- Booth, A.J., 2004. *Spatial statistics and aquatic Geographic Information Systems*. GIS/ spatial analyses in fishery and aquatic sciences (Volume 2). Fishery-aquatic GIS research group Kawagoe, Saitama, Japan, December 2004. 3-44 pp.
- Bowen, W.D., 1997. Role of marine mammals in aquatic ecosystems. *Marine Ecology Progress Series*, 158: 267-274.
- Bräger, S., Harraway, J.A. and Manly, B.F.J., 2003. Habitat selection in a coastal dolphin species (*Cephalorhynchus hectori*). *Marine Biology*, 143: 233-244.
- Bremner, J.M., 1979a. The Bathymetry of Algoa Bay. Report No. 1980-0007, Geological Survey, Department of Mines. 13 pp.
- Bremner, J.M., 1979b. *Sediment dispersal processes in Algoa Bay*. 4th National Oceanographic Symposium. Cape Town. 17 pp.
- Bremner, J.M., 1983. Properties of logarithmic spiral beaches with particular reference to Algoa Bay. In: A. McLachlan and T. Erasmus (Editors), *Sandy beaches as ecosystems*. DR W. Junk Publishers, The Hague, 757 pp.
- Bremner, J.M., Du Plessis, A., Glass, J.G.K. and Day, R.W., 1991. Algoa Bay Marine Geoscientific Investigations. Geological Survey Bulletin 100. Pretoria. 173 pp.
- Bristow, T., Glanville, N. and Hopkins, J., 2001. Shore-based monitoring of bottlenosed dolphins (*Tursiops truncatus*) by trained volunteers in Cardigan Bay, Wales. *Aquatic Mammals*, 27(2): 115-120.
- Brown, J.H. and Lomolino, M.V., 1998. *Biogeography*, Second edition. Sinauer Associates, Sunderland, MA, 560 pp.
- Burnell, S.R. and Bryden, M.M., 1997. Coastal residence periods and reproductive timing in southern right whales, *Eubalaena australis. Journal of Zoology, London*, 241: 613-621.
- Campbell, G.S., Bilgre, B.A. and Defran, R.H., 2002. Bottlenose dolphins (*Tursiops truncatus*) in Turneffe Atoll, Belize: occurrence, site fidelity, group size, and abundance. *Aquatic Mammals*, 28(2): 170-180.
- Cañadas, A. and Hammond, P.S., 2006. Model-based abundance estimates for bottlenose dolphins off southern Spain: implications for conservation and management. *Journal of Cetacean Resource Management*, 8(1): 13-27.
- Cañadas, A. and Hammond, P.S., 2008. Abundance and habitat preferences of the short-beaked common dolphin *Delphinus delphis* in the southwestern Mediterranean: implications for conservation. *Endangered Species Research*, 4: 309-331.
- Cañadas, A., Sagarminaga, R., De Stephanis, R., Urquiola, E. and Hammond, P.S., 2005. Habitat preference modelling as a conservation tool: proposals for marine protected areas for cetaceans in southern Spanish waters. *Aquatic Conservation: Marine Freshwater Ecosystems*, 15: 495-521.

- Cañadas, A., Sagarminaga, R. and García-Tiscar, S., 2002. Cetacean distribution related with depth and slope in the Mediterranean waters off southern Spain. *Deep-Sea Research I*, 49: 2053-2073.
- Chilvers, B.L. and Corkeron, P.J., 2003. Abundance of Indo-Pacific bottlenose dolphins, *Tursiops* aduncus, off Point Lookout, Queensland, Australia. *Marine Mammal Science*, 19: 85-95.
- Christensen, M.S., 1980. Sea-surface temperature charts for southern Africa, south of 26°S. *South African Journal of Science*, 76: 541-546.
- Clapham, P.J., 2009. Humpback whale: *Megaptera novaeangliae*. In: W.F. Perrin, B. Würsig and J.G. Thewissen (Editors), *Encyclopaedia of marine mammals*, Second edition. Academic Press, United Kingdom, 582-585 pp.
- Clapham, P.J., Young, S.B. and Brownell Jr., R.L., 1999. Baleen whales: conservation issues and the status of the most endangered populations. *Mammal Review*, 29(1): 35-60.
- Clark, C.W. and Clapham, P.J., 2004. Acoustic monitoring on a humpback whale (*Megaptera novaeangliae*) feeding ground shows continual singing into late spring. *Proceedings of the Royal Society of London*, B 271: 1051-1057.
- Coastal and Environmental Services, n.d. *The Greater Addo Elephant National Park strategic* environmental assessment: Final. Grahamstown. 133 pp.
- Cockcroft, V.G. and Peddemors, V.M., 1990. Seasonal distribution and density of common dolphins *Delphinus delphis* off the south-east coast of southern Africa. *South African Journal of Marine Science*, 9: 371-377.
- Cockcroft, V.G. and Ross, G.J.B., 1989. Age, growth and reproduction of bottlenose dolphins *Tursiops truncatus* from the east coast of southern Africa. *Fishery Bulletin*, 88: 289-302.
- Cockcroft, V.G. and Ross, G.J.B., 1990. Food and feeding of the Indian Ocean bottlenose dolphin off southern Natal, South Africa. In: S. Leatherwood and R. Reeves (Editors), *The bottlenose dolphin*, 295-308 pp.
- Cockcroft, V.G., Ross, G.J.B., Peddemors, V.M. and Borchers, D.L., 1992. Estimates of abundance and undercounting of bottlenose dolphins off northern Natal, South Africa. *South African Journal of Wildlife Research*, 22: 102-109.
- Constantine, R., Brunton, D.H. and Dennis, T., 2004. Dolphin-watching tour boats change bottlenose dolphin (*Tursiops truncatus*) behaviour. *Biological Conservation*, 117: 229-307.
- Corkeron, P.J., 2004. Whale watching, iconography, and marine conservation. *Conservation Biology*, 18(3): 847-849.
- Costa, P., Praderi, R., Piedra, M. and Franco-Fraguas, P., 2005. Sightings of southern right whales, *Eubalaena australis*, off Uruguay. *Latin American Journal of Aquatic Mammals*, 4(2): 157-161.
- CSIR, 2004. Eastern Cape State of the Environment Report. CSIR Division of Water, Environment and Forestry Technology. Durban, South Africa. Produced on behalf of the Eastern Cape Department of Economic Affairs, Environment and Tourism. Bisho. 94-117 pp.
- CSIR, 2007. Proposed extension to the container berth and construction of an administration craft basin at the Port of Ngqura. Chapter 6: marine ecology, sediment toxicology and dredging. Draft scoping report. 63 pp.

- Cull, P., 2010. The Weekend Post, 18 October 2010. Waterfront 'would generate billions'. http://www.theherald.co.za/article.aspx?id=615940.
- Cummings, W.C., 1985a. Right whales *Eubalaena glacialis* and *Eubalaena glacialis*. In: S.H. Ridgway and S.R. Harrison (Editors), *Handbook of marine mammals, Volume 3: The sirenians and baleen whales*. Academic Press, Florida, 275-304 pp.
- Cummings, W.C., 1985b. Bryde's whale Balaenoptera edeni. In: S.H. Ridgway and S.R. Harrison (Editors), Handbook of marine mammals, Volume 3: The sirenians and baleen whales. Academic Press, Florida, 137-154 pp.
- Cunningham, R.B. and Lindenmayer, D.B., 2005. Modeling count data of rare species: some statistical issues. *Ecology*, 86(5): 1135-1142.
- Dalrymple, M.L., Hudson, I.L. and Ford, R.P.K., 2003. Finite mixture, zero-inflated poisson and hurdle models with application to SIDS. *Computational Statistics and Data Analysis*, 41: 491-504.
- Davis, R.W., Fargion, G.S., May, N., Leming, T.D., Baumgartner, M., Evans, W.E., Hansen, L.J. and Mullin, K., 1998. Physical habitat of cetaceans along the continental slope in the north-central and western Gulf of Mexico. *Marine Mammal Science*, 14(3): 490-507.
- Davis, R.W., Ortega-Ortiz, J.G., Ribic, C.A., Evans, W.E., Biggs, D.C., Ressler, P.H., Cady, R.B., Leben, R.R., Mullin, K.D. and Würsig, B., 2002. Cetacean habitat in the northern oceanic Gulf of Mexico. *Deep-Sea Research*, 49: 121-142.
- Dawbin, W.H., 1966. The seasonal migratory cycle of humpback whales. In: K.S. Norris (Editor), *Whales, dolphins and porpoises*. University of California Press, Berkeley and Los Angeles, 145-170 pp.
- Dawson, S., Wade, P., Slooten, E. and Barlow, J., 2008. Design and field methods for sighting surveys of cetaceans in coastal and riverine habitats. *Mammal Review*, 38(1): 19-49.
- de Bruyn, C., 2010. Engineering News, 29 June 2010. Coega oil refinery ranks among top global infrastructure projects. http://www.engineeringnews.co.za/article/coega-oil-refinery-ranks-among-top-global-infrastructure-projects-2010-06-29.
- de Boer, M.N., 2010. Spring distribution and density of minke whale *Balaenoptera acutorostrata* along an offshore bank in the central North Sea. *Marine Ecology Progress Series*, 408: 265-274.
- Deakos, M.H., Branstetter, B.K., Mazzuca, L., Fertl, D. and Mobley, J.R.J., 2010. Two unusual interactions between a bottlenose dolphin (*Tursiops truncatus*) and a humpback whale (*Megaptera novaeangliae*) in Hawaiian waters. *Aquatic Mammals*, 46(2): 121-128.
- Defran, R.H. and Weller, D.W., 1999. Occurrence, distribution, site fidelity, and school size of bottlenose dolphins (*Tursiops truncatus*) off San Diego, California. *Marine Mammal Science*, 15(2): 366-380.
- Durham, B., 1994. The distribution and abundance of the humpback dolphin (Sousa chinensis) along the Natal coast, South Africa. M.Sc. Thesis. University of Natal, Pietermaritzburg. 83 pp.
- Elwen, S.H. and Best, P.B., 2004. Environmental factors influencing the distribution of southern right whales (*Eubalaena Australis*) on the south coast of South Africa I: broad scale patterns. *Marine Mammal Science*, 20(3): 567-582.

- Elwen, S.H., Best, P.B., Reeb, D. and Thornton, M., 2009. Diurnal movements and behaviour of Heaviside's dolphins, *Cephalorhynchus heavisidii*, with some comparative data for dusky dolphins, *Lagenorhynchus obscurus*. *South African Journal of Wildlife Research*, 39(2): 143-154.
- ESRI Inc., 2007. ArcGIS 9.2 desktop help. http://webhelp.esri.com/arcgisdesktop/9.2/index.cfm?TopicName=Thiessen%20%28Cove rage%29. Accessed: 01/11/2010.
- ESRI® Inc., 2008. ArcMapTM. Version 9.3.
- Evans, P.G.H. and Hammond, P.S., 2004. Monitoring cetaceans in European waters. *Mammal Review*, 34(1): 131-156.
- Evans, W.E., 1994. Common dolphin, white-bellied porpoise *Delphinus delphis*, Linnaeus, 1758. In: S.H. Ridgway and S.R. Harrison (Editors), *Handbook of marine mammals, Volume 5: The first book of dolphins*. Academic Press, London, 191-224 pp.
- Filby, N.E., Bossley, M., Sanderson, K.J., Martinez, E. and Stockin, K.A., 2010. Distribution and population demographics of common dolphins (*Delphinus delphis*) in the Gulf St. Vincent, South Australia. *Aquatic Mammals*, 36(1): 33-45.
- Findlay, K.P. and Best, P.B., 1995. Summer incidence of humpback whales on the west coast of South Africa. *South African Journal of Marine Science*, 15: 279-282.
- Findlay, K.P., Best, P.B., Peddemors, V.M. and Gove, D., 1994. The distribution and abundance of humpback whales on their southern and central Mozambique winter grounds. *Report of the International Whaling Commission*, 44: 331-320.
- Findlay, K.P., Best, P.B., Ross, G.J.B. and Cockcroft, V.G., 1992. The distribution of small odontocete cetaceans off the coasts of South Africa and Namibia. *South African Journal of Marine Science*, 12: 237-270.
- Fotheringham, A.S., Brunsdon, C. and Charlton, M., 2000. *Quantitative geography: perspectives* on spatial data analysis. SAGE Publications, Ltd., London, 270 pp.
- Frantzis, A. and Herzing, D.L., 2002. Mixed-species associations of striped dolphins (*Stenella coeruleoalba*), short-beaked common dolphins (*Delphinus delphis*), and Risso's dolphins (*Grampus griseus*) in the Gulf of Corinth (Greece, Mediterranean Sea). Aquatic Mammals, 28(2): 188-197.
- Friedmann, Y. and Daly, B. (Editors), 2004. *Red data book of the mammals of South Africa: a conservation assessment*. CBSG Southern Africa, Conservation Breeding Specialist Group (SSC/IUCN), Endangered Wildlife Trust., South Africa, 722 pp.
- Fury, C.A. and Harrison, P.L., 2008. Abundance, site fidelity and range patterns of Indo-Pacific bottlenose dolphins (*Tursiops aduncus*) in two Australian subtropical estuaries. *Marine* and Freshwater Research, 59: 1015-1027.
- Gilles, A., Scheidat, M. and Siebert, U., 2009. Seasonal distribution of harbour porpoises and possible interference of offshore wind farms in the German North Sea. *Marine Ecology Progress Series*, 383: 295-307.
- Gómez de Segura, A., Hammond, P.S. and Raga, J.A., 2008. Influence of environmental factors on small cetacean distribution in the Spanish Mediterranean. *Journal of the Marine Biological Association of the UK*, 88(6): 1185-1192.

- Goschen, W.S., 1991. Ocean dynamics associated with prominent capes and bays off the south coast of South Africa. Ph.D. Thesis. University of Port Elizabeth, South Africa. 313 pp.
- Goschen, W.S. and Schumann, E.H., 1988. Ocean current and temperature structures in Algoa Bay and beyond in November 1986. *South African Journal of Marine Science*, 7: 101-116.
- Government Gazette, 2008. Gazette Number: 31209, 4 July 2008. General notice 722: Marine Living Resources Act (18/1998): Policy on boat-based whale and dolphin watching. Pretoria.
- Government Gazette, 2009a. Government Gazette Media Statement, 17 August 2009. Allocation process for boat-based whale watching and white shark cage diving operating permits to commence. Pretoria.
- Government Gazette, 2009b. Gazette Number: 32492, 14 August 2009. Government Notice 839. Marine Living Resources Act (18/1998): Invitation to apply for permits to undertake boatbased whale and dolphin watching and white shark cage diving. Pretoria.
- Griffiths, C.L., Robinson, T.B., Lange, L. and Mead, A., 2010. Marine biodiversity in South Africa: an evaluation of current states of knowledge. *PLoS ONE*, 5(8): 1-13.
- Guinet, C., Dubroca, L., Lea, M.A., Goldsworthy, S., Cherel, Y., Duhamel, G., Bonadonna, F. and Donnay, J., 2001. Spatial distribution in female Antarctic fur seals Arctocephalus gazella in relation to oceanographic variables: a scale-dependent approach using geographic information systems. Marine Ecology Progress Series, 219: 251-264.
- Guisan, A., Edwards Jr., T.C. and Hastie, T., 2002. Generalized linear and generalized additive models in studies of species distributions: setting the scene. *Ecological Modelling*, 157: 89-100.
- Guisan, A. and Thuiller, W., 2005. Predicting species distribution: offering more than simple habitat models. *Ecology letters*, 8: 993-1009.
- Guissamulo, A. and Cockcroft, V.G., 2004. Ecology and population estimates of Indo-Pacific humpback dolphins (*Sousa chinensis*) in Maputo Bay, Mozambique. *Aquatic Mammals*, 30(1): 94-102.
- Gurmu, S., 1997. Semi-parametric estimation of hurdle regression models with an application to medicaid utilization. *Journal of Applied Econometrics*, 12(3): 225-242.
- Hammond, P.S., 2009. Estimating the abundance of marine mammals. In: I.L. Boyd, D. Bowen and S. Iverson (Editors), *Marine mammal ecology and conservation: a handbook of techniques*. Oxford University Press, Oxford, UK, 448 pp.
- Hammond, P.S., Bearzi, G., Bjørge, A., Forney, K., Karczmarski, L., Kasuya, T., Perrin, W.F., Scott, M.D., Wang, J.Y., Wells, R.S. and Wilson, B., 2008a. *Delphinus capensis*. IUCN Red List of Threatened Species 2010. Version 2010.2. www.iucnredlist.org. Accessed: 12/08/10.
- Hammond, P.S., Bearzi, G., Bjørge, A., Forney, K., Karczmarski, L., Kasuya, T., Perrin, W.F., Scott, M.D., Wang, J.Y., Wells, R.S. and Wilson, B., 2008b. *Tursiops aduncus*. IUCN Red List of Threatened Species 2010. Version 2010.2. www.iucnredlist.org. Accessed: 12/08/10.
- Hanson, M.T. and Defran, R.H., 1993. The behaviour and feeding ecology of the Pacific coast bottlenose dolphin, *Tursiops truncatus*. *Aquatic Mammals*, 19(3): 127-142.

- Harrison, T.D. and Whitfield, A.K., 1990. Composition, distribution and abundance of ichthyoplankton in the Sundays River estuary. *South African Journal of Zoology*, 25(3): 161-168.
- Harwood, J., 2001. Marine mammals and their environment in the Twenty-First Century. *Journal* of Mammalogy, 82(3): 630-640.
- Hastie, G.D., Barton, T.R., Grellier, K., Hammond, P.S., Swift, R.J., Thompson, P.M. and Wilson, B., 2003. Distribution of small cetaceans within a candidate special area of conservation; implications for management. *Journal of Cetacean Resource Management*, 5(3): 261-266.
- Hastie, T. and Tibshirani, R., 1987. Generalized additive models: some applications. *Journal of the American Statistical Association*, 82(398): 371-386.
- Hastie, T. and Tibshirani, R., 1990. *Generalised additive models*. Chapman & Hall, London, 335 pp.
- Hengl, T., 2006. Finding the right pixel size. Computers and Geosciences, 32: 1283-1298.
- Heyning, J.E. and Perrin, W.F., 1994. Evidence for two species of common dolphins (genus Delphinus) from the eastern North Pacific. Natural History Museum Los Angeles, Contributions in Science no. 442: 1-35.
- Holt, R.S., Gerrodette, T. and Cologne, J.B., 1987. Research vessel survey design for monitoring dolphin abundance in the eastern tropical Pacific. *Fishery Bulletin*, 85(3): 435-446.
- Hooker, S.K. and Gerber, L.R., 2004. Marine reserves as a tool for ecosystem-based management: the potential importance of megafauna. *BioScience*, 54(1): 27-39.
- Hooker, S.K., Whitehead, H. and Gowans, S., 1999. Marine protected area design and the spatial and temporal distribution of cetaceans in a submarine canyon. *Conservation Biology*, 13(3): 592-602.
- Hoyt, E., 2001. Whale watching 2001: Worldwide tourism number, expenditures and expanding socioeconomic benefits. International Fund for Animal Welfare. Crowborough, UK. 158 pp.
- Hoyt, E., 2005. *Marine protected areas: for whales, dolphins and porpoises*. Earthscan, London, 492 pp.
- Hui, C.A., 1979. Undersea topography and distribution of dolphins of the genus Delphinus in the southern California Bight. *Journal of Mammalogy*, 60(3): 521-527.
- Illenberger, W.K., 1986. The Alexandria coastal dunefield: morphology, sand budget and history. M.Sc. Thesis. University of Port Elizabeth, South Africa. 87 pp.
- Ingram, S.N. and Rogan, E., 2002. Identifying critical areas and habitat preferences of bottlenose dolphins *Tursiops truncatus*. *Marine Ecology Progress Series*, 244: 247-255.
- International Whaling Commission, 1986. *Right whales, past and present status*. Report of the International Whaling Commission, Special Issue 10, 289 pp.
- International Whaling Commission, 2009. Annex L. Report of the sub-committee on small cetaceans. *Journal of Cetacean Resource Management*, 11 (Supplement): 1-22.
- Jefferson, T.A., 2000. Population biology of the Indo-Pacific hump-backed dolphin in Hong Kong waters. *Wildlife Monographs*, 144: 1-65.

Jefferson, T.A. and Karczmarski, L., 2001. Sousa chinensis. Mammalian Species, 655: 1-9.

- Jefferson, T.A. and Schiro, A.J., 1997. Distribution of cetaceans in the offshore Gulf of Mexico. *Mammal Review*, 27(1): 27-50.
- Karczmarski, L., 1996. Ecological studies of humpback dolphins *Sousa chinensis* in the Algoa Bay region, Eastern Cape, South Africa. Ph.D. Thesis. University of Port Elizabeth. 202 pp.
- Karczmarski, L., 1999. Group dynamics of humpback dolphins (*Sousa chinensis*) in Algoa Bay region, South Africa. *Journal of Zoology, London*, 249: 283-293.
- Karczmarski, L., 2000. Conservation and management of humpback dolphins: the South African perspective. *Oryx*, 34(3): 207-216.
- Karczmarski, L. and Cockcroft, V.G., 1999. Daylight behaviour of humpback dolphins *Sousa* chinensis in Algoa Bay, South Africa. Zeitschrift für Säugetierkunde, 64: 19-29.
- Karczmarski, L., Cockcroft, V.G. and McLachlan, A., 1999. Group size and seasonal pattern of occurrence of humpback dolphins *Sousa chinensis* in Algoa Bay, South Africa. *South African Journal of Marine Science*, 21: 89-97.
- Karczmarski, L., Cockcroft, V.G. and McLachlan, A., 2000a. Habitat use and preferences of Indo-Pacific humpback dolphins *Sousa chinensis* in Algoa Bay, South Africa. *Marine Mammal Science*, 16(1): 65-79.
- Karczmarski, L., Cockcroft, V.G., McLachlan, A. and Winter, P.E.D., 1998. Recommendations for the conservation and management of humpback dolphins in the Algoa Bay region, South Africa. *Koedoe*, 41(2): 121-129.
- Karczmarski, L., Thornton, M. and Cockcroft, V.G., 1997. Description of selected behaviours of humpback dolphins *Sousa chinensis*. *Aquatic Mammals*, 23(3): 127-133.
- Karczmarski, L., Thornton, M. and Cockcroft, V.G., 2000b. Daylight occurrence of humpback dolphins *Sousa chinensis* in Algoa Bay, South Africa. *African Journal of Ecology*, 38: 86-90.
- Kaschner, K., Watson, R., Trites, A.W. and Pauly, D., 2006a. Forage fish consumption by marine mammals and seabirds. In: Alder, J. and Pauly, D. (Editors), On the multiple uses of forage fish: from ecosystems to markets. *Fisheries Centre Research Reports*, 14: 33-46.
- Kaschner, K., Watson, R., Trites, A.W. and Pauly, D., 2006b. Mapping world-wide distributions of marine mammal species using a relative environmental suitability (RES) model. *Marine Ecology Progress Series*, 316: 285-310.
- Kato, H. and Perrin, W.F., 2009. Bryde's whales: Balaenoptera edeni/brydei. In: W.F. Perrin, B. Würsig and J.G. Thewissen (Editors), Encyclopaedia of marine mammals, Second Edition. Academic Press, United Kingdom, 158-163 pp.
- Katona, S. and Whitehead, H., 1988. Are cetacea ecologically important? *Oceanography and Marine Biology: Annual Review*, 26: 553-568.
- Keiper, C.A., Ainley, D.G., Allen, S.G. and Harvey, J.T., 2005. Marine mammal occurrence and ocean climate off central California, 1986 to 1994 and 1997 to 1999. *Marine Ecology Progress Series*, 289: 285-306.
- Keith, M., 1999. Population biology of humpback dolphins in Richards Bay, South Africa. University of Pretoria. 107 pp.

- Keith, M., Peddemors, V.M., Bester, M.N. and Ferguson, J.W.H., 2002. Population characteristics of Indo-Pacific humpback dolphins at Richards Bay, South Africa: implications for incidental capture in shark nets. South African Journal of Wildlife Research, 32(2): 153-162.
- Kenney, R.D., Scott, G.P., Thompson, T.J. and Winn, H.E., 1997. Estimates of prey consumption and trophic impacts of cetaceans in the U.S. northeast continental shelf ecosystem. *Journal of Northwest Atlantic Fishery Science*, 22: 155-171.
- Kenney, R.D. and Winn, H.E., 1986. Cetacean high-use habitats of the northeast United States continental shelf. *Fishery Bulletin*, 84(2): 345-357.
- Kernohan, B., 2010. The Weekend Post, 26 March 2010. Mandela Bay oil 'on tap from 2016'. http://www.weekendpost.co.za/business/article.aspx?id=546271.
- Klages, N.T.W., 2006. Establishment of the Port Elizabeth humpback dolphin sanctuary and the demarcation of restricted areas for aquatic motorsport craft. Arcus Gibb, Port Elizabeth. 13 pp.
- Klages, N.T.W., Willis, A.B. and Ross, G.J.B., 1992. Variability in the diet of the Cape gannet at Bird Island, Algoa Bay, South Africa. *South African Journal of Marine Science*, 12: 761-771.
- Klinowska, M., 1991. Dolphins, porpoises and whales of the world. The IUCN red data book. ICUN Gland, Switzerland and Cambridge, U.K., 429 pp.
- Lasiak, T.A., 1982. Structural and functional aspects of the surf-zone community in the Eastern Cape. Ph.D. Thesis. University of Port Elizabeth. 459 pp.
- Lasiak, T.A., 1983. The impact of surf-zone fish communities on aunal assemblages associated with sandy beaches. In: A. McLachlan and T. Erasmus (Editors), *Sandy beaches as ecosystems*. DR W. Junk Publishers, The Hague, 757 pp.
- Laws, R.M., 1977. Seals and whales of the Southern Ocean. *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences*, 279(963): 81-96.
- Legendre, P., 1993. Spatial autocorrelation: trouble or new paradigm? *Ecology*, 76(6): 1659-1673.
- Lockyer, C., 1990. Review of incidents involving wild sociable dolphins worldwide. In: S. Leatherwood and R. Reeves (Editors), *The bottlenose dolphin*. Academic Press, San Diego, 337-353 pp.
- Longley, P.A., Goodchild, M.F., Maguire, D.J. and Rhind, D.W., 2005. *Geographic information systems*, Second edition. John Wiley & Sons, Ltd., West Sussex, 517 pp.
- Lusseau, D. and Higham, J.E.S., 2004. Managing the impacts of dolphin-based tourism through the definition of critical habitats: the case of bottlenose dolphins (*Tursiops* spp.) in Doubtful Sound, New Zealand. *Tourism Management*, 25: 657-667.
- Lusseau, D., Wilson, B., Hammond, P.S., Grellier, K., Durban, J.W., Parsons, K.M., Barton, T.R. and Thompson, P.M., 2006. Quantifying the influence of sociality on population structure in bottlenose dolphins. *Journal of Animal Ecology*, 75(1): 14-24.
- Lutjeharms, J.R.E., 1981. Features of the southern Agulhas Current Circulation from satellite remote sensing. *South African Journal of Science*, 77: 231-236.

- Lutjeharms, J.R.E., Monteiro, P.M.S., Tyson, P.D. and Obura, D., 2001. The oceans around southern Africa and regional effects of global change. *South African Journal of Science*, 97: 119-120.
- MacLeod, C.D., 2009. Global climate change, range changes and potential implications for the conservation of marine cetaceans: a review and synthesis. *Endangered Species Research*, 7: 125-136.
- MacLeod, C.D., Bannon, S.M., Pierce, G.J., Schweder, C., Learmonth, J.A., Herman, J.S. and Reid, R.J., 2005. Climate change and the cetacean community of north-west Scotland. *Biological Conservation*, 124: 477-483.
- Maguire, D.J., Batty, M. and Goodchild, M.F. (Editors), 2005. *GIS, spatial analysis, and modeling*. ESRI Press, Redlands, 480 pp.
- Mann, J., 1999. Behavioural sampling methods for cetaceans: a review and critique. *Marine Mammal Science*, 15(1): 102-122.
- Mann, J., Connor, R.C., Barre, L.M. and Heithaus, M.R., 2000. Female reproductive success in bottlenose dolphins (*Tursiops* sp.): life history, habitat, provisioning, and group-size effects. *Behavioural Ecology*, 11(2): 210-219.
- Mate, B.R., Best, P.B., Lagerquist, B.A. and Winsor, M.H., 2010. Coastal, offshore, and migratory movements of South African right whales revealed by satellite telemetry. *Marine Mammal Science*, no. doi: 10.1111/j.1748-7692.2010.00412.x.
- Maunder, M.N. and Punt, A.E., 2004. Standardizing catch and effort data: a review of recent approaches. *Fisheries Research*, 70: 141-159.
- McCullagh, P., 2002. What is a statistical model? The Annals of Statistics, 30(5): 1225-1310.
- McKillup, S., 2006. *Statistics explained: an introductory guide for life scientists*. Cambridge University Press, Cambridge, 267 pp.
- McLachlan, A., 1983. Sandy beach ecology a review. In: A. McLachlan and T. Erasmus (Editors), *Sandy beaches as ecosystems*. DR W. Junk Publishers, The Hague, 757 pp.
- Melville-Smith, R. and Baird, D., 1980. Abundance, distribution and species composition of fish larvae in the Swartkops estuary. *South African Journal of Zoology*, 15(2): 72-78.
- Merriman, M.G., Markowitz, T.M., Harlin-Cognato, A.D. and Stockin, K.A., 2009. Bottlenose dolphins (*Tursiops truncatus*) abundance, site fidelity, and group dynamics in the Marlborough Sounds, New Zealand. *Aquatic Mammals*, 35(4): 511-522.
- Microsoft® Office Excel, 2003. Microsoft Office.
- Mignucci-Giannoni, A.A., 1998. Zoogeography of cetaceans off Puerto Rico and the Virgin Islands. *Caribbean Journal of Science*, 34(3-4): 173-190.
- Millar, D.H., 2000. Distribution and abundance of Cape hakes (Merluccius capensis and Merluccius paradoxus) in relation to environmental variation in the southern Benguela system. M.Sc. Thesis. University of Cape Town, Cape Town. 129 pp.
- Moore, S.E., DeMaster, D.P. and Dayton, P.K., 2000. Cetacean habitat selection in the Alaskan Arctic during summer and autumn. *Arctic*, 53(4): 432-447.
- Mullahy, J., 1986. Specification and testing of some modified count data models. *Journal of Econometrics*, 33: 341-365.

- Mullin, K.D., Hoggard, W., Roden, C.L., Lohoefener, R.R., Rogers, C.M. and Taggart, B., 1994. Cetaceans on the upper continental slope in the north-central Gulf of Mexico. *Fishery Bulletin*, 92: 773-786.
- Nelder, J.A. and Wedderburn, R.W.M., 1972. Generalized linear models. *Journal of the Royal Statistical Society. Series A (General)*, 135(3): 370-384.
- Nelson, T.A., Duffus, D., Robertson, C., Laberee, K. and Feyrer, L.J., 2009. Spatial-temporal analysis of marine wildlife. *Journal of Coastal Research*, Special Issue 56: 1537-1541.
- Nelson, T.A., Duffus, D.A., Robertson, C. and Feyrer, L.J., 2008. Spatial-temporal patterns in intra-annual gray whale foraging: characterizing interactions between predators and prey in Clayquot Sound, British Columbia, Canada. *Marine Mammal Science*, 24(2): 356-370.
- NMBM, 2010. Waterfront development to south of harbour, including King's Beach Precinct. http://www.nelsonmandelabay.gov.za/Content.aspx?objID=460. Accessed: 25/10/2010.
- Norman, S.A., 2008. Spatial epidemiology and GIS in marine mammal conservation medicine and disease research. *EcoHealth*, 5: 257-267.
- Nowacek, D.P., Thorne, L.H., Johnston, D.W. and Tyak, P.L., 2007. Responses of cetaceans to anthropogenic noise. *Mammal Review*, 37(2): 811-115.
- O'Callaghan, T.M. and Baker, C.S., 2002. Summer cetacean community, with particular reference to Bryde's whales, in the Hauraki Gulf, New Zealand. DOC Science Internal Series 55. New Zealand Department of Conservation. 18 pp.
- O'Donoghue, S.H., Drapeau, L. and Peddemors, V.M., 2010a. Broad-scale distribution patterns of sardine and their predators in relation to remotely sensed environmental conditions during the KwaZulu-Natal sardine run. *African Journal of Marine Science*, 32(2): 279-291.
- O'Donoghue, S.H., Whittington, P.A., Dyer, B.M. and Peddemors, V.M., 2010b. Abundance and distribution of avian and marine mammal predators of sardine observed during the 2005 KwaZulu-Natal sardine run survey. *African Journal of Marine Science*, 32(2): 361-374.
- Omura, H., 1959. Bryde's whale from the coast of Japan. Scientific Reports of the Whales Research Institute, Tokyo, 14: 1-33.
- Oosthuizen, W.H., 1991. General movements of South Africa (Cape) fur seals Arctocephalus pusillus from analysis of recoveries of tagged animals. South African Journal of Marine Science, 11: 21-29.
- O'Rourke, D. and Connolly, S., 2003. Just oil? The distribution of environmental and social impacts of oil production and consumption. *Annual Review of Environmental Resources*, 28: 587-617.
- Paradis, E., 2010. Moran's autocorrelation coefficient in comparative methods. R Foundation for statistical computing, Vienna, Austria. http://cran.r-project.org/.
- Parra, G.J., 2006. Resource partitioning in sympatric delphinids: space use and habitat preferences of Australian snubfin and Indo-Pacific humpback dolphins. *Journal of Animal Ecology*, 75: 862-874.
- Parra, G.J. and Ross, G.J.B., 2009. Humpback dolphins: S. chinensis and S. teuszii. In: G.J. Parra,
 B. Würsig and J.G. Thewissen (Editors), Encyclopaedia of marine mammals. Academic Press, United Kingdom, 576-582 pp.

- Parra, G.J., Schick, R. and Corkeron, P.J., 2006. Spatial distribution and environmental correlates of Australian snubfin Indo-Pacific humpback dolphins. *Ecography*, 29: 396-406.
- Patenaude, N.J., 2003. *Sightings of southern right whales around 'mainland' New Zealand*. DOC Science Publishing, Science and Research Unit. New Zealand Department of Conservation. 36 pp.
- Pattrick, P. and Strydom, N.A., 2008. Composition, abundance, distribution and seasonality of larval fishes in the shallow nearshore of the proposed Greater Addo Marine Reserve, Algoa Bay, South Africa. *Estuarine, Coastal and Shelf Science*, 79: 251-262.
- Payne, R., 1986. Long term behavioral studies of the southern right whale (*Eubalaena australis*). *Report of the International Whaling Commission*, Special Issue 10: 161-167.
- Payne, R., Rowntree, V., Perkins, J.S., Cooke, J.G. and Lankester, K., 1990. Population size, trends and reproductive parameters of right whales (*Eubalaena australis*) off Peninsula Valdes, Argentina. *Report of the International Whaling Commission*, Special Issue 12: 271-278.
- Peddemors, V.M., 1999. Delphinids of southern Africa: a review of their distribution, status and life history. *Journal of Cetacean Resource Management*, 1(2): 157-165.
- Penry, G.S., 2010. The biology of South African Bryde's whales. Ph.D. Thesis. University of St. Andrews. 173 pp.
- Perrin, W.F., Würsig, B. and Thewissen, J.G. (Editors), 2009. *Encyclopedia of marine mammals*, Second edition. Academic Press, United Kingdom, 1316 pp.
- Peschak, T.P., 2005. The sardine run: the greatest shoal on earth. In: T.P. Peschak (Editor), *Currents of contrast: life in southern Africa's two oceans*. Struik Publishers, Cape Town, 200 pp.
- Picanço, C., Carvalho, I. and Brito, C., 2009. Occurrence and distribution of cetaceans in São Tomé and Príncipe tropical archipelago and their relation to environmental variables. *Journal of the Marine Biological Association of the UK*, 89(5): 1071-1076.
- Pierpoint, C., Allan, L., Arnold, H., Peter, E., Perry, S., Wilberforce, L. and Baxter, J., 2009. Monitoring important coastal sites for bottlenose dolphin in Cardigan Bay, UK. *Journal of the Marine Biological Association of the United Kingdom*, 89(5): 1033-1043.
- Ports and Ships, 2010. Ports of southern Africa. http://ports.co.za/port-elizabeth.php. Accessed: 26/05/10.
- Qi, Y. and Wu, J., 1996. Effects of changing spatial resolution on the results of landscape pattern analysis using spatial autocorrelation indices. *Landscape Ecology*, 11(1): 39-49.
- Randall, B.M., Randall, R.M. and Compagno, L.J.V., 1988. Injuries to jackass penguins (*Spheniscus demersus*): evidence for shark involvement. *Journal of Zoology, London*, 214(4): 589-599.
- Redfern, J.V., Ferguson, M.C., Becker, E.A., Hyrenbach, K.D., Good, C., Barlow, J., Kaschner, K., Baumgartner, M.F., Forney, K.A., Ballance, L.T., Fauchald, P., Halpin, P., Hamazaki, T., Pershing, A.J., Qian, S.S., Read, A., Reilly, S.B., Torres, L. and Werner, F., 2006. Techniques for cetacean-habitat modeling. *Marine Ecology Progress Series*, 310: 271-295.
- Reeves, R.R., Dalebout, M.L., Jefferson, T.A., Karczmarski, L., Laidre, K., O'Corry-Crowe, G., Rojas-Bracho, L., Secchi, E.R., Slooten, E., Smith, B.D., Wang, J.Y. and Zhou, K., 2008.

Sousa chinensis. IUCN Red List of Threatened Species 2010. Version 2010.2. www.iucnredlist.org. Accessed: 12/08/10.

- Reilly, S.B., 1990. Seasonal changes in distribution and habitat differences among dolphins in the eastern tropical Pacific. *Marine Ecology Progress Series*, 66: 1-11.
- Reilly, S.B., Bannister, J.L., Best, P.B., Brown, M., Brownell Jr., R.L., Butterworth, D.S., Clapham, P.J., Cooke, J., Donovan, G.P., Urbán, J. and Zerbini, A.N., 2008a. *Balaenoptera edeni*. IUCN Red List of Threatened Species 2010. Version 2010.2. www.iucnredlist.org. Accessed: 12/08/10.
- Reilly, S.B., Bannister, J.L., Best, P.B., Brown, M., Brownell Jr., R.L., Butterworth, D.S., Clapham, P.J., Cooke, J., Donovan, G.P., Urbán, J. and Zerbini, A.N., 2008b. *Eubalaena australis*. IUCN Red List of Threatened Species 2010. Version 2010.2. www.iucnredlist.org. Accessed: 12/08/10.
- Reilly, S.B., Bannister, J.L., Best, P.B., Brown, M., Brownell Jr., R.L., Butterworth, D.S., Clapham, P.J., Cooke, J., Donovan, G.P., Urbán, J. and Zerbini, A.N., 2008c. *Megaptera novaeangliae*. IUCN Red List of Threatened Species 2010. Version 2010.2. www.iucnredlist.org. Accessed: 12/08/10.
- Reilly, S.B. and Fiedler, P.C., 1994. Interannual variability of dolphin habitats in the eastern tropical Pacific. I: Research vessel surveys, 1986-1990. *Fishery Bulletin*, 92: 434-450.
- Reilly, S.B., Fiedler, P.C., Forney, K.A. and Barlow, J., 1997. Cetacean habitats in the California Current: partitioning oceanographic and geographic patterns. Report No. SC/49/022. 23 pp.
- Reisinger, R.R. and Karczmarski, L., 2009. Population size estimate of Indo-Pacific bottlenose dolphins in the Algoa Bay region, South Africa. *Marine Mammal Science*, 26(1): 86-97.
- Reyes, J.C., 1996. A possible case of hybridism in wild dolphins. *Marine Mammal Science*, 12(2): 301-307.
- Rice, D.W., 1998. *Marine mammals of the world: systematics and distribution*, Special publication: Number 4. The Society for Marine Mammalogy, 231 pp.
- Richardson, W.J., Green Jr., C.R., Malme, C.I. and Thomson, D.H., 1995. *Marine mammals and noise*. Academic Press, Inc., San Diego, California, 576 pp.
- Robinson, K.P., Tetley, M.J. and Mitchelson-Jacob, E.G., 2009. The distribution and habitat preference of coastally occurring minke whales (*Balaenoptera acutorostrata*) in the outer southern Moray Firth, northeast Scotland. *Journal of Coastal Conservation*, 13: 39-48.
- Ross, G.J.B., 1977. The taxonomy of bottlenosed dolphins *Tursiops* species in South African waters, with notes on their biology. *Annals of the Cape Provincial Museums (Natural History)*, 11(14): 259-327.
- Ross, G.J.B., 1984. The smaller cetaceans of the south east coast of southern Africa. Annals of the Cape Provincial Museums (Natural History), 15(Part 2): 173-410.
- Ross, G.J.B. and Best, P.B., 1989. Smaller whales and dolphins. In: A.I.L. Payne and R.J.M. Crawford (Editors), *Oceans of life off southern Africa*. Vlaeberg Publishers, Cape Town, 303-314 pp.
- Ross, G.J.B., Cockcroft, V.G. and Butterworth, D.S., 1987. Offshore distribution of bottlenosed dolphins in Natal coastal waters and Algoa Bay, Eastern Cape. *South African Journal of Zoology*, 22(1): 50-56.

- Ross, G.J.B., Cockcroft, V.G., Melton, D.A. and Butterworth, D.S., 1989. Population estimates for bottlenose dolphins *Tursiops truncatus* in Natal and Transkei waters. *South African Journal of Marine Science*, 8: 119-129.
- Ross, G.J.B., Heinsohn, G.E. and Cockcroft, V.G., 1994. Humpback dolphins Sousa chinensis (Osbeck, 1765), Sousa plumbea (G. Cuvier, 1829), Sousa teuszii (Kukenthal, 1892). In: S.H. Ridgway and S.R. Harrison (Editors), Handbook of marine mammals, Volume 5: The first book of dolphins. Academic Press, London, 23-42 pp.
- Rowntree, V.J., Payne, R.S. and Schell, D.M., 2001. Changing patterns of habitat use by southern right whales (*Eubalaena australis*) on their nursery ground at Península Valdés, Argentina, and their long-range movements. *Journal of Cetacean Resource Management*, Special Issue 2: 133-143.
- Saayman, G.S., Bower, D. and Tayler, C.K., 1972. Observations on inshore and pelagic dolphins on the south-eastern cape coast of South Africa. *Koedoe*, 15: 1-24.
- Saayman, G.S. and Tayler, C.K., 1979. The socioecology of humpback dolphins (Sousa sp.). In: H.E. Winn and B.L. Olla (Editors), *Behaviour of marine animals. Current perspectives in research 3: Cetaceans*. Plenum Press, New York, 165-226 pp.
- Saayman, G.S., Taylor, C.K. and Bower, D., 1973. Diurnal activity cycles in captive and freeranging Indian Ocean bottlenose dolphins (*Tursiops aduncus* Ehrenburg). *Behaviour*, 44: 212-233.
- Samaai, T., Best, P.B. and Gibbons, M.J., 2005. Taxonomic status of common dolphins *Delphinus* spp. in South African waters. *African Journal of Marine Science*, 27: 449-458.
- Santora, J.A., Reiss, C.S., Loeb, V.J. and Veit, R.R., 2010. Spatial association between hotspots of baleen whales and demographic patterns in Antarctic krill *Euphausia superba* suggests size-dependent predation. *Marine Ecology Progress Series*, 405: 255-269.
- Schumann, E.H., Perrins, L.A. and Hunter, I.T., 1982. Upwelling along the south coast of the Cape Province, South Africa. *South African Journal of Science*, 78: 238-242.
- Schumann, E.H. and van Heerden, I.L., 1988. Observations of Agulhas Current frontal features south of Africa, October 1983. *Deep-Sea Research*, 35(8): 1355-1362.
- Sekiguchi, K., Klages, N.T.W. and Best, P.B., 1992. Comparative analysis of the diets of smaller odontocete cetaceans along the coast of southern Africa. South African Journal of Marine Science, 12: 843-861.
- Selzer, L.A. and Payne, M.P., 1988. The distribution of white-sided (*Lagenorhynchus acutus*) and common dolphins (*Delphinus delphis*) vs. environmental features of the continental shelf of the northeastern United States. *Marine Mammal Science*, 4(2): 141-153.
- Seo, C., Thorne, J.H., Hannah, L. and Thuiller, W., 2009. Scale effects in species distribution models: implications for conservation planning under climate change. *Biology letters*, 5: 39-43.
- Shane, S.H., Wells, R.S. and Würsig, B., 1986. Ecology, behavior and social organization of the bottlenose dolphin: a review. *Marine Mammal Science*, 2(1): 34-63.
- Shelton, P.A., Crawford, R.J.M., Cooper, J. and Brooke, R.K., 1984. Distribution, population size and conservation of the jackass penguin *Speniscus demersus*. *South African Journal of Marine Science*, 2: 217-257.

- Silverman, B.W., 1986. *Density estimation for statistics and data analysis*. Chapman and Hall, London, UK, 22 pp.
- Simmonds, M.P. and Eliott, W.J., 2009. Climate change and cetaceans: concerns and recent developments. *Journal of the Marine Biological Association of the UK*, 89(1): 203-210.
- Simmonds, M.P. and Isaac, S.J., 2007. The impacts of climate change on marine mammals: early signs of significant problems. *Oryx*, 41(1): 19-26.
- Sironi, M., López, J.C., Bubas, R., Carribero, A., García, C., Harris, G., Intrieri, E., Iñígues, M. and Payne, R., 2008. Predation by killer whales (Orcinus orca) on southern right whales (Eubalaena australis) off Patagonia, Argentina: effects on behavior and habitat choice. Paper SC/60/BRG23 presented to the International Whaling Commission Scientific Committee. 18 pp.
- Skinner, J.D. and Chimimba, C.T. (Editors), 2005. *The mammals of the Southern African Subregion*, Third edition. Cambridge University Press, South Africa, 814 pp.
- Smith, J.N., Goldizen, A.W., Dunlop, R.A. and Noad, M.J., 2008. Songs of male humpback whales, *Megaptera novaeangliae*, are involved in intersexual interactions. *Animal Behaviour*, 76: 467-477.
- Smith, R.C., Dustan, P., Au, D., Baker, K.S. and Dunlap, E.A., 1986. Distribution of cetaceans and sea-surface chlorophyll concentrations in the California Current. *Marine Biology*, 91: 385-402.
- Smolker, R.A., Richards, A.F., Connor, R.C. and Pepper, J.W., 1992. Sex differences in patterns of association among Indian Ocean bottlenose dolphins. *Behaviour*, 123: 38-69.
- Socioeconomic Data and Applications Centre, 2010. Species distribution grids. http://www.gbif.org/. Accessed: 01/10/2010.
- SRK Consulting, 2007. Draft Coastal Management Programme for Nelson Mandela Bay. Report No. 366145-1. 181 pp.
- Stensland, E., Carlén, I., Särnblad, A., Bignert, A. and Berggren, P., 2006. Population size, distribution, and behavior of Indo-Pacific bottlenose (*Tursiops aduncus*) and humpback (*Sousa chinensis*) dolphins off the south coast of Zanzibar. *Marine Mammal Science*, 22(3): 667-682.
- Stone, A.W., Weaver, A.B. and West, W.O., 1998. Climate and weather. In: R. Lubke and I. de Moor (Editors), *Field guide to the Eastern and Southern cape coasts*, Second edition. University of Cape Town Press, Cape Town, 559 pp.
- Swartzman, G., Stuetzle, W., Kulman, K. and Powojowski, M., 1994. Relating the distribution of pollock schools in the Bering Sea to environmental factors. *International Council for the Exploration of the Sea*, 51: 481-492.
- Tershy, B.R., Breese, D. and Strong, C.S., 1990. Abundance, seasonal distribution and population composition of Balaenopterid whales in the Canal De Ballenas, Gulf of California, Mexico. *Report of the International Whaling Commission*, Special Issue 12: 369-375.
- The R project for statistical computing, 2010. R. Version 2.11.0.
- Thomas, P.O. and Taber, S.M., 1984. Mother-infant interaction and behavioral development in southern right whales, *Eubalaena australis*. *Behaviour*, 88(1/2): 42-60.

- Thompson, P.M., Wilson, B., Grellier, K. and Hammond, P.S., 2000. Analysis to compare traditional and precautionary approaches to conservation of coastal cetaceans. *Conservation Biology*, 14(5): 1253-1263.
- Townend, J., 2003. *Practical statistics for environmental and biological scientists*. John Wiley & Sons, Ltd., Chichester, 276 pp.
- Tynan, C., 1996. Characterization of oceanographic habitat of cetaceans in the Southern Indian Ocean between 82° - 115° E: cruise report from World Ocean Circulation Experiment (WOCE) 18S and 19S. Report No. NOAA Technical Memorandum NMFS-AFSC-64, U.S. Department of Commerce. 53 pp.
- Van Blaricom, G.R., Gerber, L.R. and Brownell, R.L.J., 2000. Extinctions of marine mammals. In: S.A. Levin (Editor), *Encyclopaedia of biodiversity (Volume 4)*. Academic Press, San Diego, 37-69 pp.
- van der Lingen, C.D., Coetzee, J.C. and Hutchings, L., 2010. Overview of the KwaZulu-Natal sardine run. *African Journal of Marine Science*, 32(2): 271-277.
- Van Parijs, S.M. and Corkeron, P.J., 2001. Boat traffic affects the acoustic behaviour of Pacific humpback dolphins, *Sousa chinensis. Journal of the Marine Biological Association of the UK*, 81: 533-538.
- Viddi, F.A., Hucke-Gaete, R., Torres-Florez, J.P. and Ribeiro, S., 2010. Spatial and seasonal variability in cetacean distribution in the fjords of northern Patagonia, Chile. *ICES Journal of Marine Science*, 67(5): 959-970.
- Vigness-Raposa, K.J., Kenney, R.D., Liliana Gonzalez, M. and August, P.V., 2009. Spatial patterns of humpback whale (*Megaptera novaeangliae*) sightings and survey effort: insight into North Atlantic population structure. *Marine Mammal Science*, 26(1): 161-175.
- Wang, J.Y. and Yang, S.C., 2009. Indo-Pacific bottlenose dolphin: *Tursiops aduncus*. In: W.F. Perrin, B. Würsig and J.G. Thewissen (Editors), *Encyclopaedia of marine mammals*, Second edition. Academic Press, United Kingdom, 602-608 pp.
- Watson, J.J., Kerley, G.I.H. and McLachlan, A., 1997. Nesting habitat of birds breeding in a coastal dunefield, South Africa and management implications. *Journal of Coastal Research*, 13(1): 36-45.
- Webb, P. and Wooldridge, T.H., 1990. Diel horizontal migration of *Mesopodopsis slabberi* (Crusacea:Mysidacea) in Algoa Bay, southern Africa. *Marine Ecology Progress Series*, 62: 73-77.
- Weigle, B., 1990. Abundance, distribution and movements of bottlenose dolphins (*Tursiops truncatus*) in lower Tampa Bay, Florida. *Report of the International Whaling Commission*, Special Issue 12: 195-201.
- Weir, C.R., MacLeod, C.D. and Calderan, S.V., 2009. Fine-scale habitat selection by whitebeaked and common dolphins in the Minch (Scotland, UK): evidence for interspecific competition or coexistence? *Journal of the Marine Biological Association of the United Kingdom*, 89(5): 951-960.
- Wells, R.S., Rhinehart, H.L., Hansen, L.J., Sweeney, J.C., Townsend, F.I., Stone, R., Casper, D.R., Scott, M.D., Hohn, A.A. and Rowles, T.K., 2004. Bottlenose dolphins as marine sentinels: developing a health monitoring system. *EcoHealth*, 1: 246-254.

- Wells, R.S. and Scott, M.D., 1999. Bottlenose dolphin *Tursiops truncatus* (Montagu, 1821). In: S.H. Ridgway and S.R. Harrison (Editors), *Handbook of marine mammals, Volume 6: The* second book of dolphins and porpoises. Academic Press, London, 137-182 pp.
- Welsh, A.H., Cunningham, R.B., Donnelly, C.F. and Lindenmayer, D.B., 1996. Modelling the abundance of rare species: statistical models for counts with extra zeros. *Ecological Modelling*, 88: 297-308.
- Wilson, B., Thompson, P.M. and Hammond, P.S., 1997. Habitat use by bottlenose dolphins: seasonal distribution and stratified movement patterns in the Moray Firth, Scotland. *Journal of Applied Ecology*, 34: 1365-1374.
- Winn, H.E. and Reichley, N.E., 1985. Humpback whale Megaptera novaeangliae. In: S.H.a.H. Ridgway, Sir R. (Editor), Handbook of marine mammals, Volume 3: The sirenians and baleen whales. Academic Press, Florida, 241-274 pp.
- Wiseman, N., 2008. Genetic identity and ecology of Bryde's whales in the Hauraki Gulf, New Zealand. Ph.D. Thesis. University of Auckland, New Zealand. 259 pp.
- World Bank, 2003. GEF project document on a proposed grant from the Global Environment Facility Trust Fund in the amount of US\$5.50 million equivalent to the Republic of South Africa for the Greater Addo Elephant National Park Project. Report No. 119 pp.
- World Bank, 2004. Project appraisal document on a proposed grant from the Global Environmental Facility trust Fund in the amount of US\$ 5.50 million equivalent to the republic of South Africa for the greater Addo Elephant National Park project. Report No. 25655-SA. 126 pp.
- Würsig, B., 1989. Cetaceans. Science, 244: 1550-1557.
- Young, J.W., 2000. Do large whales have an impact on commercial fishing in the South Pacific Ocean? *Journal of International Wildlife Law & Policy*, 3(3): 1-32.
- Zerbini, A.N., Secchi, E.R., Siciliano, S. and Simões-Lopes, P.C., 1997. A review of the occurrence and distribution of whales of the genus *Balaenoptera* along the Brazilian coast. *Report of the International Whaling Commission*, 47: 407-417.
- Zornetzer, H.R. and Duffield, D.A., 2003. Captive-born bottlenose dolphin x common dolphin (*Tursiops truncatus x Delphinus capensis*) intergeneric hybrids. *Canadian Journal of Zoology*, 81(10): 1755-1762.
- Zuur, A.F., Ieno, E.N., Walker, N.J., Saveliev, A.A. and Smith, G.M., 2009. *Mixed effect models and extensions in ecology with R.* Springer Science and Business Media, New York, 574 pp.

Personal Communications

Best, P.B. 2010. Mammal Research Unit, University of Pretoria, Pretoria.

- Booth, A.J., 2010. Department of Ichthyology and Fisheries Science, Rhodes University, Grahamstown.
- Oosthuizen, A., 2010. Park planning and development and national marine co-ordinator for the South African National Parks, Port Elizabeth.